

Review



# Review of the Terminology, Approaches, and Formulations Used in the Guidelines on Quantitative Risk Assessment of Chemical Hazards in Food

Eva Doménech <sup>1,\*</sup> and Sebastián Martorell <sup>2</sup>

- <sup>1</sup> Instituto Universitario de Ingeniería de Alimentos Food-UPV, Universitat Politècnica de València, Camino de Vera s/n, 46022 Valencia, Spain
- <sup>2</sup> MEDASEGI Research Group, Department of Chemical and Nuclear Engineering, Universitat Politècnica de València, Camino de Vera s/n, 46022 Valencia, Spain; smartore@iqn.upv.es
- \* Correspondence: evdoan@tal.upv.es

**Abstract:** This paper reviews the published terminology, mathematical models, and the possible approaches used to characterise the risk of foodborne chemical hazards, particularly pesticides, metals, mycotoxins, acrylamide, and polycyclic aromatic hydrocarbons (PAHs). The results confirmed the wide variability of the nomenclature used, e.g., 28 different ways of referencing exposure, 13 of cancer risk, or 9 of slope factor. On the other hand, a total of 16 equations were identified to formulate all the risk characterisation parameters of interest. Therefore, the present study proposes a terminology and formulation for some risk characterisation parameters based on the guidelines of international organisations and the literature review. The mathematical model used for non-genotoxic hazards is a ratio in all cases. However, the authors used the probability of cancer or different ratios, such as the margin of exposure (MOE) for genotoxic hazards. For each effect studied per hazard, the non-genotoxic effect was mostly studied in pesticides (79.73%), the genotoxic effect was mostly studied in PAHs (71.15%), and both effects were mainly studied in metals (59.4%). The authors of the works reviewed generally opted for a deterministic approach, although most of those who assessed the risk for mycotoxins or the ratio and risk for acrylamide used the probabilistic approach.

Keywords: health risk; safety margin; cancer risk; margin of exposure; hazard index; hazard quotient

## 1. Introduction

Chemical substances in food play an important role in nutrition and food preservation. However, some of the compounds incorporated or formed along the food chain can endanger the health of consumers [1]. Heavy metals are an example of chemical hazards of environmental origin, which are transferred from soil, water, air, etc., to raw materials [2,3]. Even at low concentrations, these highly toxic substances are non-biodegradable and accumulate in the body's target organs. Other contaminants are formed in food processing, such as acrylamide, which is produced from the Maillard reaction of asparagine with reducing sugars at high temperatures, or polycyclic aromatic hydrocarbons (PAHs), which form in processing stages, such as drying or smoking and cooking, e.g., grilling, roasting, and frying [4–6]. Hazardous compounds can also come from toxins of fungi, plants, and algae [7]. For example, mycotoxins are secondary metabolites of moulds that grow on numerous foodstuffs and can cause serious illnesses such as cancer or liver disease [8]. Chemical hazards can also arise from deliberate use to control crop pests, such as pesticides, or from on-farm veterinary treatments [9], while food contact materials such as formaldehyde, melamine, and phthalates can also be a source of chemicals [10].

Chemicals were the most frequently reported hazards in the Rapid Alert System for Food and Feed in 2021 [11], with pesticides in first place (1231 notifications) at 27% of health-related notifications, and mycotoxins in food in third position (450 notifications).



Citation: Doménech, E.; Martorell, S. Review of the Terminology, Approaches, and Formulations Used in the Guidelines on Quantitative Risk Assessment of Chemical Hazards in Food. *Foods* **2024**, *13*, 714. https://doi.org/10.3390/ foods13050714

Academic Editor: Robert L. Buchanan

Received: 18 January 2024 Revised: 15 February 2024 Accepted: 16 February 2024 Published: 26 February 2024



**Copyright:** © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). The other most frequently reported chemical hazards were allergens (198 notifications) and food additives and flavourings (176 notifications), mainly due to unauthorised or additive content levels that were too high.

In 1991, the FAO/WHO Conference on Food Standards, Chemicals, and the Food Trade recommended that the Codex Alimentarius Commission (CAC) incorporate risk assessment principles into decision-making processes. Since then, risk analysis has been accepted as an essential part of food safety consisting of three basic elements: risk assessment, risk management, and risk communication [12]. These three components represent essential and complementary parts, which must be integrated and fed back to obtain a practical risk analysis. In 2003, the working draft for applying risk analysis within the CAC framework was compiled. In 2007, guidelines were issued for national authorities. The FAO/WHO meeting in 2009 drafted the harmonisation, updating, and consolidation of principles and methods for risk analysis of chemicals in food, and in 2010 a guide for chemical risk assessment was published [13].

Risk management is a decision-making process in which political, social, economic, and technical factors are considered to control a hazard. Thus, risk managers must weigh the possible safety measures, choose the most appropriate, implement them, and monitor their effectiveness. For example, regulating an MRL, defining a safety factor, or banning a pesticide are risk management decisions. The risk analysis process usually begins with risk management, which, as a first step, defines the problem, articulates the objectives of the risk analysis and defines the questions to be answered by the risk assessment.

Risk communication is exchanging information about risk, such as risk assessment findings, risk management decisions, opinions, etc., throughout the risk analysis process between risk assessors, risk managers, consumers, industry, the academic community, and other interested parties.

Risk assessment is defined as the process of calculating the risk to a given target organism, system, or sub-population, including the identification of inherent uncertainties following exposure to a particular agent, plus the relationships between exposure and dose–response adverse effects [14]. This process may be carried out using either a deterministic or probabilistic approach, and the former means that each parameter of the risk equation takes a single value, e.g., the mean, the 95th percentile, the "worst-case", etc., so that the result would be a single value representing the risk for a single virtual consumer. This method tends to overestimate the risk and does not take into account the uncertainty inherent to the variability of the input data, such as the food consumption, chemical concentrations, physical differences between groups of exposed individuals, etc., [15–18]. To reduce this drawback, various methods were studied to evaluate the uncertainty-related results of deterministic models [19].

On the other hand, the probabilistic approach allows for the classification of problems and outcomes, the consideration and treatment of the variability, and uncertainty of the input parameters of the risk equation, defined using a probability density function; the calculations are performed using stochastic methods, such as Monte Carlo simulations [20] where the result is a risk probability distribution [17] and permits the application of optimization processes. However, it is pointed out that each probabilistic approach to risk analysis involves deterministic arguments, which help to decide how the likelihood of events will be addressed [21].

The bases for risk assessment and implementation are defined by expert advisory bodies, such as the US Environmental Protection Agency (EPA), the Joint Expert Committee on Food Additives (JECFA), the Food and Drug Administration (FDA), the European Food Safety Authority (EFSA), the European and Mediterranean Plant Protection Organisation (EPPO), the Council of Europe, and the European Centre for Ecotoxicology and Toxicology of Chemicals (ECETOC), and research groups have led to the creation of different approaches and nomenclatures, leading to confusion.

This review aimed to analyse how quantitative risk assessment is carried out for some of the most critical chemical hazards, such as pesticides, metals, mycotoxins, acrylamide, and PAHs. The document is organised as follows: Section 2 describes the materials and methods used, while Section 3 gives the results: to provide a basis for our findings, the fundamentals of assessing the risks of chemical hazards in food are first introduced, followed by the terms and formulations most frequently used in quantitative risk assessment, and finally we propose suggestions for harmonizing terminology and formulations. Section 4 discusses how the risks of pesticides, metals, mycotoxins, acrylamide, and PAHs are calculated in food, while Section 5 contains the concluding remarks.

### 2. Materials and Methods

The review included papers on the quantitative risk assessment (QRA) of heavy metals, pesticides, mycotoxins, acrylamide, and PAHs with deterministic and probabilistic approaches [17]. A systematic review was conducted considering the PRISMA guidelines, including the search strategy, article selection, and evaluation criteria [22].

A total of 348 articles were selected, of which 74 dealt with pesticides, 133 with metals, 63 mycotoxins, 27 acrylamide, and 51 with PAHs. The search strategy was conducted according to the Cochrane protocol [23], in which the best keywords and synonyms were found using MeSH terms and checking keywords in relevant articles and review papers. These terms were used to retrieve all the related articles in the Scopus, Google Scholar, PubMed, and ISI Web of Science international databases. The title, abstract, and keywords were used to apply the selection criteria to the articles published between 2015 and 2022.

The exclusion criteria applied were as follows:

- Books, clinical studies, abstracts, presentations, theses, reviews, commentaries, metaanalyses, conference papers, editorials, and letters to the editor.
- Duplicate content, not written in English, or in non-peer-reviewed journals.
- Articles in which risk was not assessed.
- Acute toxicological endpoint.
- Articles related to environmental risk, soil, water, pollution, and dust.
- Herbal medicines and breastmilk.
- Biomonitoring studies.
- Experimental lab studies to check the influence of treatment conditions, etc.
- Studies with less than five analysed samples.
- Non-marketable products.
- The same authors with the same terminology and risk formulation.

#### 3. Results

#### 3.1. Background of Risk Assessment of Chemical Hazards in Food

Risk assessment consists of four stages: hazard identification, hazard characterisation, exposure assessment, and risk characterisation. Figure 1 shows an outline of the terms used in the studies reviewed.

Hazard identification decides whether a chemical present in a given food or group of foods has the inherent capacity to cause adverse health effects to consumers and should, therefore, be considered a hazard [24]. All the available data on toxicity and its mode of action (MOA) must be considered to determine the type and nature of an adverse effect. The first key question is to identify whether the compound or its active metabolite reacts covalently with DNA (genotoxic) or whether it has an epigenetic action (non-genotoxic) [25]. In the former case, there is no (threshold) dose that has no potential effect, and DNA damage increases with the dose administered. However, in non-genotoxic cases, it is often assumed that there is an exposure level below which no significant effect will be induced [26]. This difference often determines the choice of the risk assessment methodology.

To identify chemicals that may have an adverse effect on health, risk assessors can rely on the sources of hazard information published by international organisations. These include the Openfoodtox database, which compiles chemical and toxicological data on chemicals evaluated by EFSA since its inception [27]. CompTox Chemicals Dashboard is the EPA's computational toxicological research database that provides chemical, toxicological, and exposure information on more than 900,000 chemicals, with more than 300 lists of chemicals, based on their structure or category [28]. Another advisory source for the self-classification of chemicals, which lists more than 54,000 substances, is published by the Danish Environmental Protection Agency [29]. The New Zealand Environmental Protection Agency also publishes detailed information on each chemical's hazards, classification, studies, and physical properties [30].



Figure 1. Elements and main parameters of the quantitative risk assessment.

Hazard characterisation describes the relationship between the administered dose of a chemical and its MOA or adverse health effect [13]. This relationship is obtained by fitting epidemiological or experimental data obtained from animal or human studies to a dose–response curve [14,31], so that low or no effects to the dose associated with a hazard are identified as the point of departure (POD) or reference point (RP), Figure 1 [32,33]. The no-observable-adverse-effect level (NOAEL) or the lowest-observed-adverse-effect level (LOAEL) are usually taken as the baselines or PODs for non-genotoxic effects. The reference dose (BMD) is another POD derived from the dose–response curve. A predetermined response (BMR) thus identifies a corresponding dose (BMD) or its lower limit (BMDL) defined using the statistical confidence level, typically with 95% confidence, meaning that at a 95% confidence level, the chosen BMR is not exceeded [32–34].

The BMD or BMDL can be used for non-genotoxic and genotoxic damages. In the first case, it is preferred to the NOAEL or LOAEL for several reasons. Firstly, it is a starting-point estimate based on a NOAEL, which relies solely on identifying a no-effect dose and does not consider the shape of the dose–response curve, and so does not allow the estimation of the probability of response for any dose level. The experimental response observed in the NOAEL may vary between studies, which makes comparison difficult. While NOAEL identification is highly dependent on sample size, a low response rate will have a lower statistical sensitivity to detect small changes, so this type of study tends to generate higher

NOAELs. The NOAEL method does not account for variability and uncertainties in the data due to random errors such as animal dosing and response measurements [33].

The next step is to specify a reference value (RV), defined as the estimated maximum dose (based on body mass) or concentration of an agent to which an individual can be exposed during a given period without an appreciable risk or a predetermined change in the response rate of an adverse effect. To derive an RV from the NOAEL, this POD must be divided by uncertainty factors to account for interspecies and intraspecies variability, data quality, and other uncertainties arising from the study [35]. Some of the best-known RVs are the health-based guideline values (HBGVs), which include the acceptable daily intake (ADI), developed for food and feed additives, pesticides, and food-contact materials; the tolerable upper intake levels (UL) for vitamins and minerals; the tolerable daily intake (TDI) [35]; the tolerable weekly intake (TWI), in cases where compounds tend to accumulate in the body, such as cadmium, dioxins, or ochratoxin A. The reference dose (RfD) is also an RV. This term is described as an estimate of the daily oral exposure of the human population that is likely to have no appreciable adverse health effects [36]. Another possible RV is the TTC, i.e., threshold of toxicological concern, used for compounds of a known structure, for which exposure is low, but without sufficient experimental data for a fully quantitative risk assessment [37]. Finally, PODs such as the LD 50 (lethal dose to 50% of the population), the T25 (dose that causes a tumour incidence of 25%), and the TD50 (daily dose rate necessary to halve the probability of animals remaining tumour-free at the end of their lives), divided by a safety factor can be used as RVs [38,39]. For example, 50,000 is the uncertainty factor for mycotoxins, equivalent to a risk level of one person in 100,000 inhabitants [40].

To obtain an RV from the BMDL, an increase in response (BMDL%) must be defined, e.g., p = 1%, which means that the incidence (level of response) has increased by 1% relative to the background response. A linear dose–response relationship is assumed at very low doses below the BMDL%, while the increase in the population risk of the effect expressed by the daily dose of a chemical hazard consumed is called the slope factor (SF), Figure 1.

Exposure assessment is the qualitative or quantitative evaluation of the likely intake of a chemical via food [41,42] whose calculation is a function of consumption, concentration of the chemical hazard, and personal weight. As food consumption data are the basis for assessing human exposure, comprehensive information is needed. In this respect, EFSA in the database FoodEx2's harmonised food consumption data across the European Union (EU). The information is divided by country, food category, age, and sex [43]. The FAO/WHO also provides a chronic individual food consumption database known as CIFOCOss [44]. To estimate dietary exposure to food chemicals, official bodies have developed freely available tools for assessing dietary exposure to food chemicals, such as, DietEx Tool, FAIM for additives, and PRIMO for pesticides, EPA ExpoBox [45–48].

Risk characterisation is the final step in risk assessment and results from the combination of hazard characterisation and exposure assessment. It is defined as the qualitative and/or quantitative estimation of the exposure assessment and the severity of known or potential adverse health effects in a given population [49]. Ratio metrics are applied to assess the risk of non-genotoxic effects, e.g., the hazard quotation, while there are two options for genotoxic effects: to estimate the probability of developing cancer or to assess a ratio, e.g., the margin of exposure (MOE), Figure 1.

#### 3.2. Risk Assessment and Hazards

Figure 2 shows the number of articles published per year (according to hazards studied) and the trend in using the deterministic or probabilistic approach in risk calculation. The results indicated that the number of articles published increased annually, especially in 2021, possibly due to the COVID-19 pandemic [50,51]. We also found that the deterministic approach was the most frequently used.



**Figure 2.** Distribution of the number of papers selected by year of publication and hazard using a deterministic or probabilistic approach to QRA.

Tables 1–5 for pesticides, metals, mycotoxins, acrylamide, and PAHs, respectively, give the information about the utilised risk assessment approach, i.e., deterministic (D) or probabilistic (P); the exposure metric (abbreviations in Tables 6 and 7), and the equations used by the authors (see Section 3.2.1); the damage, indicating the MOA (G = genotoxic and NG = non-genotoxic) and the RV applied. The last column, on risk characterisation, gives the metric used to calculate the risk (see Tables 8–10) and the equation used (see Section 3.2.2).

The articles on pesticides were mainly from China (23.5%), India (10.6%), and Iran (9.4%). The products analysed were mainly on "fruits, vegetables, and legumes". Almost half the manuscripts studied the risk in adults and children (45.9%), followed by the adult population (38.8%). Exposure was assessed using Equation (1) in 71.8% of cases. The non-genotoxic risk effect was calculated using the ratio with Equations (4) and (5), using the ADI as the reference value at 71.2%, followed by the RfD at 22%. The risk of cancer, Equation (14), was the first option for assessing the genotoxic effect (66.7%).

Fable 1. Exposure, damage	e, and risk characterisatio	on methodology used	l in pesticide studies.
---------------------------	-----------------------------	---------------------	-------------------------

Reference	D/P	Exposure		Damage (	Effect)	<b>Risk Charac</b>	terisation
		Metric	Eq.	MOA	RV	Metric	Eq.
Beekeeping products							
[52]	D	ADI	(1)	NG	RfD	HQ, HI	(4), (6)
[53]	D	Exposure	(1)	NG	ADI	HQ, HI	(4), (6)
[54]	D	EDI	(3)	NG	RfD	THQ	(4)
				G	SF	CR	(14)
[55]	D	NEDI	(1)	NG	ADI	%ADI	(6)
[56]	D	EDI	(1)	NG	ADI, RfD	HQ, HI	(4) **, (6)
				G	CPF	CRk, CR	(14) **, (16)
Cereals and bakery products							
[57]	D	EDI	(1)	NG	ADI	HI	(4)
				G	CBC	HR	(12)
[58]	Р	CDI	(3)	NG	RfD	HQ, HI	(4), (6)
[59]	D	EDI	(2)	NG	ADI	HQ	(4)

Table 1. Cont.

Reference	D/P	Exposure		Damage (Ef	fect)	<b>Risk Characte</b>	risation
		Metric	Eq.	MOA	RV	Metric	Eq.
[60]	D	EDI	(1)	G	CBC	HR	(12)
[61]	D	ADD	(3)	NG	RfD	HQ, HI	(4), (6)
				G	CSF	CR	(14)
[62]	D	EDI	(1)	NG	ADI	RQ	(4)
[63]	Р	у	(1)	NG	TDI	MOS	(7)
Fats and oils							
[64]	Р	CDI	(3)	NG	RfD	THQ, TTHQ	(4), (6)
				G	CSF	CR	(16)
Fish and shellfish							
[65]	D	EDI	(1)	NG	ADI, RfD	HQ	(4)
	_			G	BMC	HR	(12)
[66]	D	-	-	NG	RfD	THQ	(4) **
[67]	D	EDI	(1)	NG	TDI	HQ	(4)
[(0]	D	EDI	(2)	G	SF	ILCR	(14) **
[68]	D	EDI	(3)	NG	RfD	THQ	(4)
Fruits, vegs, and legumes							
[69]	D	EADI	(1)	NG	ADI	HRI, CHI	(4), (6)
<b>F</b> =01	-			G	CBC	HR	(12)
[70]	D	EDI	(1)	NG	ADI	%HQ	(4)
[71]	D	EDD	(1)	NG	ADI		(4)
[72]	D	EDI	(1)	NG	ADI	%cHQ, HI	(4)
[73]	D	EDI	(1)	NG	ADI	HKI, ZHI	(4), (6)
[74]	D	EDI	( <i>L</i> ) (1) *	NG NC	ADI	%KQ	(4)
[73]	D	NEDI	(1)	NG	ADI	MQ HO HI	(4) (6)
[16]	Р	EDI	(1)	NG	ADI	and p-FSM	(4), (0) and (8)
[76]	D	NEDI	(1)	NG	ADI	%ADI	(6)
[77]	D	EDI	(1)	NG	ADI	%HQ	(4)
[78]	D	EDI	(1)	NG	ADI	%HQ, cHI	(4), (6)
[79]	D	Exposure	(1)	NG	ADI	%HQ	(4)
[80]	Р	EDI	(3)	NG	RfD	HQ	(4)
[81]	D	-	-	NG	ADI	%ADI	(6)
[82]	D	EDI	(1)	NG	ADI	HHI	(4)
[83]	D	EDI	(1)	NG	ADI	%HQ	(4)
[84]	D	NEDI	(1)	NG	ADI	%ADI, HI	(5), (6)
	D	EDI	(1)	NG	ADI	THQ, HI	(4), (6)
[85]	D	EDI	(2) *	NG	ADI	HQ	(4)
[86]	D	EDI	(1)	NG NC	ADI	%HQ, CHI	(4), (6)
[07]	D	EDI	(1) (1)	NG	ADI		(4)
[00]	D	EDI	(1) (1)	NG		THO HI	(0) (4) (6)
[09]	D	FDI	(1) (2) *	NG		IFS	(4), (0)
[90]	D	FDI	(2) (1)	NG	ADI	%ADI	(4)
[92]	D	FDI	(1)	NG	ADI RfD	THO HI	(4) (6)
[92]	D	EDI	(1)	NG	ADI	HO. HI	(4), (6)
[94]	D	EDI	(1)	NG	ADI	%HO, HI	(4), (6)
	D		(1)	NG	ADI,	%ADI and	
[95]	D	AFE	(1)	NG	NOAEL	MOE, MOEt	(6) and (7), 10
[96]	Р	CDI	(3)	NG	ADI	HQ, HI	(4), (6)
[97]	D	EDI	(1)	NG	RfD	HI	(4) **
[00]	-		( <b>1</b> )	G	SFO	TCR	(14) **
[98]	D	EDI	(1)	NG	ADI	HQ	(4)
[99]	D D	EDI	(2) *	NG	ADI	HQ, HI	(4), (6)
[100]	D	EDI	(1)	NG	ADI	%ADI	(6)
[101]	D	EDI	(1)	NG	ADI	ihq, hi	(4), (6)

Reference	D/P	Exposure		Damage (E	ffect)	<b>Risk Charac</b>	terisation
		Metric	Eq.	MOA	RV	Metric	Eq.
Milk and dairy products							
[102]	D	EDI	(2)	NG	ADI	THQ, HI	(4), (6)
[103]	D	EDI	(1)	NG	RfD	HR	(4)
				G	BMC	HR	(12)
[104]	D	Exposure	(1)	NG	RfD	HQ, HI	(4), (6)
[105]	D	cEDI	(1)	NG	RfD	HQ, HI	(4), (6)
[106]	Р	CDI	(3)	NG	RfD	HQ, HI	(4), (6)
				G	CSF	CR	(14)
[107]	D	EDI	(1)	NG	RfD	HR	(4)
				G	BMC	HR	(12)
[108]	D	EDI	(1)	NG	ADI	-	(4)
				G	CBC	HR	(12)
Miscellaneous							
[109]	D	EDI	(1)	NG	ADI	cHQ, cHI	(4), (6)
[110]	D	EDI	(1)	NG	ADI	HRĨ	(4)
[111]	D	EDI	(1)	NG	ADI	HQ, HI	(4), (6)
[112]	D	Exposure	(1)	NG	ADI	HQ, HI	(4), (6)
[113]	Р	EDI	(1)	NG	RfD	HI	(4)
				G	CSF	LCR	(14)
Nuts, nuts products, and seeds							
[114]	D	EDI	(1)	NG	ADI	%cHQ	(4)
[115]	D	EDI	(1)	NG	ADI	%HQ, cHI	(4), (6)
[116]	Р	CDI	(3)	NG	RfD	HQ, THQ	(4), (6)
				G	CSF	CR	(16)
Tea, herbs, and spices							
[117]	D	EDI	(1)	NG	ADI	HQ, HI	(4), (6)
[118]	Р	LADD	(1)	NG	ADI	HQ, HI	(4), (6)
[119]	Р	CDI	(1)	NG	RfD	%HQ, THQ	(4), (6)
[120]	D	EDI	(1)	NG	ADI	HQ, HI	(4), (6)
Total diet studies							
[121]	Р	IEDI	(1)	NG	ADI	HQ, HI	(4), (6)
[122]	D	EDI	(1)	NG	RfD	HQ	(4)
[123]	D	EDI	(1)	NG	ADI	HQ, HI	(4), (6)

Table 1. Cont.

\* When the exposure units are mg/day per person. \*\* When the exposure used is Equation (3).

Table 2 shows the results for heavy metals. "Fish and shellfish" were the most studied group of foodstuffs. Most manuscripts came from China (21.4%), Bangladesh (13.5%), and Iran (11.16%) and were mainly focused on the adult population (59.6%). Exposure was assessed using Equations (1) and (3) (approximately 45% for both). Equation (4) was used to characterise the risk in 98.4% of the non-genotoxic studies, and RfD was the most frequently used RV (92%). The cancer risk was assessed using Equation (14) for 95.3% of the cases of studies on the genotoxic effects.

Table 2. Exposure, damage, and risk characterisation methodology used in metal studies.

Reference	Elements	D/P	Exposure		Damage (Effect)		<b>Risk Characterisation</b>	
			Metric	Eq.	MOA	RV	Metric	Eq.
Beekeeping prod	ducts							
[124]	Cd, Cr, Cu, Mn, Pb, Zn	D	EDI	(3)	NG	RfD	THQ, HI	(4), (6)
	Cd, Cr				G	CPS	TCR	(14)
[52]	Mn, Cu, Zn	D	ADI	(1)	NG	RfD	HQ, HI	(4), (6)
[125]	Fe, Ni, Cu, Zn, Pb	D	ADI	(3)	NG	RfD	HQ, HI	(4), (6)
	Cr, Cd, As, Ni				G	SF	CR, CRt	(14), (16)

Reference	Elements	D/P	Exposure	2	Damage (	Effect)	Risk Chara	cterisation
			Metric	Eq.	MOA	RV	Metric	Eq.
[126]	Pb	D	DIM	(1)	NG	RfD	THQ, HI	(4) **, (6)
[54]	Pb Pb, Cd, As, Hg, Cu, Zn, Ni	D	CDI EDI	(3) (3)	G NG	CSF RfD	ILCR THQ	(14) (4)
	Pb, Cd, As, N1 Hg, Cd, V. Cr. Ni, Cu, As, Sb.	_			G	SF	CR	(14)
[127]	Pb, Ba, Mn	D	EDI	(3)	NG	RfD	THQ, HI	(4), (6)
	Ni, Cr, Pb, As, Cd				G	CSF	LTCR, LTCRtot	(14), (16)
Beverages								
[128]	Cd, Co, Cr, Cu, Fe, K, Mn, Na, Ni, Pb, Zn	D	EDI	(1)	NG	RfD	THQ, TTHQ	(4) **, (6)
Cereals and bake	ery products							
[129]	As, Cd, Pb	D	-	-	NG	PIWI, TWI	HQ, HI	(4), (6)
[130]	As, Cd, Fe, Ni, Pb	D	DIM	(3)	G NG	CSF RfD	CK HO, HI	(14) (4), (6)
[]	As, Cd, Ni, Pb			(0)	G	SF	CR, TCR	(14), (16)
[131]	Al, As, Cd, Cu, Mo, Pb As	Р	EDI	(3)	NG G	RfD SF	HQ, HI CR	(4), (6) (14)
[132]	Zn, Cu, Cd, Pb, As, Al	D	EDI	(3)	NG	RfD	THQ, TTHQ	(4), (6)
[133]	Cd, Cr, Pb, Zn, Ni, Cu, Hg, As	Р	EDI	(3)	NG	RfD	THQ, TTHQ	(4), (6)
	Cd, Cr, Pb, Ni, As				G	CSF	ILCR, TCR	(14), (16)
[134]	Cd, As, Sn, Pb, Hg Pb, As, Cd	D	EDI	(3)	NG G	RfD CSF	THQ, HI TR, TRt	(4), (6) (14), (16)
[135]	As, Zn, Fe, Cu	D	ADD	(1)	NG	RfD	HQ	(4)
[136]	As Ph. Cd	П	EDI FDI	(3)	G NG	CSF RfD	ILCR THO	(14) (4) **
[130]	Cd Cr Dh Cu Es Mr. Zr	D	DED	(1)	NG	RfD and	HQ, HI	(4), (6)
[137]	Ca, Cr, Pb, Cu, Fe, Mn, Zn	D	DED	(1)	NG	ADI	and R	and (4)
[63]	Pb As Cd Pb	D P	EDI v	(1) (1)	G NG	CSF PTDI	CR MOS	(14) (7)
[138]	Ni, Pb, Zn, Cd, Cr, Cu,	D		(3)	NG	RfD	но ні	(4) (6)
[100]	Mn, As, As, Cr, Ni	D	MDD	(0)	G	CSE	II CR	(1), (0) (14)
	As, Al, B, Ca, Cd, Cr, Cu, Fe,				G	Cor	ILCK	(14)
[139]	Hg, K, Mg, Mn, Na, Ni, Pb, Se, Zn	D	EDI	(1)	NG	RfD	THQ, HI	(4), (6)
Coffee, cocoa, an	d preparations							
[140]	Cd, Pb	D	EDI	(1)	NG	TWI, BMDL,	%TWI, %BMDL,	(5), (4), (6)
	Cd, Pb		CDI	(3)	G	CSF	THQ, HI CR	(14)
Fats and oils	,			( )				· · /
[141]	Pb, Cu, Cd, Cr, As, Zn	D	CDI	(3)	NG	RfD	THQ, HI	(4), (6)
	As, Cd, Cr, Pb				G	CSF	ILCR, ΣILCR	(15), (16)
[142]	Pb, As, Cd, Cr	Р	-	-	NG	RfD	THQ, TTHO	(4) **, (6)
					G	CFS	ILCR	(14) **
Fish and shellfis	h							
[143]	As, Cd, Hg, Pb	D	EDI	(3)	NG	RfD	THQ, TTHQ	(4), (6)

Table 2. Cont.

Table 2. Cont.	
----------------	--

Reference	Elements	D/P	Exposur	xposure Damage (Effect)		<b>Risk Characterisation</b>		
			Metric	Eq.	MOA	RV	Metric	Eq.
[144]	Cd, Cr, Cu, Fe, Mn, Pb, Zn	D	EDI	(3)	NG	RfD	THQ, TTHO	(4), (6)
	Ph				G	CSF	CR	(14) **
[145]	Al Sn Zn	D	FDI	(1)	NG	RfD	HRI	(14)
[146]	As Cr Cd Ph	D	FDI	(1)	NG	RfD	THO HI	(4) ** (6)
	As Cr Cd Ph	D	LDI	(1)	G	CSE	CR	(14) **
[147]	Но	D	FWI	(1)	NG	PTWI	%PTWI	(14) (5)
	As Cd Cr Cu Fe Ho Mn	D	1111	(1)	Ĩ	1 1 1 1 1	/01 1 //1	(0)
[148]	Ph $Zn$	D	EDI	(1)	NG	RfD	THQ, HI	(4) **, (6)
	As Cd Ph				G	CPS	TR	(14) **
[149]	Cd. Cr. Fe. Ni. Zn. Mn. Ph	D	-	-	NG	RfD	THO, HI	(4) **. (6)
[1 -> ]	Cd. Cr. Ni. Pb	2			G	CSF	TR	(14) **
	As. Cd. Cr. Cu. Hg. Mn. Ni.				C	001		(11)
[150]	Ph Zn	D	EDI	(1)	NG	RfD	THQ, HI	(4) **, (6)
	Cd. Pb				G	CPSo	TR	(14) **
[151]	As, Cd, Hg, Pb	D	-	-	Ğ	LD50	RI	(11)
[152]	As, Cd, Cu, Cr, Pb, Zn	P	EDI	(1)	NG	RfD	THO, HI	(4) **, (6)
[]				(-)			Risk.	(-) /(-)
	As, Cd, Cr, Pb				G	CSF	Σrisk	(14) **, (16)
[153]	Al, As, Cd, Cr, Fe, Cu, Hg, Ni, Ph. Zn	D	EDI	(3)	NG	RfD	THQ, HI	(4), (6)
	As, Cd, Cr, Ni, Pb				G	SF	CR	(14)
[154]	As, Cd, Hg, Pb	D	-	-	NG	RfD	THQ,	(4) **, (6)
	As. Cd. Co. Cr. Hg. Mn. Ni.	_					TINQ	
[155]	Pb, Zn	D	-	-	NG	RfD	THQ, HI	(4) **, (6)
[156]	Hg, Pb, Cd	D	EDI	(3)	NG	RfD	THQ, TTHO	(4), (6)
[157]	LL- Dh C d NI: Cr. As Cr.	п	EDI	( <b>2</b> )	NC	ВſЪ	THQ,	(A) $(C)$
[157]	Hg, Pb, Ca, NI, Cr, As, Sn	Р	EDI	(3)	NG	RID	TTHQ	(4), (6)
[158]	Hg	D	EWI	(1)	NG	PTWI,	%Risk	(5)
	As, Pb, Hg, Cd, De, Sn, Zn, Cr, Fe,					RfD	ТНО	(1) **
	Co, Ni, Al					KID	ШQ	(4)
[150]	Co Cu Ni Ph Zn	Р	FDI	(3)	NC	RfD	THQ,	(4) (6)
[157]	C0, Cu, Ni, 10, Zit	1	LDI	(5)	NG	КШ	TTHQ	(4), (0)
[160]	Cu, Fe, Hg, Zn, Pb	D	Ι	(3)	NG	RfD	TS	(4)
	Pb				G	SF	CR	(14)
[161]	Ag, As, Be, Cd, Co, Cr, Cu, Fe,	D	EDI,	(1)	NG	RfD,	THQ,	(4) ** (6)
	Mn, Mo, Ni, Sn, Pb, Zn	D	EWI	(1)	Ĩ	PTWI	TTHQ	(1) ,(0)
	iAs, Cr, Ni, Pb		CDI	(3)	G	CSF	ILCR	(14)
[162]	As, Cd, Co, Cr, Hg, Pb	D	DIM	(1)	NG	RfD	HRI, HI	(4), (6)
[163]	Mn, Fe, Co, Ni, Cu, Zn	D	EDI	(1)	NG	RfD	THQ, HI	(4) **, (6)
	Ni				G	CPS	TR	(14) **
[164]	Cd, Cr, Mn, Ni, Zn	D	EDI	(1)	NG	RfD	THQ, HI	(4), (6)
	Mn, Zn, Ni, Pb, Cr, Cd				G	CSF	CR	(14)
[165]	As, Cd, Cr, Pb	D	EDI	(1) *	NG	RfD	THQ	(4) **
	As, Cd, Cr, Pb	_			G	CSF	CR	(14) **
[166]	Hg, Cu, As, Cd, Pb	D	EDI	(1)	NG	RfD	THQ, HI	(4) **, (6)
	iAs				G	CPSo	TR	(14) **
[167]	As, Cr, Cu, Fe, Mn, Ni, Sn, Pb, Zn	D	EDI	(1) *	NG	RfD	THQ, TTHO	(4) **, (6)
	As, Pb				G	CSF	CR	(14) **
[168]	Pb, Cu, Cr, Zn, Cd, As, Hg	Р	EDI	(1)	NG	RfD	THQi,	(4) **, (6)
	A a Dh				C	OCE	THQS	(11) **
[1/0]	AS, FU Ze Me Cu Nº Ce C L DI	Р	EDI	(1)		USF DED		$(14)^{**}$
[109]	$\Delta \Pi$ , $\Pi$ , $\Omega$ , $\Pi$ , $\Omega$ , $\Pi$ , $\Omega$ , $\Omega$	υ	EUI	(1)	ING	MD	INQ	(4)

# Table 2. Cont.

Reference	Elements	D/P	Exposure	2	Damage (Effect)		<b>Risk Characterisation</b>	
			Metric	Eq.	MOA	RV	Metric	Eq.
[170]	As, Cd, Pb, Cr, Cu, Ni Zn	D	-	-	NG	RfD	THQ, TTHQ	(4) **, (6)
[171]	As, Cr, Pb Pb, As, Mn, Fe, Zn, Ni, Cr	D	EDI	(3)	G NG	CSF RfD	CR HO, HI	(14) ** (4), (6)
	Ni, As, Pb, Cr, Cd			. /	G	CSF	TCR	(14)
[172]	Cr, Cu, Fe, Pb, Cd	D	EDI	(1) *	NG	RfD	THQ, TTHQ	(4) **, (6)
[170]	Pb Zr. Ph. Cr. Cd. Cr.	D	EDI	(1)	G	CSF	CR	(14)
[173]	Cr, Pb, Fe, Zn Ni	D	EDI EDI	(1) (1)	NG	RfD RfD	HQ, HI	(4) ** (4) **, (6)
[175]	Cr, N1, Pb Hg, Cd, Pb	D	EDI	(3)	G NG	CSF RfD	CR THQ, HI	(14) ** (4)
[176]	Cd Pb, Cd, Hg, As, Al, Fe, Zn, Cu, Ni,	D	EWI	(1)	G NG	RfD	CK THQ, HI	(14) (4) **, (6)
[177]	Co, Cr As, Cd, Co, Cr, Mn, Mo, Ni, Pb, Se	D	EDI	(1)	NG	RfD	THQ, HI	(4) **, (6)
[178]	As, Cr, Ni, Pb As, Cd, Pb, Hg	D	PTWI	(1)	G NG	CSF RfD	CR HQ	(14) ** (4) **
	As, Pb	P		(1)	G	CSF	CR	(14) **
[179]	Cr, Mn, Cu, Zn, Pb, Co, Rb, V Cr. Pb	D	EDI	(1)	NG G	RfD CSF	THQ, HI CR TCR	(4) **, (6) (14) ** (16)
[180]	Cd, Cu, Pb, Zn	D	-	-	NG	RfD	THQ, HI	(4) **, (6)
[181]	Cr, Ni, Cu, Zn, As, Cd, Pb	D	EDI	(1)	NG C	RfD CSE	- TP	(4) (14) **
[182]	As, Cd, Co, Cr, Cu, Mn, Ni, Pb,	D	EDI	(3)	NG	RfD	THQ,	(14)
	Se, Zn iAs				G	CSF	CR	(14)
[183]	Zn, Cd, Mn, Cu, Cr, Pb, Fe, Co	D	EDI	(1)	NG	RfD	THQ, HI	(4) **, (6)
<b>5</b> ( <b>0</b> )	Pb, Cr, Cd	-		(=)	G	CSF	CR	(14) **
[184]	Ni, Zn, Cu, Cr, Cd, Pb Ni, Cr, Cd, Pb	D	EDI	(3)	NG G	RfD CPS	THQ, HI TR	(4), (6) (14) **
[185]	Zn, Cu, Cr, Pb, Cd, Hg. Cd. Cr. Pb	D	EDI	(1)	NG G	RfD CSF	THQ TR	(4) ** (14)
[186]	As	D	EDI	(1)	NG	RfD	HQ, THQ	(11) (4), (6) (14)
[187]	As, Cd, Co, Cr, Cu, Hg, Mn, Ni,	D	EDI,	(1)	G NG	RfD	THQ,	(14)
[188]	Pb, Se, V, Zn Hg, Cd, Pb, V, Ni, Co, Cr, Cu, Zn	D	EWI EDI	(1)	NG	RfD	TTHQ HO, HI	(4) **. (6)
	Cd, Pb, Ni, Cr	D	CDI	(3)	G	CSF	ILCR	(1) (0) (14)
[189]	As, Cr, Cd, Pb As	D	EDI	(1)	NG G	RfD CSF	THQ CR	(4) ** (14) **
[190]	As, Cd, Cr, Cu, Li, Hg, Fe, Pb, Zn	D	EWI	(1)	NG	PTWI, RfD	THQ, HI	(4) **, (6)
	As, Cd, Pb				G	CSF	CR	(14) **
[191]	Pb, Cd, Cr, As, Hg	D	EDI	(1) *	NG	RfD	THQ, TTHQ	(4) **, (6)
[100]	Pb, As	р	EDI	(2)	G	CSF	CR	(14)
[192]	Pb, Cd, Hg, As, Cr	D P	EDI EDI	(3)	NG NG	RfD	HQ, HI HQ	(4), (6) (4)
	Pb, Cd, As, Cr				G	SF	ILCR	(14)
[194]	Al, V, Cr, Fe, Co, Zn, Cu, Cd Pb	D	-	-	NG	RfD	TTHQ, TTHQ	(4) **, (6)
[195]	Cu, Pb, Zn, Fe, Mn, Cd, Cr, Ba, As As, Pb	D	EDI	(1)	NG G	RfD CSF	THQ CR	(4) ** (14) **
[196]	Cr, As, Cd, Pb, Zn, Cu, Ni, Hg	D	-	-	NG	RfD	THQ, TTHQ	(4) **, (6)

Reference	Elements	D/P	Exposur	e	Damage	(Effect)	Risk Chara	cterisation
			Metric	Eq.	MOA	RV	Metric	Eq.
[197]	As, Cr, Cd, Hg, Cu, Zn, Pb, Fe	D	EDI	(1)	NG	RfD	THQ, TTHO	(4) **, (6)
[198]	As, Cd, Cr, Cu, Hg, Pb As, Cd, Pb	D	-	-	NG G	RfD CSF	THQ, HI TR	(4) **, (6) (14) **
[199]	Pb, As, Cd, Cr, Zn, Cu, Mn, Ni iAs	D	EDI	(1)	NG G	RfD CSF	THQ THQcarcin.	(4) ** (14) **
Fruits, vegs,	and legumes							
[200]	Cd, Cr, Co, Cu, Fe, Pb, Zn	D	EDI	(1) *	NG	RfD	HRI and THO, HI	(4) and (4) **. (6)
[201]	Cd, Cr, Pb Al, As, Cd, Pb As	D	EDI	(1)	G NG G	CPS RfD CPSo	TCR THQ, HI TCR	(14) ** (4) **, (6) (14) **
[202]	Cd, Cu, Hg, Pb, Se, Zn	D	DIM	(1)	NG	RfD	HRI	(4)
[203]	Cd, Co, Cr, Cu, Fe, Li, Mn, Pb, Zn	D	-		NG	RfD	HRI and THQ, HI	(4) and (4) **, (6)
[204]	Cd, Cr, Pb As, Cd, Cr, Cu, Pb, Zn	D	-	-	G NG	CSF RfD	TCR THQ, HI	(14) ** (4) **, (6)
[205]	Cu, Ni, Zn	D	EDI	(1)	NG	RfD	THQ, TTHO	(4) **, (6)
[206] [207]	Cd, Cr, Cu, Fe, Ni, Pb, Zn Hg, Pb	D D	EDI -	(1) -	NG NG	RfD RfD	THQ, HI THQ, HI	(4), (6) (4) **, (6)
[18]	As, Cd, Cu, Pb, Zn	Р	EDI	(3)	NG	RfD	THQ, HI THO	(4), (6)
[208]	As, Cd, Pb, Cu Zn	Р	EDI	(1)	NG	RfD	TTHQ	(4) **, (6)
[209]	As, Pb Cd, As, Pb Cd, As, Pb	D	EDI	(3)	G NG G	SF RfD CPS	TR THQ, HI CR	(14) ** (4), (6) (14)
[210]	As, Pb, Cd	Р	EDI	(1)	NG	PTDI	%HQ	(5)
[211]	Fe, Zn, Mn, Cu, Pb, Cr, As, Co, Ni, Cd, Hg	Р	CDI	(3)	NG	RfD	THQ, TTHQ	(4), (6)
	Pb, As, Ni, Co, Cd				G	CSF	CR THO	(14)
[212]	Cd, Pb, As, Hg, Cr	D	EDI	(1)	NG	RfD	TTHQ,	(4) **, (6)
[213]	Cd, Pb, As Cd Pb Cr Cu Ni Zn Cd Cr Pb	D	CDI	(3)	G NG	CSF RfD	CRi, CR HQ, HI CR TCR	(14), (16) (4), (6) (14), (16)
[214]	As, Al, Cd, Co, Cr, Cu, Fe, Hg,	D	EWI,	(1)	G NG	Sf RfD	THQ,	(14), (16) (4) **, (6)
	Mn, Mo, Ni, Pb, Zn As, Cd, Cr, Pb		EDI		G	CSF	CR, ILCR	(14) **, (16)
[215]	Al, As, Cd, Co, Cu, Fe, Mn, Pb, Cr(VI), Ni, Hg, Zn	D	CDI	(3)	NG	RfD	THQ, HI	(4), (6)
	As, Pb, Cr(VI), Cd, Ni				G	CSF	CR, CCR	(14), (16)
[216]	Mg, Ca, K, P, Na, Cr, Mn, Fe, Ni, Cu, Zn, Mo, As, Cd, Pb	D	EDD	(3)	NG	RfD	HQ, HI	(4), (6)
[217]	Pb, Cd, Cr Ni	D	DIR	(1)	NG	RfD	THQ, TTHQ	(4) **, (6)
[218]	Al, As, Ca, Cd, Cr, Cu, Fe, K, Mg, Mn, Ni, Pb, Zn	D	EDI	(1)	NG	RfD	THQ, HI	(4) **, (6)
[219]	Cd, Cu, Cr, Pb	D	EDI	(2)	NG	RfD	HQ, HI	(4), (6)
[220]	Cd, Cr, Pb As, Cd, Cr, Hg, Pb As	D	-	-	G NG G	CSF RfD SF	CR, TCR HQ, HI CR	(14), (16) (4) **, (6) (14)
[221]	As, Cd, Hg, Pb	Р	EDI	(3)	NG	RfD	THQ, TTHO	(4), (6)
	As				G	CSF	CR	(14)

Table 2. Cont.

Reference	Elements	D/P	Exposur	e	Damage (	Effect)	Risk Chara	cterisation
			Metric	Eq.	MOA	RV	Metric	Eq.
[222]	As, Cd, Pb, Cr, Mn, Ni, Cu, Zn	D	EDI	(1)	NG	RfD	THQ, HI	(4) **, (6)
[223]	As, Pb Cd, Pb	D	EDI	(3)	G NG	Csfo RfD	TR HQ, HI	(14) ** (4), (6)
[96]	Pb As, Pb, Cd, Cr, Cu, Fe, Hg, Ni, Pb	Р	CDI	(3)	G NG	SF ADI	ELCR HQ, HI	(14) (4), (6)
[22.4]	As, Pb	Л	EDI	(2)	G	CSF	ILCR	(15)
[224]	Cu, Zh, Cr, Ni, Cd, As, Pb Hg Cr, As Pb	D	EDI	(3)	G	CSF	CR	(4), (6) (14)
Tea, herbs ar	nd spices							
[225]	As, Cd, Pb	D	ADI	(1)	NG	RfD	HQ, HI THO	(4), (6)
[226]	As, Cd, Hg, Pb	D	EDI	(1)	NG	RfD	TTHQ, TTHQ	(4), (6)
[227]	Cd, Pb, As, Mn, Ni, Cr	D	EDI	(3)	NG	RfD	THQ, HI	(4), (6)
[228]	As, Cd, Cr, Cu, Hg, Ni, Pb, Zn	D	ADI	(3)	NG	RfD	HQ, THQ Risk,	(4), (6)
	As, Cd, Cr, Pb	_			G	SF	Riskt	(14), (16)
[229]	Cd, Cr, Cu, Pb, Zn	D	EDI	(1)	NG	RfD	THQ, HI THO	(4), (6)
[230]	As, Cd, Hg, Pb	D	EWI	(1)	NG	RfD	TTHQ,	(4) **, (6)
[001]	As, Cd, Pb	D		( <b>2</b> )	G	CSF	CR	(14) **
[231]	Cr, Co, Ni, Cu, Zn, Cd, Pb Cr, Cd	D	ADI	(3)	NG G	SF	hq, hi CR	(4), (6) (14)
Meat and me	eat products							( )
[232]	Cd, Cr, Cu, Hg, Pb, Zn	D	EDI	(1)	NG	RfD	HQ, HI	(4) **, (6)
	Cd, Pb	D	EDI and CDI	(1) and (3)	G	BMDL and CSF	MOE and ILCR	(9) and (14)
[233]	As, Fe, Cu, Zn, Mn, Ni, Sr, V, Al, Cr, Cd, Pb	D	EWI	(1)	NG	RfD	THQ, TTHQ	(4) **, (6)
	As, Cd, Cr, Pb				G	CsF	CR	(14) **
Milk and da	iry products							
[234]	As, Cd, Cr, Cu, Fe, Hg, Mn, Ni, Pb. Se. Zn	D	EDI	(1)	NG	RfD	THQ, HI	(4), (6)
[105]	Cd, Pb, Cu, Zn	D	cEDI	(1)	NG	RfD	HQ, HI	(4), (6)
Miscellaneou	15							
[235]	As, Cd, Cu, Pb, Zn As, Cd, Pb	D	D	(3)	NG G	RfD SF	HQ, HI CR	(4), (6) (11)
[236]	As, Cr, Pb, Ni, Cu, Mn	Р	EDMI	(1)	NG	RfD	THQ, TTHQ	(4) **, (6)
[007]	As, Pb	F	EDI	(1)	G	CSF	TCR	(14) **
[237]	Cr, N1, Cu, As, Cd Pb As, Pb	D	EDI	(1)	NG G	RfD CSFo	THQ TR	(4) ** (14) **
[238]	Pb. As. Cd. Hg	D	EDI	(1)	NG	RfD	THQ,	(4) **. (6)
	Δς	2	201	(-)	G	SE	TTHQ CR	(1) $(0)$ $(14)$
[239]	As, Al, Cd, Cr, Cu, Hg, Mn, Pb, Se, Zn	D	DIM	(1)	NG	RfD	THQ, HI	(1 <del>1</del> ) (4) **, (6)
[240]	As Mn Mo Co Zn Hg Pb Ni Cr Se	Р	EDI	(3)	NG	RfD	THQ, HI	(4), (6)
[241]	As, Cd, Hg, Tl, Pb, U, Cr, Mn, Fe, Ni, Cu, Zn, Se	D	EDI	(1)	NG	RfD	THQ, HI	(4), (6)
[242]	As Cu, Zn, TAs, iAs, MeHg, Se, Cd, Pb iAs	Р	ADD	(3)	G NG G	CSF RfD CSF	TR HQ, HI ILCR	(14) (4), (6) (14)

Table 2. Cont.

Reference	Elements	D/P	Exposur	Exposure		Damage (Effect)		cterisation
			Metric	Eq.	MOA	RV	Metric	Eq.
[243]	Cd, Cr, Pb, Hg, As	Р	EDI	(1)	NG	RfD	THQ, TTHQ	(4) **, (6)
Prepared dis	hes and snacks							
[244]	Cd, Pb, Cr, Mo, Co, Ni, As	Р	EDI	(1)	NG	NOAEL	MOE	(7)
Total diet stu	ıdies							
[245]	Cd, Hg, Ni, Cu, Mo, Zn	Р	-	-	NG	TDI, TWI, UL	Factor of RV	(4)
	iAs, Pb				G	BMDL	MOE	(9)
[246]	As	D	EDI	(1)	G	BMDL	MOE	(9)
[247]	As, Pb	Р	EDI	(1)	G	BMDL	MOE and POE	(9) and 13
[248]	Pb	D	DDI	(1)	G	BMDL	MOE	(9)
[249]	Hg	Р	Ι	(1) *	NG	RfD	THO	(4) **

Table 2. Cont.

\* When the exposure units are mg/day per person. \*\* When the exposure used is Equation (3).

Table 3 gives the results for mycotoxins, of which "nuts, nut products, and seeds" and "milk and dairy products" were the most studied foodstuffs, and are mainly from China and Iran (28%). The most studied age groups were adults (50.6%), followed by adolescents and children, and adults and children (around 12% in each age group). Almost all the authors opted for Equation (1) (91.5%) to calculate exposure. Equation (4) was used for non-genotoxic studies in half the cases, mainly using the TDI as the RV. For the genotoxic effect, 37% of the authors assessed the MOE (Equation (9)) and 19% the Hazard index (Equation (11)). In total, 7% assessed the cancer risk using Equation (14), and 35% assessed both the MOE and the cancer risk.

Table 3. Exposure, damage, and risk characterisation methodology used in mycotoxin studies.

Reference	Mycotoxins	D/P	Exposu	re	Damag	e (Effect)	<b>Risk Characterisation</b>	
			Metric	Eq.	MOA	RV	Metric	Eq.
Beverage								
[250]	AFT	D	Exp <sup>a</sup>	-	G	BMDL	MOE	(9)
[251]	AFT	D	EDI	(1)	G	TD50/Safety factor	HI	(11)
Cereals and I	bakery products							
[252]	FB1, FB2, FB3, DON	Р	-	-	NG	PMTDI,	%PMTDI	(5)
	ZEA					TDI	%PMTDI	(5)
[253]	OTA, FB1, FB2, DON, NIV	Р	-	-	NG	PMTDI, PTWI	HQ, HI	(4), (6)
	AFB1, AFB2, AFG1, AFM1	Р	-	-	G	BMDL	MOE, MOET	(9), (10)
[254]	AFB1	Р	CDI	(1)	G	BMDL	MOE	(9)
[255]	AF, OTA, DON	D	EDI	(1)	G	TD50/Safety factor	HI	(11)
[256]	AFB1	D	EDI	(1)	G	BMDL and PCP	MOE and CR	(9) and (14)
[257]	A EB1	р	Eve	(1)	C	BMDL and	MOE and	(0) and $(14)$
[237]	Arbi	D	Ехр	(1)	G	Pcancer	CR	(9) and (14)
[258]	DON, ZEN, OTA, TeA	D	EDI	(1)	NG	TDI, TTC	%HQ	(4)
	AME, AOH				G	TTC	HQ	(11)
[259]	AFB1, AFB2, AFG1, AFG2	D	EDI	(1)	G	TD50/Safety factor	HI	(11)
[260]	AFB1 AFB2 AFG1 AFG2	D	EDI	(1)	G	BMDL and	MOE and	(9) and (14)
[200]	111 D1/111 D2/111 01/111 02	D	LDI	(1)	G	Pcancer	Risk	()) unu (11)
[261]	ZEA, BEA, DAS, STER	D	EDI	(1)	NG	PMTDI	%HQ	(4)
	AFB1				G	AP	Risk	(14)

Reference	Mycotoxins	D/P	Exposı Metric	ıre Ea.	Dama MOA	ge (Effect) RV	Risk Characte Metric	erisation Eq.
[2(2]	A ED1	D		(1)	<u> </u>		MOE and	(0) and (14)
[262]	AFBI	D	APDI	(1)	G	BMDL and AP	cancer rate	(9) and (14)
[263]	FUM, OCHRA, DON AFT	D	EDI	(1)	NG G	IDI BMDL	HQ MOE	(4) (9)
[63]	AFB1, DON, OTA	Р	у	(1)	NG	PTDI, PMTDI	MOS	(7)
Coffee, cocoa	a, and preparations							
[264]	OTA and FB2	D		(1)	NG	HBGV	%HBGV %HBCV	(5)
	CIT, ENA, ENA1, ENB1, BEA	D	-	(1)	NG	HBGV, TTC	and %TTC	(5) and (4)
	AME				G	TTC	%TTC	(11)
[265]	AFB1, STC 21 mycotoxins	D	EDI	(1)	G NG	BMDL TDI PTWI	MOE %TDI	(9) (5)
[266]	OTA, TENT, AME, AOH, ENB,	D	וחק	(1)	NG	TDI	% TDI	(5)
[200]	ZEN	D	I DI	(1)	C	PMDI	MOE	(0)
[267]	OTA, AFS, SIG OTA	Р	CDI	(3)	G NG	TDI	HO	(9)
				( )	G	BMDL	MÕE	(9)
Eggs		-						
[268]	DON, ZEN AFB1_OTA	D	EDI	(1)	NG G	TDI BMDI	HQ MOF	(4) (9)
Fats and oils								(*)
[142]	AFB1	Р	Exp	-	G	BMDL and AP	MOE and Pr	(9) and (14)
Fruits, vegs,	and legumes							
[269]	OTA A FB1	D	DDE	(1)	NG G	PTWI BMDI	MOE MOE	(7) (9)
[270]	Patulin	D	Ι	(2)	NG	PMTD	HQ	(4)
Tea, herbs an	d spices							
[226]	AFB1, TAF	D	EDI	(1)	G	BMDL	MOE	(9) (5)
[271]	AFB1	D	PDI	(1)	G	BMDL	MOE	(3)
[272]	HT2	D	PDI	(1)	NG	TDI	HQ	(4)
	AFB1, AFB2, TAF, OTA, STE				G	BMDL	MOE, MOFT	(9), (10)
[273]	AFB1, AFB2, AFG1, AFG2	Р	ADD	(1)	G	SF	R	(14)
[274]	AFB1, AFB2, AFG1, AFG2, ZEA	D	EDI	(1)	NG	TDI	%TDI	(5)
[275]	OTA, ZEN, DON, T-2, and FB	D	ADD	(1)	NG	RfD	HQ, HI	(4), (6)
	AFT	Р	ADD	(1)	G	SF	R	(14)
Meat and me	at products	D	DE	(1)	C	RM IN	MOE	(0)
[276]	AFDI, AFDZ, AFGI, AFGZ	D	DE	(1)	G	DIVIDL	NICE	(9)
[277]	AFM1	D	EDI	(1)	NG	ISIRI	HI	(4)
				. ,	G	TD50/Safety factor	HI	(11)
[278]	AFM1	D	EDI	(1)	G	BMDL and Peancer	MOE and Risk	(9) and (14)
						BMDL and		(0) and $(11)$
[279]	AFM1	Р	DE	(1)	G	TD50/safety factor	HI and CR	(9) and (11) and (14)
						and Pcancer	HI and	
[39]	AFM1	Р	ADI	(1)	G	TD50/Safety factor	MOE and	(11) and (9) and (14)
						RMDL and	LCR	ana (14)
[280]	AFM1	D	DE	(1)	G	Pcancer and	MOE and	(9) and (14)
				、 /		TD50/Safety factor	CK and HI	and (11)

Table 3. Cont.

Reference	Mycotoxins	D/P	Exposi Metric	ıre Ea	Damaş MOA	ge (Effect) RV	Risk Characterisation Metric Eq.	
[281]	AFM1	р	EDI	(1)	NG	Т	Н	(4)
[201]	AFM1	1		(1)	G	TD50/Safety factor and Pcancer	HI and HCC	(1) (11) and (14)
[282]	AFM1	D	EDI	(1)	G	BMDL and AP	MOE and risk	(9) and (14)
[283]	OTA AFM1 AFB1	D	EDI	(1)	NG G G	PTWI TD50/Safety factor BMDI	%PTWI HI MOF	(5) (11) (9)
[284]	AFM1	D	EDI	(1)	G	TD50/Safety factor	HI	(11)
[285]	AFB1	D	EDI	(1) *	G	BMDL and Pcancer	MOE and CR	(9) and (14)
[40]	AFM1	Р	EDI	(1)	G	BMDL and Pcancer and TD50/Safety factor	MOE and HCCrisk and HI	(9) and (14) and (11)
Miscellaneou	IS							
[286]	OTA, OTB, FB1, FB2 AFB1 DON 34DON 154DON T-2	D	EDI	(1)	NG G	TDI, TWI BMDL	EDI/TDI MOE	(4) (9)
[287]	HT-2, NEO, NIV, ZEA, ENNB, ENNB1, ENNA, ENNA1, BEA,	D	PDI	(1)	NG	TDI	%TDI	(5)
[288]	OTA, FB1, ZEN	D	Exp	(1)	NG	PMTDI, PMTWI	%HBGV	(4)
	AFB1				G	BMDL and AP	MOE and PR	(9) and (14)
[289]	26 mycotoxins	D	APDI	(1)	G	BMDL and AP	MOE and LCR	(9) and (14)
[290]	AFB1, AFB2, AFG1, AFG2, AFT	D	PDI	(1)	G	BMDL	MOE	(9)
[291]	OTA	D	ADD	(3)	NG	RfD	HQ	(4)
[292]	AFM1, AFT	D	DE	(1)	G	BMDL and Pcancer	MOE and R	(9) and (14)
[293] [294]	AFB1, AFB2, AFG1, AFG2 AFB1, MCLR	D P	EDI DI	(1) (1)	G G	BMDL Toxicity factor	MOE ORP	(9) (9)
[295]	AFB1	D	EDI	(1)	G	BMDL and Pcancer	MOE and CR	(9) and (14)
[296]	FBs, OTA AFB1, AFT, BEA, CIT	D D	PDI PDI	(1) (1)	NG G	TDI, TWI BMDL	HBGV MOE	(5) (9)
Nuts, nuts p	roducts and seeds							
[297]	AFB1	D	PDI	(1)	G	BMDL and AP	MOE and PR	(9) and (14)
[298]	OTA, FB1, FB2, ZEA, DON, 15AC-DON, 3AC-DON, T-2 HT-2	D	EDI	(1)	NG	TDI, PMTDI, PTWI	%TDI	(5)
[299]	AFB1, AFB2, AFG1, AFG2	D	ADD	(3)	G	BMDL	MOE	(9)
[300]	BEAU, CPA AFB1, AFT	D	PDI	(1)	NG G	TDI BMDL	%TDI MOE	(5) (9)
[301]	AFB1, AFB2, AFG1, AFG2,	D	EDI	(1)	NG	RfD	HQ	(4)
	AFB1, AFB2, AFG1, AFG2, AFT, STC				G	BMDL and AP	MOE and LCR	(9) and (14)
[302]	AFT	D	EDI	(1)	G	BMDL and AP	MOE and CR	(9) and (14)
[303]	AFT	Р	DE	(1)	G	BMDL and Cancer potency	MOE and PR	(9) and (14)

Table 3. Cont.

Reference	Mycotoxins	D/P	Exposu	ıre	Damaş	ge (Effect)	Risk Characte	erisation
	2		Metric	Eq.	MOA	RV	Metric	Eq.
Prepared dis	hes and snacks							
[304]	AFB1, AFT, OTA	D	DE	(1)	G	BMDL	MOE	(9)
	AFB1, AFB2, AFG1, AFG2,							
[305]	OTA, ZEA, FB1, FB2, FUS,	D	EDI	(1)	NG	TDI	%EDI-TDI	(5)
	BEA, ENB, ENB1, ENA, ENA1							
Total diet stu	dies							
[306]	OTA	D	Exp	(1)	NG	PTWI	MOS	(7)
[307]	AFB1	Р	EDI	(1)	G	BMDL, T25 and Pcancer	MOE and CR	(9) and (14)

Table 3. Cont.

\* When the exposure units are mg/day per person. <sup>a</sup> Exp = Exposure.

Table 4 gives the results for acrylamide. The most often studied products were "cereals and bakery products" and "prepared dishes and snacks". Most publications come from Iran (28.6%), followed by Lebanon and China (both with 9.52%). Adults were the most studied population group (38%), followed by adolescents and children (31%). Equation (1) was the most frequently used equation for exposure (71.4%). Equation (4) was used for 76.9% of the cases to characterise the non-genotoxic risk with RfD (69%) as the RV. The MOE was chosen for around 61.5% of the cases, and the cancer risk (Equation (14)) in 26.9% for the genotoxic effects.

Table 4. Exposure, damage, and risk characterisation methodology used in acrylamide studies.

Reference	D/P	Exposur	e	Damag	e (Effect)	Risk Characterisa	ation
		Metric	Eq.	MOA	RV	Metric	Eq.
Cereals and bakery products							
[308]	D	CDI	(3)	NG	RfD	THQ	(4)
				G	OSF	CR	(14)
[309]	D	CDI	(3)	NG	RfD	THQ	(4)
				G	CSF	ILCR	(14)
[310]	Р	CDI	(3)	NG	RfD	THQ	(4)
				G	BMDL and CSF	MOE and ILCR	(9) and (14)
Coffee, cocoa and preparations							
[311]	D	Y	(1)	NG	RfD	MOE	(7)
				G	BMDL	MOE	(9)
[312]	Р	EDI	(2)	G	BMDL and PF	MOE and Risk	(9) and (14)
Fruits, vegs and legumes							
[313]	D	DE	(1)	G	BMDL	MOE	(9)
Tea herbs and spices							
[314]	D	EDI	(1)	G	BMDL	MOE	(9)
			(-)				(* )
Meat and meat products	р	CDI	(2)	NC	D		(A) $(C)$
[315]	P	CDI	(3)	NG C	KID SE	InQ, ni UCP	(4), (6)
				G	56	ILCK	(14)
Miscellaneous	-	-					<u></u>
[316]	Р	E	(1)	NG	BMDL	MOE	(7)
[04]	5		(4)	G	BMDL	MOE	(9)
[317]	D	ADD	(1)	NG	RfD	HI	(4)
[210]	р	Б	(1)	G	CPS PMDI	IK MOE	(14)
[318]	D	E DE	(1) (1)	G	BMDI	MOE	(9)
[320]	r D		(1)	C	SE		(2) (14)
[320]	P		(1)	G	BMDI	MOF	(14)
[322]	D	DE	(1)	G	BMDL	MOF	(9)
[044]	D		(1)	0	DITIDL	MUCL	(-)

Reference	D/P	Exposur	e	Damag	e (Effect)	Risk Characterisa	<b>Risk Characterisation</b>	
		Metric	Eq.	MOA	RV	Metric	Eq.	
[323]	Р	DE	(1)	NG	RfD	THQ	(4)	
				G	BMDL and SF	MOE and ILCR	(9) and (14) **	
[324]	Р	CDI	(3)	NG	RfD	THQ	(4)	
				G	CSF	ILCR	(14)	
[325]	D	Exp	(1)	NG	RP	RPQ, RPI	(4), (6)	
[326]	D	Exp	(1)	G	BMDL	MOE	(9)	
Prepared dishes and snacks								
[327]	D	EDI	(1)	NG	TDI	HQ	(4)	
				G	BMDL	MOE	(9)	
[328]	D	DE	(1)	G	BMDL	MOE	(9)	
[329]	D	DI	(1)	G	BMDL	MOE	(9)	
[330]	D	EDI	-	NG	TDI	MOE	(7)	
				G	BMDL, T25	MOE	(9)	
[244]	Р	DI	(1)	NG	NOAEL	MOE	(7)	
				G	BMDL	MOE	(9)	
[331]	Р	EWI	(1)	NG	TWI	MOE	(7)	
		Exp	(1)	G	BMDL	MOE	(9)	
Total diet studies								
[332]	D	D	(1)	NG	NOAEL	MOE	(7)	
				G	BMDL	MOE	(9)	
[333]	D	EDI	(1)	G	CSF	ELCR	(14) **	
[334]	D	Е	(1)	NG	RfD		(5)	
				G	BMDL	MOE	(9)	

Table 4. Cont.

\*\* When the exposure used is Equation (3).

Table 5 gives the PAH results. Most of the authors dealt with "fish and shellfish". Iran (16%), Nigeria (14.3%), and China (12.5%) were the main countries of publication, while the adult population was the most studied group (65.5%). When calculating exposure, 68% of the studies opted for Equation (1). Equation (4) was applied to characterise the risk of non-genotoxic effects in all cases, using the RfD as RV in 93%. Equations (14) and (9), for genotoxic effects, were used in percentages of 43% and 29.4%, respectively.

Table 5. Exposure, damage, and risk characterisation methodology used in PAH studies.

Reference	РАН	D/P	Exposi	ure Fa	Damaş MO4	ge (Effect) RV	Risk Characteris	ation Fa
			wittitte	LA	MOA	KV	Methe	Lq.
Beekeeping p	products							
[126]	BaP, 4PAH, 8PAH, 16PAH and BaPeq	D	CDI	(3)	G	BMDL and CSF	MOE and ILCR	(9) and (14)
Cereals and l	bakery products							
[335]	BaP, 2PAH, 4PAH and 8PAH	D	EDI	(1)	G	BMDL	MOE	(9)
[136]	8PAH and BaPeq	D	$E_D$	(1) *	G	SF	ILCR	(14) **
Fats and oils								
[336]	15PAH, 7PAH and BaPeq	D	-	-	G	SF	ILCR	(14) **
[337]	4PAH and BaPeq	D	DE	(1)	G	BMDL	MOE	(9)
[338]	16PAH and BaPeq	Р	ED	(3)	G	SF	ILCR	(14)
[339]	15PAH and BaPeq	D	EDI	(1) *	G	SF	ILCR	(14) **
[340]	13PAH and BaPeq	Р	CDI	(3)	G	BMDL and SF	MOE and ILCR	(9) and (14)
Fish and she	llfish							
[341]	6PAH	D	EDDI	(1)	NG	RfD	THQ, HI	(4) **, (6)
	11PAH and BaPeq				G	SV and Q	HR and ILCR	(12) and (14) **

Reference	РАН	D/P	Exposu	ıre	Dama	ge (Effect)	Risk Characteris	ation
			Metric	Eq.	MOA	RV	Metric	Eq.
[342]	8PAH	D	-	-	NG	RfD	HI	(4) **
[0.40]	7PAH and BaPeq	D	E <sub>D</sub>	(1) *	G	SF	ILCR	(14) **
[343]	8PAH 7Da h	D	ADD	(2)	NG	KID SV and CSE	HQ, HI HR and Biole	(4), (6) (12) and (14)
[344]	ЛГАП 4РАН	Р	Exposu	ura61)	G	SV and CSF BMDI	MOF	(12) and $(14)$
[151]	16PAH	D	- -	-	G	LD50	HI	(11)
[345]	7PAH	D	EDI	(1)	NG	RfD	THQ	(4) **
	16PAH, BaPeq		EDI	(1)	G	CSF	CR	(14) **
[346]	BaP, 4PAH	D	DDE	(1)	G	BMDL	MOE	(9)
[347]	2PAH, 4PAH and 8PAH	D	EDI	(1)	G	BMDL	MOE	(9)
[348]	7 PAH	Р	EDI	(1)	NG	RfD	HQ	(4)
	2PAH, 4PAH, 8PAH		-	-	G	BMDL and SF	MOE and ILCR	(9) and (14) **
[349]	8PAH and BaPeq	D	CDI	(3)	G	BMDL	MOE	(9)
[350]	16PAH and BaPeq	D	-	-	G	SV and CSF	HR and ILCR	(12)
[351]	BaP and 4PAH	D	T	(1)	G	BMDL	MOE	and (14) ** (9)
[352]	7PAH	D	-	-	NG	RfD	THO, HI	(4) **, (6)
[353]	BaPeq, 7PAH	D	EDI	(1)	NG	RfD	HQ	(4)
	-				G	OSF	CR	(14) **
[354]	4PAH, 8PAH, 16PAH	D	DDI	(1) *	G	BMDL and CSF	MOE and ILCR	(9) and (14) **
[355]	апа багед 6РАН	D	T	(1)	NG	RfD	ТНО	(4)
[000]	6PAH	D	1	(1)	G	SF	TR	(14)
[356]	39PAH and BaPeq	Р	CDI	(1)	NG	RfD	THQ, HI	(4) **, (6)
	-		EDI	(1)	G	SV and SF	HR and ILCR	(12)
[257]	4DALL and DaDag	р		(-)	C	CV and CE		and $(14)^{**}$
[357]	4PAH and BaPeq	D	DDI FDI	$(1)^{n}$	G NC	SV and SF	HK and ECK	(12) and $(16)$
[550]	7PAH and BaPeq	D	LDI	(1)	G	CSE	CR	(14) **
[359]	8PAH	D	-	-	NG	RfD	HQ	(4)
	16PAH andBaPeq				G	CSF	ILCR	(14) **
[360]	16PAH and BaPeq	D	-	-	G	SF	ILCR	(14) **
Fruits, vegs a	and legumes							
[96]	16PAH and BaPeq	Р	CDI	(3)	G	BMDL and CSF	MOE and ILCR	(9) and (14)
[361]	15PAH and BaPeq	D	ADD	(3)	G	CSF	ILCR	(15)
Tea, herbs an	nd spices							(1 4) ** 1
[362]	16PAH	D	LADD	(3)	G	SF	RI, ΣRI	$(14)^{11}$ and $(16)$
[363]	15PAH and BaPeq	Р	-	-	G	CSF	R	(14) **
[364]	BaP, 2PAH and 4PAH	Р	-	-	G	BMDL	MOE	(9)
Meat and me	eat products	D	EDI	(1)	6		MOL	
[365]	BaP, 4PAH and 8PAH	D	EDI	(1) *	G	BMDL	MOE	(9)
[367]	BaPeg	D		$(1)^{-1}$	G	SF CSF	CR	(14) (14)
[368]	BaP and 4PAH	D	Exposu	(3) (ré1)	G	BMDL	MOE	(9)
Milk and dai	ry products		1	( )				( )
[369]	7PAH	Р	ADD	(1)	NG	RfD	HQ, THO	(4), (6)
[]	16PAH and BaPeq		CDI	(3)	G	CSF	ILCR	(14)
[370]	7PAH	Р	ADD	(1)	NG	RfD	HQ, THQ	(4), (6)
_	16PAH and BaPeq		CDI <sub>BaP</sub>	(3)	G	CSF	ILCR	(14)
[371]	16PAH and BaPeq	Р	ED	(3)	G	BMDL and SF	MOE and ILCR	(9) and (14)
Miscellaneou	15	-	-		~			(
[372]	16PAH and BaPeq	Р	ED	(1) *	G	SF	ILCR	(15)

Table 5. Cont.

Reference	РАН	D/P	Expos	are	Damag	ge (Effect)	Risk Characteris	ation
			Metric	Eq.	MOA	RV	Metric	Eq.
[373]	2PAH, 4PAH, BaP and BaPeq	Р	EDI and E <sub>D</sub>	(1) and (1) *	G	BMDL and CSF	MOE and ILCR	(9) and (14) **
[374]	4PAH and BaPeq	D	EDI	(1)	G	BMDL	MOE	(9)
[375]	8PAH and BaPeq	D	DI	(1)	G	BMDL	MOE	(9)
[376]	8PAH and BaPeq	D	CDI	(3)	G	BMDL and BaP's cancer risk	MOE and ILCR	(9) and (14)
[377]	16PAH and BaPeq	Р	Ι	(1)	G	SF	CR	(14) **
[378]	4PAH, 16PAH and BaPeq	D	CDI	(3)	G	BMDL, BaP's cancer risk	MOE and ILCR	(9) and (14)
[379]	7PAH	Р	ADD	(1)	NG	RfD	HQ, THQ	(4), (6)
	16PAH, BaPeq	D	CDI	(3)	G	CSF <sub>BaP</sub>	ILCR, ILCR <sub>act</sub>	(14), (16)
[380]	BaP, 2PAH, 4PAH and 8PAH	D	DI	(1)	G	BMDL	MOE	(9)
[381]	16PAH and BaPeq	D	CDI	(3)	G	SF	CR	(14)
Total diet stu	Idies							
[334]	4PAH	D	E	(1)	G	BMDL	MOE	(9)

Table 5. Cont.

\* When the exposure units are mg/day per person. \*\* When the exposure used is Equation (3).

#### 3.2.1. Exposure Assessment Equations and Nomenclature

Equation (1) was the formulation most used by the authors reviewed. Exposure was usually expressed as mg/kgBw/day, although the authors also used mg/day per person [382]. This second possibility is indicated in Tables 1–5 with an asterisk next to the equation number, i.e., (1\*).

Exposure 
$$(mg/kgBw/day) = \frac{\text{Concentration of chemical in food } (mg/kg) \times \text{Food consumption}\left(\frac{kg}{day}\right)}{\text{Body weight } (kgBw)}$$
 (1)

The standard terminology (EHC 240) should be used for the consistent understanding and application of exposure. In this framework, Table 6 provides the different names used to designate exposure, while Table 7 gives the terminology used for the parameters in the equations, i.e., the concentration of the chemical hazard present in the food and food consumption, respectively.

Table 6. Abbreviations and description of the terminology for exposure.

Parameter	Description	Parameter	Description
ADD	Average daily dose	EADI	Estimated average daily intake
ADI	Average daily intake	E <sub>D</sub>	Daily dietary exposure
AFE	Average food exposure	EDD	Estimated dietary doses
APDI	Average probable daily intake	EDDI	Estimated dietary daily intake
cEDI	Aggregated exposure	EDI	Estimated daily intake
CDI	Chronic daily intake	EDMI	Daily metal intakes
D	Total daily exposure	EWI	Estimated weekly intake
DC	Daily consumption	Exp	Exposure
DDE	Daily dietary exposure	Exposure	Dietary exposure
DDI	Dietary daily intake	I	Intake
DE	Dietary exposure	IEDI	International estimated daily intake
DED	Daily exposure dose	LADD	Lifetime average daily intake
DI	Dietary intake	NEDI	National estimated daily intake
DIM	Daily intake of metals	PDI	Probable daily intake
DIR	Daily intake rates	PEC	Potency equivalent concentration
E	Exposure/Total daily exposure	Y	Daily intake

Parameter	Concentration Description	Parameter	Food Consumption Description
С	Concentration of chemical pollutants/residual concentration	AC	Average daily food consumption
СМ	Average concentration	С	Estimated consumption of commodity
Cr	Concentration	CR	Consumption rate
R	Food pesticide residues	D	Daily intake
RC	Average residue concentration	F	Consumption of food/daily food consumption/food consumption rate
RL	Residue level	FER	Food eating rate
RLi	Occurrence of each residue	Fi	Average food consumption
STMR	Standard test of residual values	FIR	Daily intake
		Ι	Ingestion
		IR	Ingestion rate
		VIR	Daily ingested vegetable rate
		W	Average daily consumption

 Table 7. Abbreviations and description of concentrations and food consumption.

Some authors adapt the calculation of daily exposure by adding an adaptation factor to Equation (1) to convert it to Equation (2). This factor is intended to simulate, for example, the possible effect of process conditions on the variations in pesticide concentration present in food [59,72,75,90,383].

$$Exposure (mg/kgBw/day) = \frac{Concentration of chemical in food  $\binom{mg}{kg} \times Food \text{ consumption } \binom{kg}{day} \times Adapting \text{ factor}}{Body \text{ weight } (kgBw)}$ (2)$$

The US Environmental Protection Agency recommends Equation (3) to estimate the average daily potential dose of an ingested contaminant through the consumption of food, water, soil, and dust [384]. In this case, the average daily exposure to a contaminant is the result of the total ingested concentration, measured in units of mass or volume per food, for example, in mg or L per kg food; the ingestion rate, e.g., the amount of contaminated food ingested by an individual during a given period expressed in units of mass or volume per unit time, such as kg/day or L/day; the duration of exposure or the amount of time an individual is exposed to the contaminant (e.g., years); and the frequency of exposure, e.g., in days per year, all divided by the average exposure time (e.g., days) and body weight (kgBw).

$$Exposure (mg/kgBw/day) = Chemical concentration {mg \ kg} \times Intake rate {kg \ day} \times Exposure frequency {days \ year} \times Exposure duration (years)$$
(3)

Body weight (kgBw)×Average time (days)

3.2.2. Risk Characterisation Equations and Nomenclature

Non-genotoxic chemical hazards

The risk ratio is the formula applied to characterise chemicals with a threshold level. The ratio is obtained by dividing the potential exposure to a non-genotoxic chemical hazard by the reference value at which no adverse effects are expected. The result is numerical but dimensionless and is considered a negligible risk when the value obtained is less than one. However, the ratio is also commonly expressed as a percentage obtained by multiplying the numerator of Equation (4) by 100. The risk ratio is given various names in the scientific literature (see Table 8).

$$Risk ratio (non - dimensional) = \frac{Exposure (mg/kgBw/day)}{Reference Value (mg/kgBw/day)}$$
(4)

Parameter	Description	Parameter	Description
cHI	Consumer health risk	MOE	Margin of exposure
HHI	Health hazard index	MOS	Margin of safety
HI	Hazard index	R	Risk
HQ	Hazard quotation/hazard quotient	RQ	Risk quotient/Risk of ingestion
HRI	Hazard risk index/Health risk index	%RV *	Percentage of a reference value
IFS	Index of food safety	THQ	Target hazard quotient
	5	TS	Toxicity score

Table 8. Abbreviations and description of risk ratio.

\* E.g., % ADI, when RV is ADI.

Equation (5) is a particular case of Equation (4), where the RV is a HBGV (e.g., ADI, TDI, PTWI, etc.) and the chronic risk is expressed as a percentage (%HBGV).

$$%HBGV (non - dimensional) = \frac{Exposure (mg/kgBw/day)}{HBGV (mg/kgBw/day)} \times 100$$
(5)

As the risk assessment of a single chemical is known to be insufficient, when the chemicals considered have the same MOA, the cumulative effect of multiple chemicals and multiple via routes should be considered, Equation (6) [121,385]. Table 9 gives the different terms used to define the cumulative risk.

Cumulative risk (non – dimensional) = 
$$\sum_{i=1}^{N} Risk \ ratio(i)$$
 (6)

Table 9. Abbreviations and descriptions used for cumulative risk.

Parameter	Description	Parameter	Description			
cHI	Cumulative hazard index	THQ	Total hazard quotation			
HI	Hazard index	TTHQ	Total target hazard quotient			

The ratio for assessing the risk of non-genotoxic effects can also be obtained by calculating the margin of safety (MOS) (Equation (7)), where for similarity with the MOE equation, some authors use denomination MOE instead of MOS for non-genotoxic hazards [244,316].

$$Margin of Safety (non - dimensional) = \frac{Reference value (mg/kgBw/day)}{Exposure (mg/kgBw/day)}$$
(7)

Doménech and Martorell [16] proposed the probabilistic safety margin (p\_FSM) (Equation (8)), which represents the probability of exposure to a hazard *i* exceeding the safety limit (herein the ADIi), although this formulation can be extended to other RVs. The value obtained from this metric also lies between zero and one, so that a value close to one indicates a wide margin, i.e., exposure to this hazard is very unlikely to have consequences for health, while a margin close to zero implies a strong probability of a non-genotoxic adverse effect.

$$p_{FSM(H_i)} (non-dimensional) = Pr\{HQ(H_i) < 1\} = \int_0^{ADI_i} EDI(H_i) dH = 1 - \int_{ADI_i}^{\infty} EDI(H_i) dH = 1 - EP(H_i)$$
(8)

- Genotoxic chemical hazards
- (a) Ratio metrics

To support risk management in hazards with genotoxic effects, the JECFA (Joint FAO/WHO Expert Committee on Food Additives) and the EFSA (European Food Safety Authority) proposed the margin of exposure (MOE) as the indicator of the level of deserved concern, Table 10. This approach makes no implicit assumptions of a "safe" in-

take and has been more widely used to assess substances that are both genotoxic and carcinogenic [14,31,386,387]. The MOE is quantified as the ratio between a defined RV for the adverse effect on the dose–response curve—generally, the BMDL% is related to a percentage increase in the response—and human exposure (Equation (9)) [388]. Equation (10) is used to assess the combined effect of substances with the same MOA.

$$Margin of exposure (non - dimensional) = \frac{Reference value (mg/kgBw/day)}{Exposure (mg/kgBw/day)}$$
(9)

$$MOE_{T}(non - dimensional) = \frac{1}{\frac{1}{MOE1} + \frac{1}{MOE2} + \frac{1}{MOE3} + \dots}$$
(10)

Alternatively, some authors use the hazard index metric or risk quotient for genotoxic effects (Equation (11)). This equation is very similar to the risk ratio (Equation (4)) proposed for non-genotoxic effects. In this case, the exposure is divided by a genotoxic reference value such as TD50/U.

$$Hazard index (non - dimensional) = \frac{Exposure (mg/kgbw/day)}{Genotoxic reference value (mg/kgbw/day)}$$
(11)

The hazard ratio or the excess cancer risk can also be calculated to assess the margin for genotoxic hazards (Equation (12)). It can be estimated in terms of the incremental probability of developing cancer over a lifetime of total exposure to a potential carcinogen to humans. The cancer benchmark concentration is calculated by dividing the maximum acceptable risk level  $(1 \times 10^{-6})$  by the slope factor, multiplying the value obtained by the body weight, and dividing this result by the consumption [57].

$$Hazard ratio (non - dimensional) = \frac{Exposure (mg/kgbw/day)}{Cancer benchmark concentration (mg/kgbw/day)}$$
(12)

For the PAHs, the authors adapted Equation (12) through changing the exposure for the potency equivalent concentration values and dividing this value by the screening value (SV), calculated in the same way as the cancer benchmark concentration [343].

The POE is a complementary metric that represents the probability of the dose of exposure to a carcinogenic hazard exceeding the benchmark [247]. It is a measure of the probability—and, therefore, is dimensionless—of the change in the population's response exceeding the predefined reference value. It could also be interpreted as the fraction of the total population exposed to an increased risk. One of the main advantages of the POE metric is that it considers the entire exposure distribution, represented with f(E) in Equation (13).

POE (non - dimensional) = Pr(Exposure > Reference value) = 
$$\int_{RV}^{\infty} f(E) dE$$
 (13)

The POE metric is thus especially appropriate for characterising public health risks when the distribution of the exposure to a hazard with a genotoxic effect is positively skewed, and thus helps draw risk-informed conclusions, or for example if the MOE is below 10,000.

(b) Risk metrics

Different terms are used to estimate the cancer risks (see Table 10). This metric assesses the potential risk associated with exposure to carcinogens over a lifetime. It is obtained by multiplying the exposure by a slope factor (see Equation (14)). The slope factor is a toxicity value that quantifies a linear dose–response per mg/kgBw/day exposed, which is referenced to the abbreviations given in Table 11.

Cancer risk (non\_dimensional) = Exposure  $(mg/kgBw/day) \times slope factor (mg/kgBw/day)^{-1}$  (14)

Parameter	Description	Parameter	Description			
AC	Annual number of cases	MOE	Margin of exposure			
CR	Cancer risk	MOET	Total margin of exposure			
CCR	Cumulative cancer risk	ORP	Overall risk probability			
ECR	Excess cancer risk	POE	Probability of exceedance			
ELCR	Excess lifetime cancer risk	PR	Population risk			
HCC	Risk of hepatocellular carcinoma	R	Risk			
HQ	Hazard quotation	Risk	Risk of cancer			
HR	Hazard ratio	TCR	Total cancer risk/ Target cancer risk			
ILCR	Incremental lifetime cancer risk	THQcarcinogenic	Target hazard quotient for carcinogenic risk			
LCR	Lifetime cancer risk	TR	Total risk			
LTCR	Lifetime cancer risk					

Table 10. Abbreviations and description for genotoxic effect.

Table 11. Abbreviations and description of slope factor.

Parameter	Description	Parameter	Description
AP	Average potency	CSF	Cancer slope factor
BMC	Cancer benchmark concentration	OSF	Oral slope factor
CBC	Cancer benchmark concentration	Pcancer	Carcinogenic potency
CFS	Cancer slope factor	PCP	Population cancer potency
CPF	Cancer potency factor	Q	Oncogenic potency/BaP carcinogenic potency
CPSo	Oral cancer slope factor	SF	Slope Factor
CPS	Carcinogenic potency slope oral	PF	Potency factor

In the particular case of mycotoxins, the slope factor, also named carcinogenic potency (Pcancer), population cancer potency (PCP), or average potency (AP) were derived from a model with epidemiological data of individuals exposed to AFB1 testing positive for the hepatitis B surface antigen (HBsAg+) and testing negative for the hepatitis B surface antigen (HBsAg-) [389,390].

Some authors introduce an age-dependent adjustment factor to represent the increase in the probability of cancer from oral exposure, according to the population group, generally considered to be three for children and one for adults (see Equation (15)). Finally, the total cancer risk was assessed using Equation (16).

Cancer risk (non - dimensional) =Exposure (mg/kgBw/day)×Slope factor (mg/kgBw/day)<sup>-1</sup>×Age\_dependent factor (15)

Total cancer risk (non – dimensional) = 
$$\sum_{i=1}^{N}$$
 Cancer risk (i) (16)

3.2.3. Harmonisation of the Terminology and Formulation

This work has shown how different authors use different terms and equations to define and formulate some of the parameters related to risk assessments. This section presents a proposal for terminology and formulation, considering the publications and guidelines recommended by some of the leading international organizations and their frequency of use in the literature review.

The term most frequently used for exposure by the FDA [391], JECFA [392], EFSA [393] and CAC [394] is the EDI (estimated daily intake). As shown in Figure 3A, it is also the most reported term by the authors (59%), so it seems appropriate to recommend EDI for exposure. Concerning its formulation, Equation (3) is the most complete, and it is simplified to Equation (1) when the product of exposure frequency per the exposure duration is equal to the average time.



**Figure 3.** Metrics and terminology used by more than 1% of the authors reviewed: (**A**) exposure; (**B**) risk characterisation for non-genotoxic hazards; (**C**) risk characterisation for genotoxic hazards calculating a margin; and (**D**) calculating the risk.

The metric to characterise the risk of non-genotoxic hazards suggested by EFSA [393], EPA [395] and ATSDR [396] is the ratio HQ (hazard quotient), which is obtained with Equation (4). However, Figure 3B shows that 43% of the reviewed articles used the target hazard quotient (THQ), compared to 34% that used the HQ. Nevertheless, it should be noted that most papers that used the THQ were related to the study of metals, whereas the HQ was used more on the other hazards studied. Therefore, HQ is the proposed terminology for general application to chemical hazards. On the other hand, since the international organisations mentioned and 71.8% of the papers have used the hazard index (HI) to estimate the cumulative effect of hazards, it seems that there is enough consensus on the use of this term and Equation (6) for its formulation.

A margin and/or a risk can be calculated to characterise the risk of genotoxic hazards. In the first case, EFSA [386], EPA [395] and JECFA [397] among others, propose the margin of exposure (MOE), obtained with Equation (9). This formulation is also the ratio most used by the authors in the reviewed works (78%), Figure 3C. For this reason, the terms' MOE for a single hazard and MOEt for the combined effect of several hazards with a genotoxic effect are proposed to assess the margin. This study also recommends combining the MOE with the POE, as the information they provide complements each other. On the other hand, international organisations such as ATSDR [27] use the term CR (cancer risk) to measure risk. This term is the one most frequently used by the authors (67%), Figure 3D, and, therefore, the one suggested to be found using Equation (14). In turn, in Equation (14), the slope factor is mostly referenced as CSF (cancer slope factor) in the studies and, therefore, the one proposed in this section. Finally, the term TCR (total cancer risk) and Equation (16) are recommended to calculate the cumulative risk of cancer.

## 4. Discussion

Figure 4 shows the MOA considered to assess the risk characterisation for each studied hazard. In pesticides, most of the authors reviewed calculated the non-genotoxic effect (79.7%), while the study of the genotoxic and non-genotoxic effect (59.4%) was the preferred option for metals. However, only the genotoxic effect was studied for mycotoxins, acrylamide, and PAHs in more than half of the cases (57.14%, 51.8%, and 71.1%, respectively), Figure 4.



**Figure 4.** Percentage of studies considering the MOA per hazard, i.e., genotoxic, non-genotoxic, or the authors reviewed studied both effects.

The findings show that the deterministic approach is the most frequently chosen option (see Table 6). This fact may be due to the advantages of this type of approach, such as its simple calculations and speed. Personnel do not need to be experts in risk analysis, and the results are usually sufficient for internal safety management [196]. However, for the genotoxic effects, the authors who assessed the risk characterisation of acrylamide, with the ratio and the risk, and the risk of mycotoxins, opted for a probabilistic approach with a higher percentage. An equal percentage was found in manuscripts that assessed the ratio and risk of metals. These findings may be related to the fact that more and more scientific papers need better realistic estimates that consider the entire distribution of model parameters. The main limitation here is that some input variables remain fixed in practice, so probabilistic and deterministic features appear in all models.

Focusing on hazards, 66.7% of the publications calculated the pesticide risk, and 33.3% assessed the ratio, Table 12. Risk was the most used option for metals (92.9%). The ratio was calculated in 58.9% of the publications on mycotoxins, while 37.5% calculated the ratio and the risk. In acrylamide, the ratio was calculated in 61.5% of the cases. The PAH risk was assessed using risk (43.1%), ratio (31.4%), and both metrics (25.5%).

**Table 12.** Percentage of publications considering the MOA, risk characterisation metric, hazard, and approach.

	Non-G	Non-Genotoxic			Genotoxic								
Hazard	Ratio			Ratio a	Ratio and Risk		Ratio	Ratio			Risk		
	%	%D	% <b>P</b>	%	%D	% <b>P</b>	%	%D	% <b>P</b>	%	%D	%P	
Pesticides	100	83.6	16.4	-	-	-	33.3	100	-	66.7	60	40	
Metals	100	82.7	17.3	2.4	50	50	4.7	75	25	92.9	83.5	16.5	
Mycotoxins	100	81.5	18.5	37.5	71.4	28.6	58.9	83.9	16.1	7.1	25	75	
Acrylamide	100	61.5	38.5	11.5	-	100	61.5	81.3	18.8	26.9	71.4	28.6	
PAHs	100	80	13.3	25.5	61.5	38.5	31.4	87.5	12.5	43.1	72.7	27.3	
Total	100	82	18	15.4	61.5	38.5	37.3	84.7	15.3	47.4	77	23	

D = Deterministic approach and P = Probabilistic approach.

#### 5. Conclusions

The development and application of risk assessment in different scientific fields worldwide has given rise to a wide variety of terms used for the same concepts. The present work analysed the terminology and formulations gathered from the field of risk characterisation of pesticides, metals, mycotoxins, acrylamide, and polycyclic aromatic hydrocarbons (PAHs), and reached the following conclusions.

The MOA of the chemical hazard determines the formulation used in risk characterisation. The ratio between the exposure and an RV is the only mathematical model used for non-genotoxic effects. This metric provides information on the level of concern. The most used ratios are the HQ for a single hazard and the HI for the cumulative effect of several hazards. For genotoxic effects, a margin and/or a risk can be calculated to characterise the risk. In the first case, the MOE is the author's preferred metric in the literature review. Many studies highlight that using different RVs in the equations makes it difficult to compare the results. On the other hand, the metric adopted to characterise the risk of these genotoxic-chemical hazards is the cancer risk.

A deterministic approach is generally preferred to characterise risks, although differences can be found depending on hazards and metrics. Thus, a probabilistic approach is mainly used in the acrylamide articles when risk and ratio metrics are calculated. The same was true for mycotoxin studies when only a risk metric is calculated.

Based on the results found in the publications of international organisations and researchers, there appears to be a majority consensus on the parameters of risk characterisation and their formulation. This is why authors bring the following proposal for harmonisation: (1) exposure assessment is to be referred to as EDI (estimated daily intake); (2) the risk characterisation of a single non-genotoxic hazard uses the HQ (hazard quotient) metric and the HI (hazard index) for cumulative effect; (3) when a margin is used to characterise the risk of a single genotoxic hazard, the MOE (margin of exposure) metric combined with the POE (probability of exceedance) is to be selected, and when a risk metric is used in this context, the CR (cancer risk) measure is to be adopted, which, in turn, should be obtained using the CSF (cancer slope factor); (4) wherever possible, a probabilistic approach should be adopted for risk characterisation studies to take into account the effect of uncertainties in the quantification of parameters.

**Author Contributions:** Conceptualization, S.M.; investigation, E.D.; writing—review and editing, E.D.; visualization, supervision, S.M. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

**Data Availability Statement:** No new data were created or analyzed in this study. Data sharing is not applicable to this article.

Conflicts of Interest: The authors declare no conflicts of interest.

## References

- Li, H.; Chang, Q.; Bai, R.; Lv, X.; Cao, T.; Shen, S.; Liang, S.; Liang, S.; Pang, G. Simultaneous determination and risk assessment of highly toxic pesticides in the market-sold vegetables and fruits in China: A 4-year investigational study. *Ecotoxicol. Environ. Saf.* 2021, 22115, 112428. [CrossRef]
- Thompson, L.A.; Darwish, W.S. Environmental Chemical Contaminants in Food: Review of a Global Problem. J. Toxicol. 2019, 2019, 2345283. [CrossRef]
- Arcella, D.; Cascio, C.; Gómez Ruiz, J.A. Chronic Dietary Exposure to Inorganic Arsenic. European Food Safety Authority (EFSA). 2021. Available online: https://efsa.onlinelibrary.wiley.com/doi/epdf/10.2903/j.efsa.2021.6380 (accessed on 15 February 2024).
- 4. National Cancer Institute (NCI). Acrylamide and Cancer Risk. 2017. Available online: https://www.cancer.gov/about-cancer/causes-prevention/risk/diet/acrylamide-fact-sheet (accessed on 15 February 2024).
- Mesias, M.; Delgado-Andrade, C.; Gómez-Narváez, F.; Contreras-Calderón, J.; Morales, F.J. Formation of Acrylamide and Other Heat-Induced Compounds during Panela Production. *Foods* 2020, 9, 531. [CrossRef]
- FDA U.S.—Food & Drugs Administration. Acrylamide. 2022. Available online: https://www.fda.gov/food/chemicalcontaminants-food/acrylamide (accessed on 15 February 2024).

- Picardo, M.; Filatova, D.; Nuñez, O.; Farré, M. Recent advances in the detection of natural toxins in freshwater environments, TrAC. *Trends Anal. Chem.* 2019, 112, 75–86. [CrossRef]
- Commission Regulation (EU) 2022/1370. Amending Regulation (EC) No 1881/2006 as Regards Maximum Levels of Ochratoxin A in Certain Foodstuffs. 2022. Available online: https://eur-lex.europa.eu/eli/reg/2022/1370/oj (accessed on 15 February 2024).
- World Health Organization (WHO). Pesticide Residues in Food. 2022. Available online: https://www.who.int/news-room/factsheets/detail/pesticide-residues-in-food (accessed on 15 February 2024).
- 10. European Food Safety Authority (EFSA). Food Contact Materials. 2022. Available online: https://www.efsa.europa.eu/en/topics/topic/food-contact-materials (accessed on 15 February 2024).
- 11. RASFF. Rapid Alert System Feed and Food. 2022. Available online: https://webgate.ec.europa.eu/rasff-window/screen/search (accessed on 15 February 2024).
- 12. Serra, J.A.; Doménech, E.; Escriche, I.; Martorell, S. Risk assessment and critical control points from the production perspective. *Int. J. Food Microbiol.* **1999**, *46*, 9–26. [CrossRef] [PubMed]
- Food and Agriculture Organization of the United Nations and the World Health Organization (FAO/WHO). Principles and Methods for the Risk Assessment of Chemicals in Food: Environmental Health Criteria 240; World Health Organization & Food and Agriculture Organization of the United Nations: Stuttgart, Germany, 2009.
- International Programme on Chemical Safety (IPCS). Principles for Modelling Dose-Response for the Risk Assessment of Chemicals; Environmental Health Criteria, No. 239; World Health Organization: Geneva, Switzerland, 2009. Available online: https: //www.who.int/publications/i/item/9789241572392 (accessed on 15 February 2024).
- Jensen, B.H.; Andersen, J.H.; Petersen, A.; Christensen, T. Dietary exposure assessment of Danish consumers to dithiocarbamate residues in food: A comparison of the deterministic and probabilistic approach. *Food Addit. Contam.* 2008, 2, 714–721. [CrossRef] [PubMed]
- Doménech, E.; Martorell, M. Assessment of safety margins of exposure to non-genotoxic chemical substances in food. *Food Control* 2017, 79, 1–9. [CrossRef]
- World Health Organization (WHO). Guidance Document on Evaluating and Expressing Uncertainty in Hazard Characterization. 2017. Available online: https://apps.who.int/iris/bitstream/handle/10665/259858/9789241513548-eng.pdf?sequence=1& isAllowed=y (accessed on 15 February 2024).
- 18. Guo, G.; Zhang, D.; Wang, Y. Probabilistic Human Health Risk Assessment of Heavy Metal Intake via Vegetable Consumption around Pb/Zn Smelters in Southwest China. *Int. J. Environ. Res. Public Health* **2019**, *16*, 3267. [CrossRef]
- 19. Uusitalo, L.; Lehikoinen, A.; Helle, I.; Myrberg, K. An overview of methods to evaluate uncertainty of deterministic models in decision support. Environ. *Model. Softw.* **2015**, *63*, 24–31. [CrossRef]
- Rivera-Velasquez, M.F.; Fallico, C.; Guerra, I.; Straface, S.A. Comparison of deterministic and probabilistic approaches for assessing risks from contaminated aquifers: An Italian case study. *Waste Manag. Res.* 2013, *31*, 1245–1254. [CrossRef] [PubMed]
- 21. Kirchsteiger, C. On the use of probabilistic and deterministic methods in risk analysis. *J. Loss Prev. Process Ind.* **1999**, *12*, 399–419. [CrossRef]
- 22. Moher, D.; Liberati, A.; Tetzlaff, J.; Altman, D.G.; The PRISMA Group. Preferred reporting items for systematic reviews and meta-analyses: The PRISMA statement. *J. Chin. Integr. Med.* 2009, 7, 889–896. [CrossRef]
- Higgins, J.P.; Thomas, J.; Chandler, J.; Cumpston, M.; Li, T.; Page, M.J.; Welch, V.A. Cochrane Handbook for Systematic Reviews of Interventions; John Wiley & Sons: Hoboken, NJ, USA, 2019.
- Codex Alimentarius Commission (CAC). Principles and Guidelines for the Conduct of Microbiological Risk Assessment CAC/GL 30-1999. 2014. Available online: https://www.fao.org/fao-who-codexalimentarius/sh-proxy/en/?lnk=1&url=https%253A%25 2F%252Fworkspace.fao.org%252Fsites%252Fcodex%252FStandards%252FCXG+30-1999%252FCXG\_030e\_2014.pdf (accessed on 15 February 2024).
- 25. Loeb, K.R.; Loeb, L.A. Significance of multiple mutations in cancer. Carcinogenesis 2000, 21, 379–385. [CrossRef] [PubMed]
- 26. Dybing, E.; O'brien, J.; Renwick, A.G.; Sanner, T. Risk assessment of dietary exposures to compounds that are genotoxic and carcinogenic—An overview. *Toxicol. Lett.* **2008**, *180*, 110–117. [CrossRef] [PubMed]
- 27. European Food Safety Authority (EFSA). Chemical Hazards Database (OpenFoodTox). Available online: https://www.efsa. europa.eu/en/data-report/chemical-hazards-database-openfoodtox (accessed on 15 February 2024).
- Environmental Protection Agency (EPA). CompTox Chemicals Dashboard. 2022. Available online: https://www.epa.gov/ chemical-research/comptox-chemicals-dashboard (accessed on 15 February 2024).
- 29. Danish Environmental Protection Agency (DEPA). The Advisory List for Self Classification of Hazardous Substances. 2022. Available online: https://eng.mst.dk/chemicals/biocides/application-in-accordance-with-the-bpr/requirements-subsequent-to-the-decision/authorisation-is-granted/classification-and-labelling (accessed on 15 February 2024).
- Environmental Protection Agency New Zealand (EPANZ). Chemical Classification and Information Database (CCID). 2022. Available online: https://www.epa.govt.nz/database-search/chemical-classification-and-information-database-ccid/ (accessed on 15 February 2024).
- Food and Agriculture Organization of the United Nations and the World Health Organization (FAO/WHO). Food Safety Risk Analysis an Overview and Framework Manual. 2005. Available online: https://www.fsc.go.jp/sonota/foodsafety\_riskanalysis. pdf (accessed on 12 July 2023).

- Environmental Protection Agency (EPA). Benchmark Dose Technical Guidance; EPA/100/R-12/001; US Environmental Protection Agency: Washington, DC, USA, 2012. Available online: https://www.epa.gov/risk/benchmark-dose-technical-guidance (accessed on 15 February 2024).
- European Food Safety Authority (EFSA). Guidance on the Use of the Benchmark Dose Approach in Risk Assessment. Available online: https://efsa.onlinelibrary.wiley.com/doi/10.2903/j.efsa.2022.7584 (accessed on 15 February 2024).
- Haber, L.T.; Dourson, M.L.; Allen, B.C.; Hertzberg, R.C.; Parker, A.; Vincent, M.J.; Maier, A.; Boobis, A.R. Benchmark dose (BMD) modeling: Current practice, issues, and challenges. *Crit. Rev. Toxicol.* 2018, 48, 387–415. [CrossRef]
- 35. European Food Safety Authority (EFSA). Update: Use of the Benchmark Dose Approach in Risk Assessment EFSA Scientific Committee. 2017. Available online: https://efsa.onlinelibrary.wiley.com/doi/10.2903/j.efsa.2017.4658 (accessed on 15 February 2024).
- 36. Environmental Protection Agency (EPA). 1993 Reference Dose (RfD): Description and Use in Health Risk Assessments Background Document. Available online: https://www.epa.gov/iris/reference-dose-rfd-description-and-use-health-risk-assessments (accessed on 15 February 2024).
- Kroes, R.; Renwick, A.G.; Feron, V.; Galli, C.L.; Gibney, M.; Greim, H.; Guy, R.H.; Lhuguenot, J.C.; van de Sandt, J.J. Application of the threshold of toxicological concern (TTC) to the safety evaluation of cosmetic ingredients. *Food Chem. Toxicol.* 2007, 45, 2533–2562. [CrossRef]
- Cartus, A.; Schrenk, D. Current methods in risk assessment of genotoxic chemicals. *Food Chem. Toxicol.* 2017, 106, 574–582. [CrossRef]
- 39. Hasninia, D.; Salimi, G.; Bahrami, G.; Sharafi, K.; Omer, A.K.; Rezaie, M.; Kiani, A. Human health risk assessment of aflatoxin M1 in raw and pasteurized milk from the Kermanshah province, Iran. *J. Food Compos. Anal.* **2022**, *110*, 104568. [CrossRef]
- 40. Udovicki, B.; Djekic, I.; Kalogianni, E.P.; Rajkovic, A. Exposure assessment and risk characterization of Aflatoxin M1 intake through consumption of milk and yoghurt by student population in Serbia and Greece. *Toxins* **2019**, *11*, 205. [CrossRef]
- 41. Food and Agriculture Organization of the United Nations and the World Health Organization (FAO/WHO). Food Consumption and Exposure Assessment of Chemicals: Report of a FAO/WHO Consultation, Geneva, Switzerland, 10–14 February 1997. Available online: https://apps.who.int/iris/bitstream/handle/10665/63988/WHO\_FSF\_FOS\_97.5.pdf?sequence=1&isAllowed= y (accessed on 15 February 2024).
- 42. European Food Safety Authority (EFSA). Overview of the Procedures Currently Used at EFSA for the Assessment of Dietary Exposure to Different Chemical Substances. 2011. Available online: https://efsa.onlinelibrary.wiley.com/doi/pdf/10.2903/j.efsa. 2011.2490 (accessed on 15 February 2024).
- European Food Safety Authority (EFSA). Food Consumption Data. 2021. Available online: https://www.efsa.europa.eu/en/ data-report/food-consumption-data (accessed on 15 February 2024).
- 44. World Health Organization (WHO). Food Safety Collaborative Platform (CIFOCOss). 2018. Available online: https://apps.who. int/foscollab/Download/DownloadConso (accessed on 15 February 2024).
- 45. Zenodo. Food Additives Intake Model (FAIM), Version 1.1. July 2013. Available online: https://zenodo.org/record/154725 (accessed on 15 February 2024).
- 46. Ioannidou, S.; Cascio, C.; Gilsenan, M.B. European Food Safety Authority open access tools to estimate dietary exposure to food chemicals. *Environ. Int.* 2021, 149, 106357. [CrossRef]
- 47. Environmental Protection Agency (EPA). EPA ExpoBox (A Toolbox for Exposure Assessors). 2022. Available online: https://www.epa.gov/expobox (accessed on 15 February 2024).
- European Food Safety Authority (EFSA). Pesticide Evaluation: Tools. 2023. Available online: https://www.efsa.europa.eu/en/ applications/pesticides/tools (accessed on 15 February 2024).
- 49. Dorne, J.L.C.M.; Fink-Gremmels, J. Human and animal health risk assessments of chemicals in the food chain: Comparative aspects and future perspectives. *Toxicol. Appl. Pharmacol.* 2013, 270, 187–195. [CrossRef]
- 50. Abramo, G.; D'Angelo, C.A.; Mele, I. Impact of COVID-19 on research output by gender across countries. *Scientometrics* **2022**, 127, 6811–6826. [CrossRef]
- 51. Riccaboni, M.; Verginer, L. The impact of the COVID-19 pandemic on scientific research in the life sciences. *PLoS ONE* 2022, 17, e0263001. [CrossRef] [PubMed]
- Bommuraj, V.; Chen, Y.; Klein, H.; Sperling, R.; Barel, S.; Shimshoni, J.A. Pesticide and trace element residues in honey and beeswax combs from Israel in association with human risk assessment and honey adulteration. *Food Chem.* 2019, 299, 125123. [CrossRef]
- Juan-Borrás, M.; Doménech, E.; Escriche, I. Mixture-risk-assessment of pesticide residues in retail polyfloral honey. *Food Control* 2016, 67, 127–134. [CrossRef]
- Pipoyan, D.; Stepanyan, S.; Beglaryan, M.; Stepanyan, S.; Asmaryan, S.; Hovsepyan, A.; Merendino, N. Carcinogenic and non-carcinogenic risk assessment of trace elements and POPs in honey from Shirak and Syunik regions of Armenia. *Chemosphere* 2020, 239, 124809. [CrossRef]
- 55. Wang, F.; Wang, Y.; Li, Y.; Zhang, S.; Shi, P.; Li-Byarlay, H.; Luo, S. Pesticide residues in beebread and honey in *Apis cerana cerana* and their hazards to honey bees and human. *Ecotoxicol. Environ. Saf.* **2022**, 238, 113574. [CrossRef] [PubMed]
- Zafeiraki, E.; Kasiotis, K.M.; Nisianakis, P.; Manea-Karga, E.; Machera, K. Occurrence and human health risk assessment of mineral elements and pesticides residues in bee pollen. *Food Chem. Toxicol.* 2022, 161, 112826. [CrossRef] [PubMed]

- Akoto, O.; Oppong-Otoo, J.; Osei-Fosu, P. Carcinogenic and non-carcinogenic risk of organochlorine pesticide residues in processed cereal-based complementary foods for infants and young children in Ghana. *Chemosphere* 2015, 132, 193–199. [CrossRef]
- Eslami, Z.; Mahdavi, V.; Mofrad, A.A. Simultaneous multi-determination of pesticide residues in barberry: A risk assessment study. *J. Food Compos. Anal.* 2022, 110, 104576. [CrossRef]
   M. Ling, M.P., M.C., Paperil, C.L. Felens, I.L. Felens
- 59. Medina, M.B.; Munitz, M.S.; Resnik, S.L. Fate and health risks assessment of some pesticides residues during industrial rice processing in Argentina. *J. Food Compos. Anal.* **2021**, *98*, 103823. [CrossRef]
- 60. Mumtaz, M.; Qadir, A.; Mahmood, A.; Mehmood, A.; Malik, R.N.; Li, J.; Yousaf, Z.; Jamil, N.; Shaikh, I.A.; Ali, H.; et al. Human health risk assessment, congener specific analysis and spatial distribution pattern of organochlorine pesticides (OCPs) through rice crop from selected districts of Punjab Province, Pakistan. *Sci. Total Environ.* **2015**, *511*, 354–361. [CrossRef]
- Sosan, M.B.; Adeleye, A.O.; Oyekunle, J.A.O.; Udah, O.; Oloruntunbi, P.M.; Daramola, M.O.; Saka, W.T. Dietary risk assessment of organochlorine pesticide residues in maize-based complementary breakfast food products in Nigeria. *Heliyon* 2020, *6*, e05803. [CrossRef] [PubMed]
- 62. Tao, Y.; Jia, C.; Jing, J.; Zhang, J.; Yu, P.; He, M.; Wu, J.; Chen, L.; Zhao, E. Occurrence and dietary risk assessment of 37 pesticides in wheat fields in the suburbs of Beijing, China. *Food Chem.* **2021**, *350*, 129245. [CrossRef]
- 63. Yang, X.; Zhao, Z.; Tan, Y.; Chen, B.; Zhou, C.; Wu, A. Risk profiling of exposures to multiclass contaminants through cereals and cereal-based products consumption: A case study for the inhabitants in Shanghai, China. *Food Control* **2020**, *109*, 106964. [CrossRef]
- 64. Razzaghi, N.; Ziarati, P.; Rastegar, H.; Shoeibi, S.; Amirahmadi, M.; Conti, G.O.; Ferrante, M.; Fakhri, Y.; Mousavi Khaneghah, A. The concentration and probabilistic health risk assessment of pesticide residues in commercially available olive oils in Iran. *Food Chem. Toxicol.* **2018**, *120*, 32–40. [CrossRef] [PubMed]
- 65. Arisekar, U.; Shakila, R.J.; Shalini, R.; Jeyasekaran, G.; Padmavathy, P. Effect of household culinary processes on organochlorine pesticide residues (OCPs) in the seafood (*Penaeus vannamei*) and its associated human health risk assessment: Our vision and future scope. *Chemosphere* 2022, 297, 134075. [CrossRef] [PubMed]
- Eissa, F.; Ghanem, K.; Al-Sisi, M. Occurrence and human health risks of pesticides and antibiotics in Nile tilapia along the Rosetta Nile branch, Egypt. *Toxicol. Rep.* 2020, 7, 1640–1646. [CrossRef] [PubMed]
- Oliva, A.L.; Girones, L.; Recabarren-Villalón, T.V.; Ronda, A.C.; Marcovecchio, J.E.; Arias, A.H. Occurrence, behavior and the associated health risk of organochlorine pesticides in sediments and fish from Bahía Blanca Estuary, Argentina. *Mar. Pollut. Bull.* 2022, 185, 114247. [CrossRef]
- 68. Supe Tulcan, R.X.; Ouyang, W.; Gu, X.; Lin, C.; Tysklind, M.; Wang, B. Typical herbicide residues, trophic transfer, bioconcentration, and health risk of marine organisms. *Environ. Int.* 2021, 152, 106500. [CrossRef]
- 69. Adeleye, A.O.; Sosan, M.B.; Oyekunle, J.A.O. Dietary exposure assessment of organochlorine pesticides in two commonly grown leafy vegetables in South-western Nigeria. *Heliyon* **2019**, *5*, e01895. [CrossRef]
- 70. Balkan, T.; Yılmaz, Ö. Method validation, residue and risk assessment of 260 pesticides in some leafy vegetables using liquid chromatography coupled to tandem mass spectrometry. *Food Chem.* **2022**, *384*, 132516. [CrossRef] [PubMed]
- Bempah, C.K.; Agyekum, A.A.; Akuamoa, F.; Frimpong, S.; Buah-Kwofie, A. Dietary exposure to chlorinated pesticide residues in fruits and vegetables from Ghanaian markets. *J. Food Compos. Anal.* 2016, 46, 103–113. [CrossRef]
- 72. Bhandari, G.; Zomer, P.; Atreya, K.; Mol, H.G.; Yang, X.; Geissen, V. Pesticide residues in Nepalese vegetables and potential health risks. *Environ. Res.* 2019, 172, 511–521. [CrossRef] [PubMed]
- 73. Calderon, R.; García-Hernández, J.; Palma, P.; Leyva-Morales, J.B.; Zambrano-Soria, M.; Bastidas-Bastidas, P.J.; Godoy, M. Assessment of pesticide residues in vegetables commonly consumed in Chile and Mexico: Potential impacts for public health. *J. Food Compos. Anal.* **2022**, *108*, 104420. [CrossRef]
- Cámara, M.A.; Cermeño, S.; Martínez, G.; Oliva, J. Removal residues of pesticides in apricot, peach and orange processed and dietary exposure assessment. *Food Chem.* 2020, 325, 126936. [CrossRef] [PubMed]
- 75. Chen, R.; Xue, X.; Wang, G.; Wang, J. Determination and dietary intake risk assessment of 14 pesticide residues in apples of China. *Food Chem.* **2021**, *351*, 129266. [CrossRef] [PubMed]
- 76. Fang, Y.; Nie, Z.; Yang, Y.; Die, Q.; Liu, F.; He, J.; Huang, Q. Human health risk assessment of pesticide residues in market-sold vegetables and fish in a northern metropolis of China. *Environ. Sci. Pollut. Res.* **2015**, *22*, 6135–6143. [CrossRef] [PubMed]
- 77. Fu, Y.; Yang, T.; Zhao, J.; Zhang, L.; Chen, R.; Wu, Y. Determination of eight pesticides in Lycium barbarum by LC-MS/MS and dietary risk assessment. *Food Chem.* 2017, 218, 192–198. [CrossRef]
- 78. Gad Alla, S.A.; Loutfy, N.M.; Shendy, A.H.; Ahmed, M.T. Hazard index, a tool for a long term risk assessment of pesticide residues in some commodities, a pilot study. *Regul. Toxicol. Pharmacol.* **2015**, *73*, 985–991. [CrossRef]
- 79. Golge, O.; Kabak, B. Determination of 115 pesticide residues in oranges by high-performance liquid chromatography–triplequadrupole mass spectrometry in combination with QuEChERS method. *J. Food Compos. Anal.* **2015**, *41*, 86–97. [CrossRef]
- Kermani, M.; Dowlati, M.; Gholami, M.; Sobhi, H.R.; Azari, A.; Esrafili, A.; Yeganeh, M.; Ghaffari, H.R. A global systematic review, meta-analysis and health risk assessment on the quantity of Malathion, Diazinon and Chlorpyrifos in Vegetables. *Chemosphere* 2021, 270, 129382. [CrossRef] [PubMed]
- Kuang, L.; Xu, G.; Tong, Y.; Li, H.; Zhang, J.; Shen, Y.; Cheng, Y. Risk assessment of pesticide residues in Chinese litchis. J. Food Prot. 2022, 85, 98–103. [CrossRef] [PubMed]

- 82. Kumari, D.; John, S. Health risk assessment of pesticide residues in fruits and vegetables from farms and markets of Western Indian Himalayan region. *Chemosphere* **2019**, 224, 162–167. [CrossRef] [PubMed]
- Lemos, J.; Sampedro, M.C.; de Ariño, A.; Ortiz, A.; Barrio, R.J. Risk assessment of exposure to pesticides through dietary intake of vegetables typical of the Mediterranean diet in the Basque Country. J. Food Compos. Anal. 2016, 49, 35–41. [CrossRef]
- Li, Z.; Nie, J.; Lu, Z.; Xie, H.; Kang, L.; Chen, Q.; Li, A.; Zhao, X.; Xu, G.; Yan, Z. Cumulative risk assessment of the exposure to pyrethroids through fruits consumption in China—Based on a 3-year investigation. *Food Chem. Toxicol.* 2016, 96, 234–243. [CrossRef] [PubMed]
- Liang, S.X.; Li, H.; Chang, Q.; Bai, R.; Zhao, Z.; Pang, G.F. Residual levels and dietary exposure risk assessment of banned pesticides in fruits and vegetables from Chinese market based on long-term nontargeted screening by HPLC-Q-TOF/MS. Ecotoxicol. *Environ. Saf.* 2022, 248, 114280. [CrossRef]
- Lozowicka, B. Health risk for children and adults consuming apples with pesticide residue. *Sci. Total Environ.* 2000, 502, 184–198. [CrossRef]
- Oyinloye, J.A.; Oyekunle, J.A.O.; Ogunfowokan, A.O.; Msagati, T.; Adekunle, A.S.; Nety, S.S. Human health risk assessments of organochlorine pesticides in some food crops from Esa-Oke farm settlement, Osun State, Nigeria. *Heliyon* 2021, 7, e07470. [CrossRef]
- 88. Picó, Y.; El-Sheikh, M.A.; Alfarhan, A.H.; Barceló, D. Target vs non-target analysis to determine pesticide residues in fruits from Saudi Arabia and influence in potential risk associated with exposure. *Food Chem. Toxicol.* **2018**, *111*, 53–63. [CrossRef]
- 89. Ping, H.; Wang, B.; Li, C.; Li, Y.; Ha, X.; Jia, W.; Li, B.; Ma, Z. Potential health risk of pesticide residues in greenhouse vegetables under modern urban agriculture: A case study in Beijing, China. *J. Food Compos. Anal.* **2022**, *105*, 104222. [CrossRef]
- 90. Qin, G.; Chen, Y.; He, F.; Yang, B.; Zou, K.; Shen, N.; Zuo, B.; Liu, R.; Zhang, W.; Li, Y. Risk assessment of fungicide pesticide residues in vegetables and fruits in the mid-western region of China. *J. Food Compos. Anal.* **2021**, *95*, 103663. [CrossRef]
- Quijano, L.; Yusà, V.; Font, G.; Pardo, O. Chronic cumulative risk assessment of the exposure to organophosphorus, carbamate and pyrethroid and pyrethrin pesticides through fruit and vegetables consumption in the region of Valencia (Spain). *Food Chem. Toxicol.* 2016, *89*, 39–46. [CrossRef]
- 92. Shao, W.C.; Zang, Y.Y.; Ma, H.Y.; Ling, Y.E.; Kai, Z.P. Concentrations and related health risk assessment of pesticides, phthalates, and heavy metals in strawberries from Shanghai, China. J. Food Prot. 2021, 84, 2116–2122. [CrossRef]
- 93. Sharma, K.K.; Tripathy, V.; Sharma, K.; Gupta, R.; Yadav, R.; Devi, S.; Walia, S. Long-term monitoring of 155 multi-class pesticide residues in Indian vegetables and their risk assessment for consumer safety. *Food Chem.* **2022**, *373*, 131518. [CrossRef]
- 94. Sivaperumal, P.; Thasale, R.; Kumar, D.; Mehta, T.G.; Limbachiya, R. Human health risk assessment of pesticide residues in vegetable and fruit samples in Gujarat State, India. *Heliyon* **2022**, *8*, e10876.
- Sójka, M.; Miszczak, A.; Sikorski, P.; Zagibajlło, K.; Karlińska, E.; Kosmala, M. Pesticide residue levels in strawberry processing by-products that are rich in ellagitannins and an assessment of their dietary risk to consumers. NFS J. 2015, 1, 31–37. [CrossRef]
- Taghizadeh, S.F.; Rezaee, R.; Azizi, M.; Hayes, A.W.; Giesy, J.P.; Karimi, G. Pesticides, metals, and polycyclic aromatic hydrocarbons in date fruits: A probabilistic assessment of risk to health of Iranian consumers. *J. Food Compos. Anal.* 2021, 98, 103815. [CrossRef]
- 97. Tesi, G.O.; Obi-Iyeke, G.E.; Ossai, J.C.; Ogbuta, A.A.; Ogbara, E.F.; Olorunfemi, D.I.; Agbozu, I.E. Human exposure to organochlorine pesticides in vegetables from major cities in south-south Nigeria. *Chemosphere* **2022**, *303*, 135296. [CrossRef] [PubMed]
- Tripathy, V.; Sharma, K.K.; Sharma, K.; Gupta, R.; Yadav, R.; Singh, G.; Aggarwal, A.; Walia, S. Monitoring and dietary risk assessment of pesticide residues in brinjal, capsicum, tomato, and cucurbits grown in Northern and Western regions of India. *J. Food Compos. Anal.* 2022, 110, 104543. [CrossRef]
- 99. Tzatzarakis, M.; Kokkinakis, M.; Renieri, E.; Goumenou, M.; Kavvalakis, M.; Vakonaki, E.; Chatzinikolaou, A.; Stivaktakis, P.; Tsakiris, I.; Rizos, A.; et al. Multiresidue analysis of insecticides and fungicides in apples from the Greek market. Applying an alternative approach for risk assessment. *Food Chem. Toxicol.* 2020, 140, 111262. [CrossRef]
- Wang, R.; Yang, Y.; Deng, Y.; Hu, D.; Lu, P. Multiresidue analysis and dietary risk assessment of pesticides in eight minor vegetables from Guizhou, China. *Food Chem.* 2022, 380, 131863. [CrossRef]
- 101. Yu, R.; Liu, Q.; Liu, J.; Wang, Q.; Wang, Y. Concentrations of organophosphorus pesticides in fresh vegetables and related human health risk assessment in Changchun, Northeast China. *Food Control* **2016**, *60*, 353–360. [CrossRef]
- Aydin, S.; Aydin, M.E.; Beduk, F.; Ulvi, A. Organohalogenated pollutants in raw and UHT cow's milk from Turkey: A risk assessment of dietary intake. Environ. Sci. Pollut. Res. 2019, 26, 12788–12797. [CrossRef] [PubMed]
- 103. Bedi, J.S.; Gill, J.P.S.; Aulakh, R.S.; Kaur, P. Pesticide residues in bovine milk in Punjab, India: Spatial variation and risk assessment to human health. Arch. Environ. *Contam. Toxicol.* **2015**, *69*, 230–240. [CrossRef] [PubMed]
- 104. Gill, J.P.S.; Bedi, J.S.; Singh, R.; Fairoze, M.N.; Hazarika, R.A.; Gaurav, A.; Satpathy, S.M.; Chauhan, A.S.; Lindahl, J.; Grace, D.; et al. Pesticide residues in peri-urban bovine milk from India and risk assessment: A multicenter study. *Sci. Rep.* 2020, 10, 8054. [CrossRef] [PubMed]
- 105. Năstăsescu, V.; Mititelu, M.; Goumenou, M.; Docea, A.O.; Renieri, E.; Udeanu, D.I.; Oprea, E.; Arsene, A.L.; Dinu-Pîvu, C.E.; Ghica, M. Heavy metal and pesticide levels in dairy products: Evaluation of human health risk. *Food Chem. Toxicol.* 2020, 146, 111844. [CrossRef]

- 106. Ramezani, S.; Mahdavi, V.; Gordan, H.; Rezadoost, H.; Conti, G.O.; Khaneghah, A.M. Determination of multi-class pesticides residues of cow and human milk samples from Iran using UHPLC-MS/MS and GC-ECD: A probabilistic health risk assessment. *Environ. Res.* 2022, 208, 112730. [CrossRef] [PubMed]
- Raslan, A.A.; Elbadry, S.; Darwish, W.S. Estimation and human health risk assessment of organochlorine pesticides in raw milk marketed in Zagazig City, Egypt. J. Toxicol. 2018, 2018, 3821797. [CrossRef] [PubMed]
- Sana, S.; Qadir, A.; Mumtaz, M.; Evans, N.P.; Ahmad, S.R. Spatial trends and human health risks of organochlorinated pesticides from bovine milk; a case study from a developing country, Pakistan. *Chemosphere* 2021, 276, 130110. [CrossRef]
- Cui, K.; Wu, X.; Wei, D.; Zhang, Y.; Cao, J.; Xu, J.; Dong, F.; Zheng, Y. Health risks to dietary neonicotinoids are low for Chinese residents based on an analysis of 13 daily-consumed foods. *Environ. Int.* 2021, 149, 106385. [CrossRef]
- 110. Fatunsin, O.T.; Oyeyiola, A.O.; Moshood, M.O.; Akanbi, L.M.; Fadahunsi, D.E. Dietary risk assessment of organophosphate and carbamate pesticide residues in commonly eaten food crops. *Sci. Afr.* **2020**, *8*, e00442. [CrossRef]
- Galani, Y.J.H.; Houbraken, M.; Wumbei, A.; Djeugap, J.F.; Fotio, D.; Gong, Y.Y.; Spanoghe, P. Monitoring and dietary risk assessment of 81 pesticide residues in 11 local agricultural products from the 3 largest cities of Cameroon. *Food Control* 2020, 118, 107416. [CrossRef]
- 112. Jensen, B.H.; Petersen, A.; Nielsen, E.; Christensen, T.; Poulsen, M.E.; Andersen, J.H. Cumulative dietary exposure of the population of Denmark to pesticides. *Food Chem. Toxicol.* **2015**, *83*, 300–307. [CrossRef] [PubMed]
- Zhang, Q.; Xia, Z.; Wu, M.; Wang, L.; Yang, H. Human health risk assessment of DDTs and HCHs through dietary exposure in Nanjing, China. *Chemosphere* 2017, 177, 211–216. [CrossRef] [PubMed]
- 114. Cui, Y.; Xu, Z.; Tang, S.; Wang, Y.; Jiang, G. Organochlorine pesticides and other pesticides in peanut oil: Residue level, source, household processing factor and risk assessment. *J. Hazard. Mater.* **2022**, *429*, 128272. [CrossRef] [PubMed]
- 115. Liu, Y.; Shen, D.; Li, S.; Ni, Z.; Ding, M.; Ye, C.; Tang, F. Residue levels and risk assessment of pesticides in nuts of China. *Chemosphere* **2016**, *144*, 645–651. [CrossRef] [PubMed]
- Tajdar-oranj, B.; Peivasteh-roudsari, L.; Mahdavi, V.; Behbahan, A.K.; Khaneghah, A.M. Simultaneous multi-determination of pesticide residues in pistachio from Iran's market: A probabilistic health risk assessment study. *J. Food Compos. Anal.* 2021, 103, 104085. [CrossRef]
- 117. Feng, C.; Xu, Q.; Qiu, X.; Ji, J.; Lin, Y.; Le, S.; Xue, L.; Chen, Y.; She, J.; Xiao, P.; et al. Profiling of pesticides and pesticide transformation products in Chinese herbal teas. *Food Chem.* **2022**, *383*, 132431. [CrossRef]
- 118. Lu, E.H.; Huang, S.Z.; Yu, T.H.; Chiang, S.Y.; Wu, K.Y. Systematic probabilistic risk assessment of pesticide residues in tea leaves. *Chemosphere* **2020**, 247, 125692. [CrossRef]
- Mahdavi, V.; Eslami, Z.; Golmohammadi, G.; Tajdar-oranj, B.; Keikavousi Behbahan, A.; Mousavi Khaneghah, A. Simultaneous determination of multiple pesticide residues in Iranian saffron: A probabilistic health risk assessment. *J. Food Compos. Anal.* 2021, 100, 103915. [CrossRef]
- 120. Wu, Y.; An, Q.; Li, D.; Kang, L.; Zhou, C.; Zhang, J.; Pan, C. Multi-residue analytical method development and risk assessment of 56 pesticides and their metabolites in tea by chromatography tandem mass spectroscopy. *Food Chem.* **2022**, 375, 131819. [CrossRef]
- Evans, R.M.; Scholze, M.; Kortenkamp, A. Examining the feasibility of mixture risk assessment: A case study using a tiered approach with data of 67 pesticides from the Joint FAO/WHO Meeting on Pesticide Residues (JMPR). *Food Chem. Toxicol.* 2015, 84, 260–269. [CrossRef] [PubMed]
- 122. Gbadamosi, M.R.; Abdallah, M.A.E.; Harrad, S. Organophosphate esters in UK diet; exposure and risk assessment. *Sci. Total Environ.* **2022**, *849*, 158368. [CrossRef] [PubMed]
- 123. Khazaal, S.; El Darra, N.; Kobeissi, A.; Jammoul, R.; Jammoul, A. Risk assessment of pesticide residues from foods of plant origin in Lebanon. *Food Chem.* 2022, 374, 131676. [CrossRef] [PubMed]
- 124. Beshaw, T.; Demssie, K.; Leka, I. Levels and health risk assessment of trace metals in honey from different districts of Bench Sheko Zone, Southwest Ethiopia. *Heliyon* **2022**, *8*, e10535. [CrossRef]
- 125. Duru, C.E.; Duru, I.A. Phytochemical evaluation and health risk assessment of honey from an Apiary in Amizi, Ikuano local government area, Abia State, Nigeria. *Sci. Afr.* **2021**, *13*, e00885. [CrossRef]
- 126. Ozoani, H.A.; Ezejiofor, A.N.; Amadi, C.N.; Chijioke-Nwauche, I.; Orisakwe, O.E. Safety of honey consumed in Enugu State, Nigeria: A public health risk assessment of lead and polycyclic aromatic hydrocarbons. *Rocz. Państwowego Zakładu Hig.* 2020, 71, 57–66.
- 127. Scivicco, M.; Squillante, J.; Velotto, S.; Esposito, F.; Cirillo, T.; Severino, L. Dietary exposure to heavy metals through polyfloral honey from Campania region (Italy). *J. Food Compos. Anal.* **2022**, *114*, 104748. [CrossRef]
- 128. Deka, A.K.; Handique, P.; Deka, D.C. Ethnic food beverages with heavy metal contents: Parameters for associated risk to human health, North-East India. *Toxicol. Rep.* 2021, *8*, 1220–1225. [CrossRef]
- 129. Al-Saleh, I.; Abduljabbar, M. Heavy metals (lead, cadmium, methylmercury, arsenic) in commonly imported rice grains (*Oryza sativa*) sold in Saudi Arabia and their potential health risk. *Int. J. Hyg. Environ. Health* **2017**, 220, 1168–1178. [CrossRef]
- Baruah, S.G.; Ahmed, I.; Das, B.; Ingtipi, B.; Boruah, H.; Gupta, S.K.; Nema, A.K.; Chabukdhara, M.; Chabukdhara, M. Heavy metal(loid)s contamination and health risk assessment of soil-rice system in rural and peri-urban areas of lower brahmaputra valley, northeast India. *Chemosphere* 2021, 266, 129150. [CrossRef] [PubMed]
- 131. Djahed, B.; Taghavi, M.; Farzadkia, M.; Norzaee, S.; Miri, M. Stochastic exposure and health risk assessment of rice contamination to the heavy metals in the market of Iranshahr, Iran. *Food Chem. Toxicol.* **2018**, *115*, 405–412. [CrossRef]

- Gu, S.Y.; Shin, H.C.; Kim, D.J.; Park, S.U.; Kim, Y.K. The content and health risk assessment of micro and toxic elements in cereals (oat and quinoa), legumes (lentil and chick pea), and seeds (chia, hemp, and flax). J. Food Compos. Anal. 2021, 99, 103881. [CrossRef]
- 133. Pirsaheb, M.; Hadei, M.; Sharafi, K. Human health risk assessment by Monte Carlo simulation method for heavy metals of commonly consumed cereals in Iran- Uncertainty and sensitivity analysis. *J. Food Compos. Anal.* **2021**, *96*, 103697. [CrossRef]
- Román-Ochoa, Y.; Choque Delgado, G.T.; Tejada, T.R.; Yucra, H.R.; Durand, A.E.; Hamaker, B.R. Heavy metal contamination and health risk assessment in grains and grain-based processed food in Arequipa region of Peru. *Chemosphere* 2021, 274, 129792. [CrossRef] [PubMed]
- 135. Tyagi, N.; Raghuvanshi, R.; Upadhyay, M.K.; Srivastava, A.K.; Suprasanna, P.; Srivastava, S. Elemental (As, Zn, Fe and Cu) analysis and health risk assessment of rice grains and rice based food products collected from markets from different cities of Gangetic basin, India. J. Food Compos. Anal. 2020, 93, 103612. [CrossRef]
- Udowelle, N.A.; Igweze, Z.N.; Asomugha, R.N.; Orisakwe, O.E. Health risk assessment and dietary exposure of polycyclic aromatic hydrocarbons (PAHs), lead and cadmium from bread consumed in Nigeria. *Rocz. Państwowego Zakładu Hig.* 2017, 68, 269–280.
- 137. Wei, J.; Cen, K. Contamination and health risk assessment of heavy metals in cereals, legumes, and their products: A case study based on the dietary structure of the residents of Beijing, China. J. Clean. Prod. 2020, 260, 121001. [CrossRef]
- 138. Zhang, Y.; Cao, S.; Zhang, Z.; Meng, X.; Hsiaoping, C.; Yin, C.; Jiang, H.; Wang, S. Nutritional quality and health risks of wheat grains from organic and conventional cropping systems. *Food Chem.* **2020**, *308*, 125584. [CrossRef]
- 139. Zioła-Frankowska, A.; Karaś, K.; Mikołajczak, K.; Kurzyca, I.; Kowalski, A.; Frankowski, M. Identification of metal (loid) s compounds in fresh and pre-baked bread with evaluation of risk health assessment. *J. Cereal Sci.* **2021**, *97*, 103164. [CrossRef]
- 140. Winiarska-Mieczan, A.; Kwiatkowska, K.; Kwiecień, M.; Zaricka, E. Assessment of the risk of exposure to cadmium and lead as a result of the consumption of coffee infusions. *Biol. Trace Elem. Res.* **2021**, *199*, 2420–2428. [CrossRef] [PubMed]
- 141. Mohajer, A.; Baghani, A.N.; Sadighara, P.; Ghanati, K.; Nazmara, S. Determination and health risk assessment of heavy metals in imported rice bran oil in Iran. *J. Food Compos. Anal.* **2020**, *86*, 103384. [CrossRef]
- 142. Niu, B.; Zhang, H.; Zhou, G.; Zhang, S.; Yang, Y.; Deng, X.; Chen, Q. Safety risk assessment and early warning of chemical contamination in vegetable oil. *Food Control* **2021**, *125*, 107970. [CrossRef]
- 143. Abd-Elghany, S.M.; Zaher, H.A.; Elgazzar, M.M.; Sallam, K.I. Effect of boiling and grilling on some heavy metal residues in crabs and shrimps from the Mediterranean Coast at Damietta region with their probabilistic health risk assessment. *J. Food Compos. Anal.* 2020, 93, 103606. [CrossRef]
- 144. Adebiyi, F.M.; Ore, O.T.; Ogunjimi, I.O. Evaluation of human health risk assessment of potential toxic metals in commonly consumed crayfish (*Palaemon hastatus*) in Nigeria. *Heliyon* 2020, *6*, e03092. [CrossRef] [PubMed]
- 145. Al Ghoul, L.; Abiad, M.G.; Jammoul, A.; Matta, J.; El Darra, N. Zinc, aluminium, tin and Bis-phenol a in canned tuna fish commercialized in Lebanon and its human health risk assessment. *Heliyon* **2020**, *6*, e04995. [CrossRef] [PubMed]
- 146. Ali, M.M.; Ali, M.L.; Rakib, M.R.J.; Islam, M.S.; Bhuyan, M.S.; Senapathi, V.; Chung, S.Y.; Roy, P.D.; Sekar, S.; Islam, A.R.M.T.; et al. Seasonal behavior and accumulation of some toxic metals in commercial fishes from Kirtankhola tidal river of Bangladesh—A health risk taxation. *Chemosphere* 2022, 301, 134660. [CrossRef]
- 147. Alva, C.V.; Mársico, E.T.; Ribeiro, R.D.O.R.; da Silva Carneiro, C.; Simões, J.S.; da Silva Ferreira, M. Concentrations and health risk assessment of total mercury in canned tuna marketed in Southest Brazil. *J. Food Compos. Anal.* 2020, *88*, 103357. [CrossRef]
- 148. Baki, M.A.; Hossain, M.M.; Akter, J.; Quraishi, S.B.; Shojib, M.F.H.; Ullah, A.A.; Khan, M.F. Concentration of heavy metals in seafood (fishes, shrimp, lobster and crabs) and human health assessment in Saint Martin Island, Bangladesh. *Ecotoxicol. Environ. Saf.* 2018, 159, 153–163. [CrossRef]
- 149. Biswas, C.; Soma, S.S.; Rohani, M.F.; Rahman, M.H.; Bashar, A.; Hossain, M.S. Assessment of heavy metals in farmed shrimp, *Penaeus monodon* sampled from Khulna, Bangladesh: An inimical to food safety aspects. *Heliyon* **2021**, *7*, e06587. [CrossRef]
- Bristy, M.S.; Sarker, K.K.; Baki, M.A.; Quraishi, S.B.; Hossain, M.M.; Islam, A.; Khan, M.F. Health risk estimation of metals bioaccumulated in commercial fish from coastal areas and rivers in Bangladesh. *Environ. Toxicol. Pharmacol.* 2021, *86*, 103666. [CrossRef]
- 151. Cano-Sancho, G.; Sioen, I.; Vandermeersch, G.; Jacobs, S.; Robbens, J.; Nadal, M.; Domingo, J.L. Integrated risk index for seafood contaminants (IRISC): Pilot study in five European countries. *Environ. Res.* **2015**, *143*, 109–115. [CrossRef]
- 152. Chai, M.; Li, R.; Gong, Y.; Shen, X.; Yu, L. Bioaccessibility-corrected health risk of heavy metal exposure via shellfish consumption in coastal region of China. *Environ. Pollut.* **2021**, 273, 116529. [CrossRef]
- 153. Chan, M.W.H.; Hasan, K.A.; Balthazar-Silva, D.; Mirani, Z.A.; Asghar, M. Evaluation of heavy metal pollutants in salt and seawater under the influence of the Lyari River and potential health risk assessment. *Mar. Pollut. Bull.* 2021, 166, 112215. [CrossRef] [PubMed]
- 154. Da Silva, C.A.; Garcia, C.A.B.; de Santana, H.L.P.; de Pontes, G.C.; Wasserman, J.C.; da Costa, S.S.L. Metal and metalloid concentrations in marine fish marketed in Salvador, BA, northeastern Brazil, and associated human health risks. *Reg. Stud. Mar. Sci.* 2021, *43*, 101716. [CrossRef]
- 155. Effah, E.; Aheto, D.W.; Acheampong, E.; Tulashie, S.K.; Adotey, J. Human health risk assessment from heavy metals in three dominant fish species of the Ankobra river, Ghana. *Toxicol. Rep.* **2021**, *8*, 1081–1086. [CrossRef] [PubMed]

- 156. El-Sherbiny, H.M.M.; Sallam, K.I. Residual contents and health risk assessment of mercury, lead and cadmium in sardine and mackerel from the Mediterranean Sea Coast, Egypt. *J. Food Compos. Anal.* **2021**, *96*, 103749. [CrossRef]
- 157. Fathabad, A.E.; Tajik, H.; Najafi, M.L.; Jafari, K.; Mousavi Khaneghah, A.; Fakhri, Y.; Thai, V.N.; Oliver Conti, G.; Miri, M. The concentration of the potentially toxic elements (PTEs) in the muscle of fishes collected from Caspian Sea: A health risk assessment study. *Food Chem. Toxicol.* 2021, 154, 112349. [CrossRef]
- Felix, C.S.; Junior JB, P.; da Silva Junior, J.B.; Cruz, A.S.; Dantas, K.G.; Ferreira, S.L. Determination and human health risk assessment of mercury in fish samples. *Talanta* 2022, 247, 123557. [CrossRef] [PubMed]
- 159. Gu, X.; Wang, Z.; Wang, J.; Ouyang, W.; Wang, B.; Xin, M.; Lian, M.; Lu, S.; Lin, C.; He, M.; et al. Sources, trophodynamics, contamination and risk assessment of toxic metals in a coastal ecosystem by using a receptor model and Monte Carlo simulation. *J. Hazard. Mater.* **2022**, 424, 127482. [CrossRef]
- Habib, M.R.; Hoque, M.M.; Kabir, J.; Akhter, S.; Rahman, M.S.; Moore, J.; Jolly, Y.N. A comparative study of heavy metal exposure risk from the consumption of some common species of cultured and captured fishes of Bangladesh. *J. Food Compos. Anal.* 2022, 108, 104455. [CrossRef]
- 161. Ikem, A.; Ayodeji, O.J.; Wetzel, J. Human health risk assessment of selected metal(loid)s via crayfish (*Faxonius virilis; Procambarus acutus acutus*) consumption in Missouri. *Heliyon* **2021**, *7*, e07194. [CrossRef]
- 162. Islam, G.R.; Habib, M.R.; Waid, J.L.; Rahman, M.S.; Kabir, J.; Akter, S.; Jolly, Y.N. Heavy metal contamination of freshwater prawn (*Macrobrachium rosenbergii*) and prawn feed in Bangladesh: A market-based study to highlight probable health risks. *Chemosphere* 2017, 170, 282–289. [CrossRef]
- 163. Javed, M.; Usmani, N. Accumulation of heavy metals and human health risk assessment via the consumption of freshwater fish *Mastacembelus armatus* inhabiting, thermal power plant effluent loaded canal. *SpringerPlus* **2016**, *5*, 776. [CrossRef] [PubMed]
- 164. Joseph, A.; Edet, U.; Etinosa-Okankan, O.; Ekanem, S. Health risk assessment of heavy metals and radionuclides in *Cynoglossus senegalensis* (Sole fish) from Qua Iboe River, South-South Nigeria. *J. Food Compos. Anal.* **2022**, *114*, 104854. [CrossRef]
- Karayakar, F.; Işık, U.; Cicik, B.; Canli, M. Heavy metal levels in economically important fish species sold by fishermen in Karatas (Adana/TURKEY). J. Food Compos. Anal. 2022, 106, 104348. [CrossRef]
- 166. Keshavarzi, B.; Hassanaghaei, M.; Moore, F.; Rastegari Mehr, M.; Soltanian, S.; Lahijanzadeh, A.R.; Sorooshian, A. Heavy metal contamination and health risk assessment in three commercial fish species in the Persian Gulf. *Mar. Pollut. Bull.* 2018, 129, 245–252. [CrossRef] [PubMed]
- 167. Korkmaz, C.; Ay, Ö.; Ersoysal, Y.; Köroğlu, M.A.; Erdem, C. Heavy metal levels in muscle tissues of some fish species caught from north-east Mediterranean: Evaluation of their effects on human health. *J. Food Compos. Anal.* **2019**, *81*, 1–9. [CrossRef]
- Lin, C.; Chen, J.; Xu, J.; Li, Y.; Liu, Y.; Lin, H. Distribution and risk assessment of heavy metals in the economic fish of the Southern Fujian Province. Environ. *Toxicol. Pharmacol.* 2022, *92*, 103834. [CrossRef] [PubMed]
- Liu, H.; Liu, G.; Yuan, Z.; Ge, M.; Wang, S.; Liu, Y.; Da, C. Occurrence, potential health risk of heavy metals in aquatic organisms from Laizhou Bay, China. *Mar. Pollut. Bull.* 2019, 140, 388–394. [CrossRef] [PubMed]
- Liu, M.; Xu, Y.; Nawab, J.; Rahman, Z.; Khan, S.; Idress, M.; Uddin, Z.; Ali, A.; Ahmad, R.; Khan, S.A.; et al. Contamination features, geo-accumulation, enrichments and human health risks of toxic heavy metal(loids) from fish consumption collected along Swat river, Pakistan. *Environ. Technol. Innov.* 2020, 17, 100554. [CrossRef]
- 171. Magna, E.K.; Koranteng, S.S.; Donkor, A.; Gordon, C. Health Risk Assessment and Levels of Heavy Metals in Farmed Nile Tilapia (*Oreochromis niloticus*) from the Volta Basin of Ghana. *J. Chem.* **2021**, 2021, 2273327. [CrossRef]
- 172. Matouke, M.M.; Abdullahi, K.L. Assessment of heavy metals contamination and human health risk in *Clarias gariepinus* [Burchell, 1822] collected from Jabi Lake, Abuja, Nigeria. *Sci. Afr.* **2020**, *7*, e00292. [CrossRef]
- 173. Maurya, P.K.; Malik, D.S.; Yadav, K.K.; Kumar, A.; Kumar, S.; Kamyab, H. Bioaccumulation and potential sources of heavy metal contamination in fish species in River Ganga basin: Possible human health risks evaluation. *Toxicol. Rep.* 2019, 6, 472–481. [CrossRef] [PubMed]
- 174. Mohiuddin, M.; Hossain, M.B.; Ali, M.M.; Hossain, M.K.; Habib, A.; Semme, S.A.; Rakib, M.R.J.; Rahman, M.A.; Yu, J.; Al-Sadoon, M.K.; et al. Human health risk assessment for exposure to heavy metals in finfish and shellfish from a tropical estuary. *J. King Saud Univ.-Sci.* 2022, 34, 102035. [CrossRef]
- 175. Montojo, U.M.; Baldoza, B.J.S.; Cambia, F.D.; Benitez, K.C.D.; Perelonia, K.B.S.; Rivera, A.T.F. Levels and health risk assessment of mercury, cadmium, and lead in green mussel (*Perna viridis*) and oyster (*Crassostrea iredalei*) harvested around Manila Bay, Philippines. *Food Control* 2021, 124, 107890. [CrossRef]
- 176. Mwakalapa, E.B.; Simukoko, C.K.; Mmochi, A.J.; Mdegela, R.H.; Berg, V.; Bjorge Müller, M.H.; Lyche, J.L.; Polder, A. Heavy metals in farmed and wild milkfish (*Chanos chanos*) and wild mullet (*Mugil cephalus*) along the coasts of Tanzania and associated health risk for humans and fish. *Chemosphere* **2019**, 224, 176–186. [CrossRef]
- 177. Nędzarek, A.; Formicki, K.; Kowalska-Góralska, M.; Dobrzański, Z. Concentration and risk of contamination with trace elements in acipenserid and salmonid roe. *J. Food Compos. Anal.* **2022**, *110*, 104525. [CrossRef]
- 178. Popovic, A.R.; Djinovic-Stojanovic, J.M.; Djordjevic, D.S.; Relic, D.J.; Vranic, D.V.; Milijasevic, M.P.; Pezo, L.L. Levels of toxic elements in canned fish from the Serbian markets and their health risks assessment. *J. Food Compos. Anal.* 2018, 67, 70–76. [CrossRef]

- 179. Rahman, M.S.; Akther, S.; Ahmed, A.S.; Saha, N.; Rahman, L.S.; Ahmed, M.K.; Arai, T.; Idris, A.M. Distribution and source apportionment of toxic and trace elements in some benthic and pelagic coastal fish species in Karnaphuli River Estuary, Bangladesh: Risk to human health. *Mar. Pollut. Bull.* **2022**, *183*, 114044. [CrossRef] [PubMed]
- Rajaram, R.; Ganeshkumar, A.; Vinothkannan, A. Health risk assessment and bioaccumulation of toxic metals in commercially important finfish and shellfish resources collected from Tuticorin coast of Gulf of Mannar, Southeastern India. *Mar. Pollut. Bull.* 2020, 159, 111469. [CrossRef] [PubMed]
- Raknuzzaman, M.; Ahmed, M.K.; Islam, M.S.; Habibullah-Al-Mamun, M.; Tokumura, M.; Sekine, M.; Masunaga, S. Trace metal contamination in commercial fish and crustaceans collected from coastal area of Bangladesh and health risk assessment. *Environ. Sci. Pollut. Res.* 2016, 23, 17298–17310. [CrossRef]
- 182. Saha, N.; Mollah, M.Z.I.; Alam, M.F.; Safiur Rahman, M. Seasonal investigation of heavy metals in marine fishes captured from the Bay of Bengal and the implications for human health risk assessment. *Food Control* **2016**, *70*, 110–118. [CrossRef]
- Selvam, S.; Venkatramanan, S.; Hossain, M.B.; Chung, S.Y.; Khatibi, R.; Nadiri, A.A. A study of health risk from accumulation of metals in commercial edible fish species at Tuticorin coasts of southern India. *Estuar. Coast. Shelf Sci.* 2020, 245, 106929. [CrossRef]
- Sharifian, S.; Mortazavi, M.S.; Nozar, S.L.M. Health risk assessment of commercial fish and shrimp from the North Persian Gulf. *J. Trace Elem. Med. Biol.* 2022, 72, 127000. [CrossRef]
- 185. Soegianto, A.; Wahyuni, H.I.; Yulianto, B.; Abd Manaf, L. Health risk assessment of metals in mud crab (*Scylla serrata*) from the East Java Estuaries of Indonesia. *Environ. Toxicol. Pharmacol.* **2022**, *90*, 103810. [CrossRef]
- 186. Sofoulaki, K.; Kalantzi, I.; Machias, A.; Pergantis, S.A.; Tsapakis, M. Metals in sardine and anchovy from Greek coastal areas: Public health risk and nutritional benefits assessment. *Food Chem. Toxicol.* **2019**, *123*, 113–124. [CrossRef]
- 187. Soltani, N.; Moore, F.; Keshavarzi, B.; Sorooshian, A.; Javid, R. Potentially toxic elements (PTEs) and polycyclic aromatic hydrocarbons (PAHs) in fish and prawn in the Persian Gulf, Iran. *Ecotoxicol. Environ. Saf.* 2019, 173, 251–265. [CrossRef] [PubMed]
- 188. Storelli, A.; Barone, G.; Dambrosio, A.; Garofalo, R.; Busco, A.; Storelli, M.M. Occurrence of trace metals in fish from South Italy: Assessment risk to consumer's health. *J. Food Compos. Anal.* **2020**, *90*, 103487. [CrossRef]
- Tan, Y.; Peng, B.; Wu, Y.; Xiong, L.; Sun, J.; Peng, G.; Bai, X. Human health risk assessment of toxic heavy metal and metalloid intake via consumption of red swamp crayfish (*Procambarus clarkii*) from rice-crayfish co-culture fields in China. *Food Control* 2021, 128, 108181. [CrossRef]
- 190. Ukaogo, P.O.; Tang, J.; Ahuchaogu, A.A.; Igwe, O.U.; Obike, A.I.; Emole, P.O.; Aljerf, L.; Onah, O.E.; Tijjani, A.; Ajong, A.B. Evaluation of concentrations of trace metal (loid) s in indigenous crab species and human health risk implications. *Emerg. Contam.* 2022, *8*, 371–380. [CrossRef]
- 191. Ullah, A.K.M.A.; Maksud, M.A.; Khan, S.R.; Lutfa, L.N.; Quraishi, S.B. Dietary intake of heavy metals from eight highly consumed species of cultured fish and possible human health risk implications in Bangladesh. *Toxicol. Rep.* **2017**, *4*, 574–579. [CrossRef]
- 192. Wang, Q.; Fan, Z.; Qiu, L.; Liu, X.; Yin, Y.; Song, C.; Chen, J. Occurrence and health risk assessment of residual heavy metals in the Chinese mitten crab (*Eriocheir sinensis*). J. Food Compos. Anal. 2021, 97, 103787. [CrossRef]
- 193. Wang, J.; Shan, Q.; Liang, X.; Guan, F.; Zhang, Z.; Huang, H.; Fang, H. Levels and human health risk assessments of heavy metals in fish tissue obtained from the agricultural heritage rice-fish-farming system in China. J. Hazard. Mater. 2020, 386, 121627. [CrossRef]
- 194. Wang, X.N.; Gu, Y.G.; Wang, Z.H. Fingerprint characteristics and health risks of trace metals in market fish species from a large aquaculture producer in a typical arid province in Northwestern China. *Environ. Technol. Innov.* 2020, 19, 100987. [CrossRef] [PubMed]
- 195. Xiong, B.; Xu, T.; Li, R.; Johnson, D.; Ren, D.; Liu, H.; Xi, Y.; Huang, Y. Heavy metal accumulation and health risk assessment of crayfish collected from cultivated and uncultivated ponds in the Middle Reach of Yangtze River. *Sci. Total Environ.* 2020, 739, 139963. [CrossRef] [PubMed]
- 196. Xu, C.; Yan, H.; Zhang, S. Heavy metal enrichment and health risk assessment of karst cave fish in Libo, Guizhou, China. *Alex. Eng. J.* **2021**, *60*, 1885–1896. [CrossRef]
- 197. Yi, Y.; Tang, C.; Yi, T.; Yang, Z.; Zhang, S. Health risk assessment of heavy metals in fish and accumulation patterns in food web in the upper Yangtze River, China. *Ecotoxicol. Environ. Saf.* **2017**, *145*, 295–302. [CrossRef]
- 198. Yu, B.; Wang, X.; Dong, K.F.; Xiao, G.; Ma, D. Heavy metal concentrations in aquatic organisms (fishes, shrimp and crabs) and health risk assessment in China. *Mar. Pollut. Bull.* **2020**, *159*, 111505. [CrossRef] [PubMed]
- 199. Zhong, W.; Zhang, Y.; Wu, Z.; Yang, R.; Chen, X.; Yang, J.; Zhu, L. Health risk assessment of heavy metals in freshwater fish in the central and eastern North China. *Ecotoxicol. Environ. Saf.* **2018**, 157, 343–349. [CrossRef]
- Abbasi, H.; Shah, M.H.; Mohiuddin, M.; Elshikh, M.S.; Hussain, Z.; Alkahtani, J.; Abbasi, A.M. Quantification of heavy metals and health risk assessment in processed fruits' products. *Arab. J. Chem.* 2020, 13, 8965–8978. [CrossRef]
- 201. Antoine, J.M.; Fung, L.A.H.; Grant, C.N. Assessment of the potential health risks associated with the aluminium, arsenic, cadmium and lead content in selected fruits and vegetables grown in Jamaica. *Toxicol. Rep.* **2017**, *4*, 181–187. [CrossRef]
- Barea-Sepúlveda, M.; Espada-Bellido, E.; Ferreiro-González, M.; Benítez-Rodríguez, A.; López-Castillo, J.G.; Palma, M.; Barbero, G.F. Metal concentrations in Lactarius mushroom species collected from Southern Spain and Northern Morocco: Evaluation of health risks and benefits. J. Food Compos. Anal. 2021, 99, 103859. [CrossRef]
- Bibi, N.; Shah, M.H.; Khan, N.; Mahmood, Q.; Aldosari, A.A.; Abbasi, A.M. Analysis and health risk assessment of heavy metals in some onion varieties. *Arab. J. Chem.* 2021, 14, 103364. [CrossRef]

- 204. Chen, Z.; Muhammad, I.; Zhang, Y.; Hu, W.; Lu, Q.; Wang, W.; Huang, B.; Hao, M. Transfer of heavy metals in fruits and vegetables grown in greenhouse cultivation systems and their health risks in Northwest China. *Sci. Total Environ.* 2021, 766, 142663. [CrossRef] [PubMed]
- 205. Cherfi, A.; Cherfi, M.; Maache-Rezzoug, Z.; Rezzoug, S.A. Risk assessment of heavy metals via consumption of vegetables collected from different supermarkets in La Rochelle, France. *Environ. Monit. Assess.* **2016**, *188*, 136. [CrossRef] [PubMed]
- 206. Getaneh, A.; Guadie, A.; Tefera, M. Levels of heavy metals in ginger (*Zingiber officinale* Roscoe) from selected districts of Central Gondar Zone, Ethiopia and associated health risk. *Heliyon* 2021, 7, e06924. [CrossRef]
- 207. Ghasemidehkordi, B.; Malekirad, A.A.; Nazem, H.; Fazilati, M.; Salavati, H.; Shariatifar, N.; Rezaei, M.; Fakhri, Y.; Khaneghah, A.M. Concentration of lead and mercury in collected vegetables and herbs from Markazi province, Iran: A non-carcinogenic risk assessment. *Food Chem. Toxicol.* 2018, 113, 204–210. [CrossRef]
- 208. Haque, M.M.; Niloy, N.M.; Khirul, M.A.; Alam, M.F.; Tareq, S.M. Appraisal of probabilistic human health risks of heavy metals in vegetables from industrial, non-industrial and arsenic contaminated areas of Bangladesh. *Heliyon* **2021**, *7*, e06309. [CrossRef]
- Hussain, N.; shafiq Ahmed, M.; Hussain, S.M. Potential health risks assessment cognate with selected heavy metals contents in some vegetables grown with four different irrigation sources near Lahore, Pakistan. Saudi J. Biol. Sci. 2022, 29, 1813–1824. [CrossRef]
- Hwang, I.M.; Ha, J.-H. Human health risk assessment of toxic elements in South Korean cabbage, Kimchi, using Monte Carlo simulations. J. Food Compos. Anal. 2021, 102, 104046. [CrossRef]
- Kharazi, A.; Leili, M.; Khazaei, M.; Alikhani, M.Y.; Shokoohi, R. Human health risk assessment of heavy metals in agricultural soil and food crops in Hamadan, Iran. J. Food Compos. Anal. 2021, 100, 103890. [CrossRef]
- Liang, H.; Wu, W.-L.; Zhang, Y.-H.; Zhou, S.-J.; Long, C.-Y.; Wen, J.; Wang, B.-Y.; Liu, Z.-T.; Zhang, C.-T.; Huang, P.-P.; et al. Levels, temporal trend and health risk assessment of five heavy metals in fresh vegetables marketed in Guangdong Province of China during 2014–2017. *Food Control* 2018, 92, 107–120. [CrossRef]
- Liu, X.; Gu, S.; Yang, S.; Deng, J.; Xu, J. Heavy metals in soil-vegetable system around E-waste site and the health risk assessment. *Sci. Total Environ.* 2021, 779, 146438. [CrossRef]
- Lučić, M.; Miletić, A.; Savić, A.; Lević, S.; Ignjatović, I.S.; Onjia, A. Dietary intake and health risk assessment of essential and toxic elements in pepper (*Capsicum annuum*). J. Food Compos. Anal. 2022, 111, 104598. [CrossRef]
- 215. Mohammadpour, A.; Emadi, Z.; Keshtkar, M.; Mohammadi, L.; Motamed-Jahromi, M.; Samaei, M.R.; Zarei, A.A.; Berizi, E.; Khaneghah, A.M. Assessment of potentially toxic elements (PTEs) in fruits from Iranian market (Shiraz): A health risk assessment study. J. Food Compos. Anal. 2022, 114, 104826. [CrossRef]
- 216. Naangmenyele, Z.; Ncube, S.; Akpabey, F.J.; Dube, S.; Nindi, M.M. Levels and potential health risk of elements in two indigenous vegetables from Golinga irrigation farms in the Northern Region of Ghana. *J. Food Compos. Anal.* **2021**, *96*, 103750. [CrossRef]
- 217. Nie, J.Y.; Kuang, L.X.; Li, Z.X.; Xu, W.H.; Cheng, W.A.N.G.; Chen, Q.S.; Li, A.; Zhao, X.; Xie, H.; Zhao, D.; et al. Assessing the concentration and potential health risk of heavy metals in China's main deciduous fruits. *J. Integr. Agric.* 2016, 15, 1645–1655. [CrossRef]
- 218. Nwachukwu, J.I.; Clarke, L.J.; Symeonakis, E.; Brearley, F.Q. Assessment of human exposure to food crops contaminated with lead and cadmium in Owerri, South-eastern Nigeria. *J. Trace Elem. Miner.* **2022**, *2*, 100037. [CrossRef]
- Ogunkunle, C.O.; Obidele, R.A.; Ayoola, N.O.; Okunlola, G.O.; Rufai, A.B.; Olatunji, O.A.; Adetunji, A.T.; Jimoh, M.A. Potential toxic elements in market vegetables from urban areas of southwest Nigeria: Concentration levels and probabilistic potential dietary health risk among the population. J. Trace Elem. Miner. 2022, 1, 100004. [CrossRef]
- 220. Quispe, N.; Zanabria, D.; Chavez, E.; Cuadros, F.; Carling, G.; Paredes, B. Health risk assessment of heavy metals (Hg, Pb, Cd, Cr and As) via consumption of vegetables cultured in agricultural sites in Arequipa, Peru. Chem. *Data Collect.* 2021, 33, 100723. [CrossRef]
- 221. Rezaei, M.; Ghasemidehkordi, B.; Peykarestan, B.; Shariatifar, N.; Jafari, M.; Fakhri, Y.; Jabbari, M.; Mousavi Khaneghah, A. Potentially Toxic Element Concentration in Fruits Collected from Markazi Province (Iran): A Probabilistic Health Risk Assessment. Biomed. *Environ. Sci.* 2019, 32, 839–853.
- 222. Shaheen, N.; Irfan, N.M.; Khan, I.N.; Islam, S.; Islam, M.S.; Ahmed, M.K. Presence of heavy metals in fruits and vegetables: Health risk implications in Bangladesh. *Chemosphere* **2016**, *152*, 431–438. [CrossRef] [PubMed]
- 223. Shokri, S.; Abdoli, N.; Sadighara, P.; Mahvi, A.H.; Esrafili, A.; Gholami, M.; Jannat, B.; Yousefi, M. Risk assessment of heavy metals consumption through onion on human health in Iran. *Food Chem. X* 2022, 14, 100283. [CrossRef] [PubMed]
- Zhang, T.; Xu, W.; Lin, X.; Yan, H.; Ma, M.; He, Z. Assessment of heavy metals pollution of soybean grains in North Anhui of China. Sci. Total Environ. 2019, 646, 914–922. [CrossRef]
- 225. Abbasi, E.; Yousefi, M.H.; Hashemi, S.; Hosseinzadeh, S.; Ghadimi, A.H.; Safapour, M.; Azari, A. Aflatoxin B1 and heavy metals in imported black tea to Bushehr, southern Iran; Contamination rate and risk assessment. J. Food Compos. Anal. 2022, 106, 104277. [CrossRef]
- 226. Akhtar, S.; Riaz, M.; Naeem, I.; Gong, Y.Y.; Ismail, A.; Hussain, M.; Akram, K. Risk assessment of aflatoxins and selected heavy metals through intake of branded and non-branded spices collected from the markets of Multan city of Pakistan. *Food Control* 2020, 112, 107132. [CrossRef]
- Krstić, M.; Stupar, M.; Đukić-Ćosić, D.; Baralić, K.; Mračević, S.Đ. Health risk assessment of toxic metals and toxigenic fungi in commercial herbal tea samples from Belgrade, Serbia. J. Food Compos. Anal. 2021, 104, 104159. [CrossRef]

- 228. Tao, C.; Song, Y.; Chen, Z.; Zhao, W.; Ji, J.; Shen, N.; Ayoko, G.A.; Frost, R.L. Geological load and health risk of heavy metals uptake by tea from soil: What are the significant influencing factors? *Catena* 2021, 204, 105419. [CrossRef]
- 229. Tefera, M.; Teklewold, A. Health risk assessment of heavy metals in selected Ethiopian spices. Heliyon 2021, 7, e07048. [CrossRef]
- Traina, A.; Bono, G.; Bonsignore, M.; Falco, F.; Giuga, M.; Quinci, E.M.; Vitale, S.; Sprovieri, M. Heavy metals concentrations in some commercially key species from Sicilian coasts (Mediterranean Sea): Potential human health risk estimation. *Ecotoxicol. Environ. Saf.* 2019, 168, 466–478. [CrossRef]
- 231. Zhang, W.H.; Yan, Y.; Yu, R.L.; Hu, G.R. The sources-specific health risk assessment combined with APCS/MLR model for heavy metals in tea garden soils from south Fujian Province, China. *Catena* **2021**, 203, 105306. [CrossRef]
- 232. Barone, G.; Storelli, A.; Meleleo, D.; Dambrosio, A.; Garofalo, R.; Busco, A.; Storelli, M.M. Levels of mercury, methylmercury and selenium in fish: Insights into children food safety. *Toxics* **2021**, *9*, 39. [CrossRef] [PubMed]
- Köşker, A.R.; Gündoğdu, S.; Ayas, D.; Bakan, M. Metal levels of processed ready-to-eat stuffed mussels sold in Turkey: Health risk estimation. J. Food Compos. Anal. 2022, 106, 104326. [CrossRef]
- Christophoridis, C.; Kosma, A.; Evgenakis, E.; Bourliva, A.; Fytianos, K. Determination of heavy metals and health risk assessment of cheese products consumed in Greece. J. Food Compos. Anal. 2019, 82, 103238. [CrossRef]
- Adam, A.A.; Sackey, L.N.; Ofori, L.A. Risk assessment of heavy metals concentration in cereals and legumes sold in the Tamale Aboabo market, Ghana. *Heliyon* 2022, *8*, e10162. [CrossRef] [PubMed]
- 236. Chakraborty, T.K.; Ghosh, G.C.; Hossain, M.R.; Islam, M.S.; Habib, A.; Zaman, S.; Bosu, H.; Nice, M.S.; Haldar, M.; Khan, A.S. Human health risk and receptor model-oriented sources of heavy metal pollution in commonly consume vegetable and fish species of high Ganges river floodplain agro-ecological area, Bangladesh. *Heliyon* 2022, *8*, e11172. [CrossRef]
- 237. Islam, M.S.; Ahmed, M.K.; Habibullah-Al-Mamun, M.; Raknuzzaman, M. The concentration, source and potential human health risk of heavy metals in the commonly consumed foods in Bangladesh. *Ecotoxicol. Environ. Saf.* 2015, 122, 462–469. [CrossRef] [PubMed]
- Kowalska, G. The safety assessment of toxic metals in commonly used herbs, spices, tea, and coffee in Poland. Int. J. Environ. Res. Public Health 2021, 18, 5779. [CrossRef] [PubMed]
- Nuapia, Y.; Chimuka, L.; Cukrowska, E. Assessment of heavy metals in raw food samples from open markets in two African cities. *Chemosphere* 2018, 196, 339–346. [CrossRef]
- Sanaei, F.; Amin, M.M.; Alavijeh, Z.P.; Esfahani, R.A.; Sadeghi, M.; Bandarrig, N.S.; Fatehizadeh, A.; Taheri, E.; Rezakazemi, M. Health risk assessment of potentially toxic elements intake via food crops consumption: Monte Carlo simulation-based probabilistic and heavy metal pollution index. *Environ. Sci. Pollut. Res.* 2021, 28, 1479–1490. [CrossRef]
- 241. Sun, S.; Zhang, H.; Luo, Y.; Guo, C.; Ma, X.; Fan, J.; Chen, J.; Geng, N. Occurrence, accumulation, and health risks of heavy metals in Chinese market baskets. *Sci. Total Environ.* **2022**, *829*, 154597. [CrossRef]
- 242. Wang, W.; Gong, Y.; Greenfield, B.K.; Nunes, L.M.; Yang, Q.; Lei, P.; Bu, W.; Wang, B.; Zhao, X.; Huang, L.; et al. Relative contribution of rice and fish consumption to bioaccessibility-corrected health risks for urban residents in eastern China. *Environ. Int.* 2021, 155, 106682. [CrossRef]
- Zheng, S.; Wang, Q.; Yuan, Y.; Sun, W. Human health risk assessment of heavy metals in soil and food crops in the Pearl River Delta urban agglomeration of China. *Food Chem.* 2020, 316, 126213. [CrossRef] [PubMed]
- Hariri, E.; Abboud, M.I.; Demirdjian, S.; Korfali, S.; Mroueh, M.; Taleb, R.I. Carcinogenic and neurotoxic risks of acrylamide and heavy metals from potato and corn chips consumed by the Lebanese population. J. Food Compos. Anal. 2015, 42, 91–97. [CrossRef]
- 245. Boon, P.E.; Pustjens, A.M.; Te Biesebeek, J.D.; Brust, G.M.H.; Castenmiller, J.J.M. Dietary intake and risk assessment of elements for 1-and 2-year-old children in the Netherlands. *Food Chem. Toxicol.* **2022**, *161*, 112810. [CrossRef] [PubMed]
- 246. Cubadda, F.; D'Amato, M.; Aureli, F.; Raggi, A.; Mantovani, A. Dietary exposure of the Italian population to inorganic arsenic: The 2012–2014 Total Diet Study. *Food Chem. Toxicol.* 2016, *98*, 148–158. [CrossRef]
- Doménech, E.; Martorell, S. Formulation and application of the probability of exceedance metric for risk characterization of non-threshold chemical hazards in food. *Food Control* 2021, 124, 107910. [CrossRef]
- 248. Malavolti, M.; Fairweather-Tait, S.J.; Malagoli, C.; Vescovi, L.; Vinceti, M.; Filippini, T. Lead exposure in an Italian population: Food content, dietary intake and risk assessment. *Food Res. Int.* **2020**, *137*, 109370. [CrossRef] [PubMed]
- Qing, Y.; Li, Y.; Yang, J.; Li, S.; Gu, K.; Bao, Y.; Zhan, Y.; He, K.; Wang, X.; Li, Y. Risk assessment of mercury through dietary exposure in China. *Environ. Pollut.* 2022, 312, 120026. [CrossRef] [PubMed]
- Okaru, A.O.; Abuga, K.O.; Kibwage, I.O.; Hausler, T.; Luy, B.; Kuballa, T.; Rehm, J.; Lachenmeier, D.W. Aflatoxin contamination in unrecorded beers from Kenya—A health risk beyond ethanol. *Food Control* 2017, *79*, 344–348. [CrossRef]
- 251. Osei, C.B.; Tortoe, C.; Kyereh, E.; Adjei-Mensah, R.; Johnson, P.N.T.; Aryee, D. Levels of aflatoxins, heavy and trace metal contaminants in two non-alcoholic beverages, Asaana and Nmedaa, and two alcoholic beverages, Burukutu and Pito produced by the informal sector in Ghana. *Sci. Afr.* **2021**, *12*, e00813. [CrossRef]
- Andrade, P.D.; Dias, J.V.; Souza, D.M.; Brito, A.P.; van Donkersgoed, G.; Pizzutti, I.R.; Caldas, E.D. Mycotoxins in cereals and cereal-based products: Incidence and probabilistic dietary risk assessment for the Brazilian population. *Food Chem. Toxicol.* 2020, 143, 111572. [CrossRef] [PubMed]
- 253. Assunção, R.; Vasco, E.; Nunes, B.; Loureiro, S.; Martins, C.; Alvito, P. Single-compound and cumulative risk assessment of mycotoxins present in breakfast cereals consumed by children from Lisbon region, Portugal. *Food Chem. Toxicol.* 2015, *86*, 274–281. [CrossRef] [PubMed]

- 254. Bandé, M.; Traoré, I.; Nikiema, F.; Kpoda, D.S.; Bazié, B.S.R.; Ouédraogo, M.; Ilboudo, I.; Sama, O.I.; Compaoré, A.K.M.; Meda, N.-I.S.D.; et al. Aflatoxins contents determination in some foodstuffs in Burkina Faso and human health risk assessment. *Toxicon X* 2022, *16*, 100138. [CrossRef] [PubMed]
- 255. Echodu, R.; Maxwell Malinga, G.; Moriku Kaducu, J.; Ovuga, E.; Haesaert, G. Prevalence of aflatoxin, ochratoxin and deoxynivalenol in cereal grains in northern Uganda: Implication for food safety and health. *Toxicol. Rep.* 2019, 6, 1012–1017. [CrossRef] [PubMed]
- 256. Gilbert Sandoval, I.; Wesseling, S.; Rietjens, I.M.C.M. Aflatoxin B1 in nixtamalized maize in Mexico; occurrence and accompanying risk assessment. *Toxicol. Rep.* 2019, *6*, 1135–114. [CrossRef]
- 257. Hanvi, M.D.; Lawson-Evi, P.; Bouka, E.C.; Eklu-Gadegbeku, K. Aflatoxins in maize dough and dietary exposure in rural populations of Togo. *Food Control* 2021, 121, 107673. [CrossRef]
- 258. Ji, X.; Xiao, Y.; Wang, W.; Lyu, W.; Wang, X.; Li, Y.; Deng, T.; Yang, H. Mycotoxins in cereal-based infant foods marketed in China: Occurrence and risk assessment. *Food Control* **2022**, *138*, 108998. [CrossRef]
- 259. Kortei, N.K.; Akomeah Agyekum, A.; Akuamoa, F.; Baffour, V.K.; Wiisibie Alidu, H. Risk assessment and exposure to levels of naturally occurring aflatoxins in some packaged cereals and cereal based foods consumed in Accra, Ghana. *Toxicol. Rep.* 2019, 6, 34–41. [CrossRef] [PubMed]
- 260. Martinez-Miranda, M.M.; Rosero-Moreano, M.; Taborda-Ocampo, G. Occurrence, dietary exposure and risk assessment of aflatoxins in arepa, bread and rice. *Food Control* **2019**, *98*, 359–366. [CrossRef]
- Sinphithakkul, P.; Poapolathep, A.; Klangkaew, N.; Imsilp, K.; Logrieco, A.F.; Zhang, Z.; Poapolathep, S. Occurrence of multiple mycotoxins in various types of rice and barley samples in Thailand. J. Food Prot. 2019, 82, 1007–1015. [CrossRef] [PubMed]
- Wielogorska, E.; Mooney, M.; Eskola, M.; Ezekiel, C.N.; Stranska, M.; Krska, R.; Elliott, C. Occurrence and Human-Health Impacts of Mycotoxins in Somalia. J. Agric. Food Chem. 2019, 67, 2052–2060. [CrossRef] [PubMed]
- 263. Wokorach, G.; Landschoot, S.; Anena, J.; Audenaert, K.; Echodu, R.; Haesaert, G. Mycotoxin profile of staple grains in northern Uganda: Understanding the level of human exposure and potential risks. *Food Control* **2021**, *122*, 107813. [CrossRef]
- Bessaire, T.; Perrin, I.; Tarres, A.; Bebius, A.; Reding, F.; Theurillat, V. Mycotoxins in green coffee: Occurrence and risk assessment. Food Control 2019, 96, 59–67. [CrossRef]
- García-Moraleja, A.; Font, G.; Mañes, J.; Ferrer, E. Analysis of mycotoxins in coffee and risk assessment in Spanish adolescents and adults. *Food Chem. Toxicol.* 2015, *86*, 225–233. [CrossRef] [PubMed]
- Oueslati, S.; Yakhlef, S.B.; Vila-Donat, P.; Pallarés, N.; Ferrer, E.; Barba, F.J.; Berrada, H. Multi-mycotoxin determination in coffee beans marketed in Tunisia and the associated dietary exposure assessment. *Food Control* 2022, 140, 109127. [CrossRef]
- 267. Yazdanfar, N.; Mahmudiono, T.; Fakhri, Y.; Mahvi, A.H.; Sadighara, P.; Mohammadi, A.A.; Yousefi, M. Concentration of ochratoxin A in coffee products and probabilistic health risk assessment. *Arab. J. Chem.* **2022**, *15*, 104376. [CrossRef]
- Omar, S.S. Prevalence, level and health risk assessment of mycotoxins in the fried poultry eggs from Jordan. *Environ. Res.* 2021, 200, 111701. [CrossRef]
- 269. Heshmati, A.; Zohrevand, T.; Khaneghah, A.M.; Mozaffari Nejad, A.S.; Sant'Ana, A.S. Co-occurrence of aflatoxins and ochratoxin A in dried fruits in Iran: Dietary exposure risk assessment. *Food Chem. Toxicol.* **2017**, *106*, 202–208. [CrossRef]
- 270. Saleh, I.; Goktepe, I. Health risk assessment of Patulin intake through apples and apple-based foods sold in Qatar. *Heliyon* **2019**, *5*, e02754. [CrossRef]
- 271. Al Ayoubi, M.; Solfrizzo, M.; Gambacorta, L.; Watson, I.; El Darra, N. Risk of exposure to aflatoxin B1, ochratoxin A, and fumonisin B1 from spices used routinely in Lebanese cooking. *Food Chem. Toxicol.* **2021**, 147, 111895. [CrossRef] [PubMed]
- 272. Caldeirão, L.; Sousa, J.; Nunes, L.C.G.; Godoy, H.T.; Fernandes, J.O.; Cunha, S.C. Herbs and herbal infusions: Determination of natural contaminants (mycotoxins and trace elements) and evaluation of their exposure. *Food Res. Int.* 2021, 144, 110322. [CrossRef]
- Cui, P.; Yan, H.; Granato, D.; Ho, C.-T.; Ye, Z.; Wang, Y.; Zhang, L.; Zhou, Y. Quantitative analysis and dietary risk assessment of aflatoxins in Chinese post-fermented dark tea. *Food Chem. Toxicol.* 2020, 146, 111830. [CrossRef] [PubMed]
- 274. Duarte, S.C.; Salvador, N.; Machado, F.; Costa, E.; Almeida, A.; Silva, L.J.; Pereira, A.; Lino, C.; Pena, A. Mycotoxins in teas and medicinal plants destined to prepare infusions in Portugal. *Food Control* **2020**, *115*, 107290. [CrossRef]
- 275. Ye, Z.; Wang, X.; Fu, R.; Yan, H.; Han, S.; Gerelt, K.; Cui, P.; Chen, J.; Qi, Z.; Zhou, Y. Determination of six groups of mycotoxins in Chinese dark tea and the associated risk assessment. *Environ. Pollut.* **2020**, *261*, 114180. [CrossRef] [PubMed]
- Elzupir, A.O.; Abdulkhair, B.Y. Health risk from aflatoxins in processed meat products in Riyadh, KSA. *Toxicon* 2020, 181, 1–5. [CrossRef]
- 277. Bahrami, R.; Shahbazi, Y.; Nikousefat, Z. Aflatoxin M1 in milk and traditional dairy products from west part of Iran: Occurrence and seasonal variation with an emphasis on risk assessment of human exposure. *Food Control* **2016**, *62*, 250–256. [CrossRef]
- Conteçotto, A.C.T.; Pante, G.C.; Castro, J.C.; Souza, A.A.; Lini, R.S.; Romoli, J.C.Z.; Abreu Filho, B.A.; Mikcha, J.M.G.; Mossini, S.A.G.; Machinski Junior, M. Occurrence, exposure evaluation and risk assessment in child population for aflatoxin M1 in dairy products in Brazil. *Food Chem. Toxicol.* 2021, 148, 111913. [CrossRef]
- Costamagna, D.; Gaggiotti, M.; Signorini, M.L. Quantitative risk assessment for aflatoxin M1 associated with the consumption of milk and traditional dairy products in Argentina. *Mycotoxin Res.* 2021, 37, 315–325. [CrossRef]

- Hooshfar, S.; Khosrokhavar, R.; Yazdanpanah, H.; Eslamizad, S.; Kobarfard, F.; Nazari, F.; Kokaraki, V.; Kokkinakis, M.; Goumenou, M.; Tsitsimpikou, C.; et al. Health risk assessment of aflatoxin M1 in infant formula milk in IR Iran. *Food Chem. Toxicol.* 2020, 142, 111455. [CrossRef]
- 281. Kaur, S.; Bedi, J.S.; Dhaka, P.; Vijay, D.; Aulakh, R.S. Exposure assessment and risk characterization of aflatoxin M1 through consumption of market milk and milk products in Ludhiana, Punjab. *Food Control* **2021**, *126*, 107991. [CrossRef]
- 282. Milićević, D.R.; Milešević, J.; Gurinović, M.; Janković, S.; Đinović-Stojanović, J.; Zeković, M.; Glibetić, M. Dietary exposure and risk assessment of aflatoxin m1 for children aged 1 to 9 years old in Serbia. *Nutrients* **2021**, *13*, 4450. [CrossRef]
- Sakin, F.; Tekeli, İ.O.; Yipel, M.; Kürekci, C. Occurrence and health risk assessment of aflatoxins and ochratoxin a in Sürk, a Turkish dairy food, as studied by HPLC. *Food Control* 2018, 90, 317–323. [CrossRef]
- Shahbazi, Y.; Nikousefat, Z.; Karami, N. Occurrence, seasonal variation and risk assessment of exposure to aflatoxin M1 in Iranian traditional cheeses. *Food Control* 2017, 79, 356–362. [CrossRef]
- Torović, L.; Popov, N.; Živkov-Baloš, M.; Jakšić, S. Risk estimates of hepatocellular carcinoma in Vojvodina (Serbia) related to aflatoxin M1 contaminated cheese. J. Food Compos. Anal. 2021, 103, 104122. [CrossRef]
- Al Jabir, M.; Barcaru, A.; Latiff, A.; Jaganjac, M.; Ramadan, G.; Horvatovich, P. Dietary exposure of the Qatari population to food mycotoxins and reflections on the regulation limits. *Toxicol. Rep.* 2019, *6*, 975–982. [CrossRef] [PubMed]
- Carballo, D.; Moltó, J.C.; Berrada, H.; Ferrer, E. Presence of mycotoxins in ready-to-eat food and subsequent risk assessment. *Food Chem. Toxicol.* 2018, 121, 558–565. [CrossRef] [PubMed]
- Do, T.H.; Tran, S.C.; Le, C.D.; Nguyen, H.B.T.; Le, P.T.T.; Le, H.H.T.; Le, T.D.; Thai-Nguyen, H.T. Dietary exposure and health risk characterization of aflatoxin B1, ochratoxin A, fumonisin B1, and zearalenone in food from different provinces in Northern Vietnam. *Food Control* 2020, *112*, 107108. [CrossRef]
- Ezekiel, C.N.; Ayeni, K.I.; Akinyemi, M.O.; Sulyok, M.; Oyedele, O.A.; Babalola, D.A.; Ogara, I.M.; Krska, R. Dietary risk assessment and consumer awareness of mycotoxins among household consumers of cereals, nuts and legumes in north-central Nigeria. *Toxins* 2021, 13, 635. [CrossRef]
- 290. Fang, L.; Zhao, B.; Zhang, R.; Wu, P.; Zhao, D.; Chen, J.; Pan, X.; Wang, J.; Wu, X.; Zhang, H.; et al. Occurrence and exposure assessment of aflatoxins in Zhejiang province, China. *Environ. Toxicol. Pharmacol.* 2022, 92, 103847. [CrossRef]
- Hajok, I.; Kowalska, A.; Piekut, A.; Ćwieląg-Drabek, M. A risk assessment of dietary exposure to ochratoxin A for the Polish population. *Food Chem.* 2019, 284, 264–269. [CrossRef]
- 292. Kabak, B. Aflatoxins in foodstuffs: Occurrence and risk assessment in Turkey. J. Food Compos. Anal. 2021, 96, 103734. [CrossRef]
- 293. Kholif, O.T.; Sebaei, A.S.; Eissa, F.I.; Elhamalawy, O.H. Determination of aflatoxins in edible vegetable oils from Egyptian market: Method development, validation, and health risk assessment. *J. Food Compos. Anal.* **2022**, 105, 104192. [CrossRef]
- Li, Y.; Wang, J.; Li, J.; Sun, X. An improved overall risk probability-based method for assessing the combined health risks of chemical mixtures: An example about mixture of aflatoxin B1 and microcystin LR by dietary intake. *Food Chem. Toxicol.* 2020, 146, 111815. [CrossRef] [PubMed]
- 295. Nugraha, A.; Khotimah, K.; Rietjens, I.M. Risk assessment of aflatoxin B1 exposure from maize and peanut consumption in Indonesia using the margin of exposure and liver cancer risk estimation approaches. *Food Chem. Toxicol.* 2018, 113, 134–144. [CrossRef]
- 296. Ojuri, O.T.; Ezekiel, C.N.; Sulyok, M.; Ezeokoli, O.T.; Oyedele, O.A.; Ayeni, K.I.; Eskola, M.K.; Šarkanj, B.; Hajšlová, J.; Adeleke, R.A.; et al. Assessing the mycotoxicological risk from consumption of complementary foods by infants and young children in Nigeria. *Food Chem. Toxicol.* 2018, 121, 37–50. [CrossRef] [PubMed]
- 297. Adetunji, M.C.; Alika, O.P.; Awa, N.P.; Atanda, O.O.; Mwanza, M. Microbiological quality and risk assessment for aflatoxins in groundnuts and roasted cashew nuts meant for human consumption. *J. Toxicol.* **2018**, 2018, 1308748. [CrossRef] [PubMed]
- Cunha, S.C.; Sa, S.V.; Fernandes, J.O. Multiple mycotoxin analysis in nut products: Occurrence and risk characterization. *Food Chem. Toxicol.* 2018, 114, 260–269. [CrossRef] [PubMed]
- 299. Kong, D.; Wang, G.; Tang, Y.; Guo, M.; Khan, Z.U.H.; Guo, Y.; Gu, W.; Ma, Y.; Sui, M.; Li, J.; et al. Potential health risk of areca nut consumption: Hazardous effect of toxic alkaloids and aflatoxins on human digestive system. *Food Res. Int.* 2022, 162, 112012. [CrossRef] [PubMed]
- Oyedele, O.A.; Ezekiel, C.N.; Sulyok, M.; Adetunji, M.C.; Warth, B.; Atanda, O.O.; Krska, R. Mycotoxin risk assessment for consumers of groundnut in domestic markets in Nigeria. *Int. J. Food Microbiol.* 2017, 251, 24–32. [CrossRef]
- Pongpraket, M.; Poapolathep, A.; Wongpanit, K.; Tanhan, P.; Giorgi, M.; Zhang, Z.; Li, P.; Poapolathep, S. Exposure assessment of multiple mycotoxins in black and white sesame seeds consumed in Thailand. *J. Food Prot.* 2020, 83, 1198–1207. [CrossRef]
- 302. Qin, M.; Liang, J.; Yang, D.; Yang, X.; Cao, P.; Wang, X.; Ma, N.; Zhang, L. Spatial analysis of dietary exposure of aflatoxins in peanuts and peanut oil in different areas of China. *Food Res. Int.* 2021, 140, 109899. [CrossRef] [PubMed]
- Wang, X.; Lien, K.W.; Ling, M.P. Probabilistic health risk assessment for dietary exposure to aflatoxin in peanut and peanut products in Taiwan. *Food Control* 2018, 91, 372–380. [CrossRef]
- Celik, D.; Kabak, B. Assessment to propose a maximum permitted level for ochratoxin A in dried figs. J. Food Compos. Anal. 2022, 112, 104705. [CrossRef]
- 305. Quiles, J.M.; Saladino, F.; Mañes, J.; Fernández-Franzón, M.; Meca, G. Occurrence of mycotoxins in refrigerated pizza dough and risk assessment of exposure for the Spanish population. *Food Chem. Toxicol.* **2016**, *94*, 19–24. [CrossRef] [PubMed]

- 306. Mitchell, N.J.; Chen, C.; Palumbo, J.D.; Bianchini, A.; Cappozzo, J.; Stratton, J.; Ryu, D.; Wu, F. A risk assessment of dietary Ochratoxin a in the United States. *Food Chem. Toxicol.* 2017, 100, 265–273. [CrossRef]
- 307. Zhang, W.; Liu, Y.; Liang, B.; Zhang, Y.; Zhong, X.; Luo, X.; Huang, J.; Wang, Y.; Cheng, W.; Chen, K. Probabilistic risk assessment of dietary exposure to aflatoxin B1 in Guangzhou, China. Sci. Rep. 2020, 10, 7973. [CrossRef]
- Basaran, B.; Anlar, P.; Oral, Z.F.Y.; Polat, Z.; Kaban, G. Risk assessment of acrylamide and 5-hydroxymethyl-2-furfural (5-HMF) exposure from bread consumption: Turkey. J. Food Compos. Anal. 2022, 107, 104409. [CrossRef]
- Eslamizad, S.; Kobarfard, F.; Tsitsimpikou, C.; Tsatsakis, A.; Tabib, K.; Yazdanpanah, H. Health risk assessment of acrylamide in bread in Iran using LC-MS/MS. *Food Chem. Toxicol.* 2019, 126, 162–168. [CrossRef]
- Yazdanpanah, H.; Kobarfard, F.; Tsitsimpikou, C.; Eslamizad, S.; Alehashem, M.; Tsatsakis, A. Health risk assessment of process-related contaminants in bread. *Food Chem. Toxicol.* 2022, 170, 113482. [CrossRef]
- El-Zakhem Naous, G.; Merhi, A.; Abboud, M.I.; Mroueh, M.; Taleb, R.I. Carcinogenic and neurotoxic risks of acrylamide consumed through caffeinated beverages among the lebanese population (Open Access). *Chemosphere* 2018, 208, 352–357. [CrossRef] [PubMed]
- 312. Ofosu, I.W.; Ankar-Brewoo, G.M.; Lutterodt, H.E.; Benefo, E.O.; Menyah, C.A. Estimated daily intake and risk of prevailing acrylamide content of alkalized roasted cocoa beans. *Sci. Afr.* **2019**, *6*, e00176. [CrossRef]
- Nguyen, K.H.; Nielsen, R.H.; Mohammadifar, M.A.; Granby, K. Formation and mitigation of acrylamide in oven baked vegetable fries. *Food Chem.* 2022, 386, 132764. [CrossRef]
- Zhu, B.; Xu, X.; Ye, X.; Zhou, F.; Qian, C.; Chen, J.; Zhang, T.; Ding, Z. Determination and risk assessment of acrylamide in thermally processed Atractylodis Macrocephalae Rhizoma. *Food Chem.* 2021, 3521, 129438. [CrossRef] [PubMed]
- 315. Seilani, F.; Shariatifar, N.; Nazmara, S.; Khaniki, G.J.; Sadighara, P.; Arabameri, M. The analysis and probabilistic health risk assessment of acrylamide level in commercial nuggets samples marketed in Iran: Effect of two different cooking methods. *J. Environ. Health Sci. Eng.* 2021, 19, 465–473. [CrossRef] [PubMed]
- Cortés, W.R.B.; Mejía, S.M.V.; Mahecha, H.S. Consumption study and margin of exposure of acrylamide in food consumed by the Bogotá population in Colombia. J. Food Compos. Anal. 2021, 100, 103934. [CrossRef]
- 317. Chiu, S.Y.; Lin, H.T.; Ho, W.C.; Lin, M.H.; Chen, P.C.; Huang, H.Y. Application of food description to the food classification system: Evidence of risk assessment from Taiwan as Acrylamide of grain products. *J. Food Drug Anal.* **2018**, *26*, 1312–1319. [CrossRef]
- 318. Cieślik, I.; Cieślik, E.; Topolska, K.; Surma, M. Dietary acrylamide exposure from traditional food products in lesser Poland and associated risk assessment. *Ann. Agric. Environ. Med.* 2020, 27, 225–230. [CrossRef]
- 319. Claeys, W.; De Meulenaer, B.; Huyghebaert, A.; Scippo, M.-L.; Hoet, P.; Matthys, C. Reassessment of the acrylamide risk: Belgium as a case-study. *Food Control* **2016**, *59*, 628–635. [CrossRef]
- Jakobsen, L.S.; Granby, K.; Knudsen, V.K.; Nauta, M.; Pires, S.M.; Poulsen, M. Burden of disease of dietary exposure to acrylamide in Denmark. *Food Chem. Toxicol.* 2016, 90, 151–159. [CrossRef] [PubMed]
- 321. Lee, S.; Kim, H.J. Dietary exposure to acrylamide and associated health risks for the korean population. *Int. J. Environ. Res. Public Health* **2020**, *17*, 7619. [CrossRef] [PubMed]
- 322. Mencin, M.; Abramovič, H.; Vidrih, R.; Schreiner, M. Acrylamide levels in food products on the Slovenian market. *Food Control* **2020**, *114*, 107267. [CrossRef]
- 323. Nematollahi, A.; Kamankesh, M.; Hosseini, H.; Ghasemi, J.; Hosseini-Esfahani, F.; Mohammadi, A.; Mousavi Khaneghah, A. Acrylamide content of collected food products from Tehran's market: A risk assessment study. Environ. *Sci. Pollut. Res.* 2020, 27, 30558–30570. [CrossRef] [PubMed]
- 324. Shahrbabki, P.E.; Hajimohammadi, B.; Shoeibi, S.; Elmi, M.; Yousefzadeh, A.; Conti, G.O.; Ferrante, M.; Amirahmadi, M.; Fakhri, Y.; Mousavi Khaneghah, A. Probabilistic non-carcinogenic and carcinogenic risk assessments (Monte Carlo simulation method) of the measured acrylamide content in Tah-dig using QuEChERS extraction and UHPLC-MS/MS. *Food Chem. Toxicol.* 2018, 118, 361–370. [CrossRef] [PubMed]
- 325. Vejdovszky, K.; Mihats, D.; Griesbacher, A.; Wolf, J.; Steinwider, J.; Lueckl, J.; Jank, B.; Kopacka, I.; Rauscher-Gabernig, E. Modified Reference Point Index (mRPI) and a decision tree for deriving uncertainty factors: A practical approach to cumulative risk assessment of food contaminant mixtures. *Food Chem. Toxicol.* 2019, 134, 110812. [CrossRef] [PubMed]
- 326. Wyka, J.; Tajner-Czopek, A.; Broniecka, A.; Piotrowska, E.; Bronkowska, M.; Biernat, J. Estimation of dietary exposure to acrylamide of Polish teenagers from an urban environment. *Food Chem. Toxicol.* **2015**, 75, 151–155. [CrossRef]
- 327. Akbari-Adergani, B.; Shahbazi, R.; Esfandiari, Z.; Kamankesh, M.; Saatloo, N.V.; Abedini, A.; Ramezankhani, R.; Sadighara, P. Acrylamide content of industrial and traditional popcorn collected from Tehran's market, Iran: A risk assessment study. *J. Food Prot.* 2023, *86*, 100001. [CrossRef]
- 328. Deribew, H.A.; Woldegiorgis, A.Z. Acrylamide levels in coffee powder, potato chips and French fries in Addis Ababa city of Ethiopia. *Food Control* 2021, 123, 107727. [CrossRef]
- 329. Esposito, F.; Nardone, A.; Fasano, E.; Triassi, M.; Cirillo, T. Determination of acrylamide levels in potato crisps and other snacks and exposure risk assessment through a Margin of Exposure approach. *Food Chem. Toxicol.* 2017, 108, 249–256. [CrossRef] [PubMed]
- González, N.; Marques, M.; Calderon, J.; Collantes, R.; Corraliza, L.; Timoner, I.; Bosch, J.; Castell, V.; Domingo, J.L.; Nadal, M. Occurrence and dietary intake of food processing contaminants (FPCs) in Catalonia, Spain. *J. Food Compos. Anal.* 2022, 106, 104272. [CrossRef]

- 331. Pratama, Y.; Jacxsens, L. Quantitative risk assessment of acrylamide in Indonesian deep fried fritters as street food products. *Curr. Res. Nutr. Food Sci.* **2019**, *7*, 662–669. [CrossRef]
- 332. Gao, J.; Zhao, Y.; Zhu, F.; Ma, Y.; Li, X.; Miao, H.; Wu, Y. Dietary exposure of acrylamide from the fifth Chinese Total Diet Study. *Food Chem. Toxicol.* **2016**, *87*, 97–102. [CrossRef]
- Li, Y.; Liu, J.; Wang, Y.; Wei, S. Cancer risk and disease burden of dietary acrylamide exposure in China, 2016. *Ecotoxicol. Environ.* Saf. 2022, 238, 113551. [CrossRef]
- 334. Sirot, V.; Rivière, G.; Leconte, S.; Vin, K.; Traore, T.; Jean, J.; Carne, G.; Gorecki, S.; Veyrand, B.; Marchand, P.; et al. French infant total diet study: Dietary exposure to heat-induced compounds (acrylamide, furan and polycyclic aromatic hydrocarbons) and associated health risks. *Food Chem. Toxicol.* 2019, 130, 308–316. [CrossRef]
- Ihedioha, J.N.; Okali, E.E.; Ekere, N.R.; Ezeofor, C.C. Risk assessment of polycyclic aromatic hydrocarbons in pasta products consumed in Nigeria. *Iran. J. Toxicol.* 2019, 13, 19–26.
- 336. Jiang, D.; Xin, C.; Li, W.; Chen, J.; Li, F.; Chu, Z.; Xiao, P.; Shao, L. Quantitative analysis and health risk assessment of polycyclic aromatic hydrocarbons in edible vegetable oils marketed in Shandong of China. *Food Chem. Toxicol.* 2015, 83, 61–67. [CrossRef]
- Lee, J.; Jeong, J.H.; Park, S.; Lee, K.G. Monitoring and risk assessment of polycyclic aromatic hydrocarbons (PAHs) in processed foods and their raw materials. *Food Control* 2018, 92, 286–292. [CrossRef]
- Li, G.; Wu, S.; Wang, L.; Akoh, C.C. Concentration, dietary exposure and health risk estimation of polycyclic aromatic hydrocarbons (PAHs) in youtiao, a Chinese traditional fried food. *Food Control* 2016, *59*, 328–336. [CrossRef]
- Ma, J.K.; Li, K.; Li, X.; Elbadry, S.; Raslan, A.A.; Li, Y.; Mulla, Z.; Tahoun, A.B.M.B.; El-Ghareeb, W.R.; Huang, X.C. Levels of polycyclic aromatic hydrocarbons in edible and fried vegetable oil: A health risk assessment study. *Environ. Sci. Pollut. Res.* 2021, 28, 59784–59791. [CrossRef]
- Yousefi, M.; Shemshadi, G.; Khorshidian, N.; Ghasemzadeh-Mohammadi, V.; Fakhri, Y.; Hosseini, H.; Mousavi Khaneghah, A. Polycyclic aromatic hydrocarbons (PAHs) content of edible vegetable oils in Iran: A risk assessment study. *Food Chem. Toxicol.* 2018, 118, 480–489. [CrossRef]
- Adewale, A.; Adegbola, P.I.; Owoade, A.O.; Aborisade, A.B. Fish as a bioindicator of polycyclic aromatic hydrocarbon pollution in aquatic ecosystem of Ogun and Eleyele Rivers, Nigeria, and risk assessment for consumer's health. *J. Hazard. Mater. Adv.* 2022, 7, 100096. [CrossRef]
- 342. Asamoah, E.K.; Nunoo, F.K.E.; Addo, S.; Nyarko, J.O.; Hyldig, G. Polycyclic aromatic hydrocarbons (PAHs) in fish smoked using traditional and improved kilns: Levels and human health risk implications through dietary exposure in Ghana. *Food Control* 2021, 121, 107576. [CrossRef]
- 343. Barhoumi, B.; El Megdiche, Y.; Clérandeau, C.; Ameur, W.B.; Mekni, S.; Bouabdallah, S.; Derouiche, A.; Touil, S.; Cachot, J.; Driss, M.R. Occurrence of polycyclic aromatic hydrocarbons (PAHs) in mussel (*Mytilus galloprovincialis*) and eel (*Anguilla anguilla*) from Bizerte lagoon, Tunisia, and associated human health risk assessment. *Cont. Shelf Res.* **2016**, *124*, 104–116. [CrossRef]
- 344. Bomfeh, K.; Jacxsens, L.; Amoa-Awua, W.K.; Gamarro, E.G.; Ouadi, Y.D.; De Meulenaer, B. Risk Assessment of Polycyclic Aromatic Hydrocarbons (PAHs) in Smoked Sardinella sp. in Ghana: Impact of an Improved Oven on Public Health Protection. *Risk Anal.* 2022, 42, 1007–1022. [CrossRef] [PubMed]
- 345. Ferrante, M.; Zanghì, G.; Cristaldi, A.; Copat, C.; Grasso, A.; Fiore, M.; Signorelli, S.S.; Zuccarello, P.; Oliveri Conti, G. PAHs in seafood from the Mediterranean Sea: An exposure risk assessment. *Food Chem. Toxicol.* **2018**, *115*, 385–390. [CrossRef]
- 346. Hasselberg, A.E.; Nøstbakken, O.J.; Aakre, I.; Madsen, L.; Atter, A.; Steiner-Asiedu, M.; Kjellevold, M. Nutrient and contaminant exposure from smoked European anchovy (*Engraulis encrasicolus*): Implications for children's health in Ghana. *Food Control* 2022, 134, 108650. [CrossRef]
- 347. Iko Afé, O.H.; Saegerman, C.; Kpoclou, Y.E.; Douny, C.; Igout, A.; Mahillon, J.; Anihouvi, V.B.; Hounhouigan, D.J.; Scippo, M.L. Contamination of smoked fish and smoked-dried fish with polycyclic aromatic hydrocarbons and biogenic amines and risk assessment for the Beninese consumers. *Food Control* **2021**, *126*, 108089. [CrossRef]
- 348. Ju, Y.R.; Chen, C.F.; Wang, M.H.; Chen, C.W.; Dong, C.D. Assessment of polycyclic aromatic hydrocarbons in seafood collected from coastal aquaculture ponds in Taiwan and human health risk assessment. J. Hazard. Mater. 2022, 421, 126708. [CrossRef] [PubMed]
- Kim, W.; Choi, J.; Kang, H.J.; Lee, J.W.; Moon, B.; Joo, Y.S.; Lee, K.W. Monitoring and risk assessment of eight polycyclic aromatic hydrocarbons (PAH8) in daily consumed agricultural products in South Korea. *Polycycl. Aromat. Compd.* 2020, 42, 1141–1156. [CrossRef]
- 350. Li, Y.; Guo, N.; Zou, X.; Li, P.; Zou, S.; Luo, J.; Yang, Y. Pollution level and health risk assessment of polycyclic aromatic hydrocarbons in marine fish from two coastal regions, the South China Sea. *Mar. Pollut. Bull.* **2021**, *168*, 112376. [CrossRef]
- 351. Malesa-Ciećwierz, M.; Szulecka, O.; Adamczyk, M. Polycyclic aromatic hydrocarbon contamination of Polish smoked fish: Assessment of dietary exposure. *J. Food Process. Preserv.* **2019**, *43*, e13962. [CrossRef]
- Maia, M.L.; Paíga, P.; Ramalhosa, M.J.; Delerue-Matos, C.; Calhau, C.; Domingues, V.F. Seasonal and spatial comparison of polycyclic aromatic hydrocarbons among decapod shrimp from coastal Portugal. *Bull. Environ. Contam. Toxicol.* 2022, 109, 511–517. [CrossRef]
- 353. Man, Y.B.; Mo, W.Y.; Zhang, F.; Wong, M.H. Health risk assessments based on polycyclic aromatic hydrocarbons in freshwater fish cultured using food waste-based diets. *Environ. Pollut.* 2020, 256, 113380. [CrossRef] [PubMed]

- 354. Moslen, M.; Miebaka, C.A.; Boisa, N. Bioaccumulation of Polycyclic Aromatic Hydrocarbon (PAH) in a bivalve (*Arca senilis*—blood cockles) and health risk assessment. *Toxicol. Rep.* **2019**, *6*, 990–997. [CrossRef]
- Oliveira, M.; Gomes, F.; Torrinha, Á.; Ramalhosa, M.J.; Delerue-Matos, C.; Morais, S. Commercial octopus species from different geographical origins: Levels of polycyclic aromatic hydrocarbons and potential health risks for consumers. *Food Chem. Toxicol.* 2018, 121, 272–282. [CrossRef] [PubMed]
- 356. Ranjbar Jafarabadi, A.; Mashjoor, S.; Riyahi Bakhtiari, A.; Jadot, C. Dietary intake of polycyclic aromatic hydrocarbons (PAHs) from coral reef fish in the Persian Gulf—Human health risk assessment. *Food Chem.* **2020**, *329*, 127035. [CrossRef]
- 357. Tongo, I.; Ogbeide, O.; Ezemonye, L. Human health risk assessment of polycyclic aromatic hydrocarbons (PAHs) in smoked fish species from markets in Southern Nigeria. *Toxicol. Rep.* **2017**, *4*, 55–61. [CrossRef]
- 358. Wang, H.; Huang, W.; Gong, Y.; Chen, C.; Zhang, T.; Diao, X. Occurrence and potential health risks assessment of polycyclic aromatic hydrocarbons (PAHs) in different tissues of bivalves from Hainan Island, China. *Food Chem. Toxicol.* 2020, 136, 111108. [CrossRef] [PubMed]
- 359. Yu, Z.; Lin, Q.; Gu, Y.; Du, F.; Wang, X.; Shi, F.; Ke, C.; Xiang, M.; Yu, Y. Bioaccumulation of polycyclic aromatic hydrocarbons (PAHs) in wild marine fish from the coastal waters of the northern South China Sea: Risk assessment for human health. *Ecotoxicol. Environ. Saf.* **2019**, *180*, 742–748. [CrossRef] [PubMed]
- Zhang, H.; Chen, Y.; Li, D.; Yang, C.; Zhou, Y.; Wang, X.; Zhang, Z. PAH residue and consumption risk assessment in four commonly consumed wild marine fishes from Zhoushan Archipelago, East China Sea. *Mar. Pollut. Bull.* 2021, 170, 112670. [CrossRef] [PubMed]
- Zhu, Y.; Huang, H.; Zhang, Y.; Xiong, G.; Zhang, Q.; Li, Y.; Tao, S.; Liu, W. Evaluation of PAHs in edible parts of vegetables and their human health risks in Jinzhong City, Shanxi Province, China: A multimedia modeling approach. *Sci. Total Environ.* 2021, 773, 145076. [CrossRef]
- 362. Benson, N.U.; Fred-Ahmadu, O.H.; Olugbuyiro, J.A.O.; Anake, W.U.; Adedapo, A.E.; Olajire, A.A. Concentrations, sources and risk characterisation of polycyclic aromatic hydrocarbons (PAHs) in green, herbal and black tea products in Nigeria. *J. Food Compos. Anal.* 2018, 66, 13–22. [CrossRef]
- Celik-Saglam, I.; Balcik, C.; Cetin, B. Concentrations, sources, and risk assessment of polycyclic aromatic hydrocarbons (PAHs) in black, green and fruit flavored tea in Turkey. J. Food Compos. Anal. 2022, 109, 104504. [CrossRef]
- 364. Menoni, C.; Donangelo, C.M.; Rufo, C. Polycyclic aromatic hydrocarbons in yerba mate (*llex paraguariensis*) infusions and probabilistic risk assessment of exposure. *Toxicol. Rep.* **2021**, *8*, 324–330. [CrossRef]
- 365. Chiang, C.F.; Hsu, K.C.; Cho, C.Y.; Tsai, T.Y.; Hsu, C.H.; Yang, D.J. Comparison and establishment of appropriate methods to determine EU priority PAHs in charcoal-grilled chicken drumsticks with different treatments and their dietary risk assessments. *Food Chem. Toxicol.* 2020, 142, 111400. [CrossRef] [PubMed]
- 366. Darwish, W.S.; Chiba, H.; El-Ghareeb, W.R.; Elhelaly, A.E.; Hui, S.P. Determination of polycyclic aromatic hydrocarbon content in heat-treated meat retailed in Egypt: Health risk assessment, benzo [a] pyrene induced mutagenicity and oxidative stress in human colon (CaCo-2) cells and protection using rosmarinic and ascorbic acids. *Food Chem.* 2019, 290, 114–124.
- Lu, F.; Kuhnle, G.K.; Cheng, Q. Heterocyclic amines and polycyclic aromatic hydrocarbons in commercial ready-to-eat meat products on UK market. *Food Control* 2017, 73, 306–315. [CrossRef]
- Rozentāle, I.; Stumpe-Vīksna, I.; Začs, D.; Siksna, I.; Melngaile, A.; Bartkevičs, V. Assessment of dietary exposure to polycyclic aromatic hydrocarbons from smoked meat products produced in Latvia. *Food Control* 2015, 54, 16–22. [CrossRef]
- 369. Kiani, A.; Ahmadloo, M.; Moazzen, M.; Shariatifar, N.; Shahsavari, S.; Arabameri, M.; Hasani, M.M.; Azari, A.; Abdel-Wahhab, M.A. Monitoring of polycyclic aromatic hydrocarbons and probabilistic health risk assessment in yogurt and butter in Iran. *Food Sci. Nutr.* 2021, 9, 2114–2128. [CrossRef]
- 370. Shariatifar, N.; Dadgar, M.; Fakhri, Y.; Shahsavari, S.; Moazzen, M.; Ahmadloo, M.; Kiani, A.; Aeenehvand, S.; Nazmara, S.; Khanegah, A.M. Levels of polycyclic aromatic hydrocarbons in milk and milk powder samples and their likely risk assessment in Iranian population. *J. Food Compos. Anal.* 2020, *85*, 103331. [CrossRef]
- Yan, K.; Li, W.; Wu, S. Dietary exposure and risk assessment of European Union priority (EU 15+ 1) polycyclic aromatic hydrocarbons from milks and milk powders in China. J. Dairy Sci. 2022, 105, 6536–6547. [CrossRef]
- 372. Aamir, M.; Yin, S.; Liu, Y.; Ullah, H.; Khan, S.; Liu, W. Dietary exposure and cancer risk assessment of the Pakistani population exposed to polycyclic aromatic hydrocarbons. *Sci. Total Environ.* **2021**, *757*, 143828. [CrossRef]
- 373. Badibostan, H.; Feizy, J.; Daraei, B.; Shoeibi, S.; Rajabnejad, S.H.; Asili, J.; Taghizadeh, S.F.; Giesy, J.P.; Karimi, G. Polycyclic aromatic hydrocarbons in infant formulae, follow-on formulae, and baby foods in Iran: An assessment of risk. *Food Chem. Toxicol.* 2019, 131, 110640. [CrossRef] [PubMed]
- Bogdanović, T.; Pleadin, J.; Petričević, S.; Listeš, E.; Sokolić, D.; Marković, K.; Ozogul, F.; Šimat, V. The occurrence of polycyclic aromatic hydrocarbons in fish and meat products of Croatia and dietary exposure. J. Food Compos. Anal. 2019, 75, 49–60. [CrossRef]
- 375. Kim, H.S.; Kim, J.; Choi, J.; Paik, Y.; Moon, B.; Joo, Y.S.; Lee, K.W. Polycyclic aromatic hydrocarbons in beverage and dairy products in South Korea: A risk characterization using the total diet study. *Food Sci. Biotechnol.* 2021, 30, 989–1002. [CrossRef] [PubMed]
- 376. Lee, J.G.; Suh, J.H.; Yoon, H.J. Occurrence and risk characterization of polycyclic aromatic hydrocarbons of edible oils by the Margin of Exposure (MOE) approach. *Appl. Biol. Chem.* **2019**, *62*, 51. [CrossRef]

- 377. Okoye, E.A.; Bocca, B.; Ruggieri, F.; Ezejiofor, A.N.; Nwaogazie, I.L.; Domingo, J.L.; Rovira, J.; Frazzoli, C.; Orisakwe, O.E. Concentrations of polycyclic aromatic hydrocarbons in samples of soil, feed and food collected in the Niger Delta region, Nigeria: A probabilistic human health risk assessment. *Environ. Res.* 2021, 202, 111619. [CrossRef] [PubMed]
- Racovita, R.C.; Secuianu, C.; Israel-Roming, F. Quantification and risk assessment of carcinogenic polycyclic aromatic hydrocarbons in retail smoked fish and smoked cheeses. *Food Control* 2021, 121, 107586. [CrossRef]
- 379. Roudbari, A.; Rafiei Nazari, R.; Shariatifar, N.; Moazzen, M.; Abdolshahi, A.; Mirzamohammadi, S.; Madani-Tonekaboni, M.; Delvarianzadeh, M.; Arabameri, M. Concentration and health risk assessment of polycyclic aromatic hydrocarbons in commercial tea and coffee samples marketed in Iran. Environ. *Sci. Pollut. Res.* 2021, *28*, 4827–4839. [CrossRef]
- Santonicola, S.; Albrizio, S.; Murru, N.; Ferrante, M.C.; Mercogliano, R. Study on the occurrence of polycyclic aromatic hydrocarbons in milk and meat/fish based baby food available in Italy. *Chemosphere* 2017, 184, 467–472. [CrossRef]
- 381. Singh, L.; Agarwal, T. PAHs in Indian diet: Assessing the cancer risk. Chemosphere 2018, 202, 366–376. [CrossRef]
- 382. European Food Safety Authority (EFSA). The 2010 European Union report on pesticide residues in food. *EFSA J.* **2013**, *11*, 3130. [CrossRef]
- 383. Jankowska, M.; Łozowicka, B.; Kaczyński, P. Comprehensive toxicological study over 160 processing factors of pesticides in selected fruit and vegetables after water, mechanical and thermal processing treatments and their application to human health risk assessment. *Sci. Total Environ.* 2019, 652, 1156–1167. [CrossRef] [PubMed]
- 384. Environmental Protection Agency (EPA). Exposure Assessment Tools by Routes—Ingestion Health Risk Assessment of Heavy Metals through the Consumption Offood Crops Fertilized by Biosolids: A Probabilistic-Based Analysis. 2021. Available online: https://www.epa.gov/expobox/exposure-assessment-tools-routes-ingestion (accessed on 15 February 2024).
- Boobis, A.R.; Ossendorp, B.C.; Banasiak, U.; Hamey, P.Y.; Sebestyen, I.; Moretto, A. Cumulative risk assessment of pesticide residues in food. *Toxicol. Lett.* 2008, 180, 137–150. [CrossRef] [PubMed]
- 386. European Food Safety Authority (EFSA). Opinion of the Scientific Committee on a request from EFSA related to a harmonised approach for risk assessment of substances which are both genotoxic and carcinogenic. *EFSA J.* **2005**, *282*, 1–31. [CrossRef]
- 387. Barlow, S.M. Risk Assessment of Foods and Chemicals in Foods. In *Encyclopedia of Food and Health*; Elsevier: Amsterdam, The Netherlands, 2016. [CrossRef]
- 388. European Food Safety Authority (EFSA). EFSA/WHO International Conference with Support of ILSI Europe on Risk Assessment of Compounds that Are Both Genotoxic and Carcinogenic. EFSA Meeting Summary Report. 2006. Available online: https://efsa.onlinelibrary.wiley.com/doi/epdf/10.2903/sp.efsa.2006.EN-92 (accessed on 15 February 2024).
- 389. JECFA. Joint FAO/WHO Expert Committee on Food Additives. Safety Evaluation of Certain Food Additives and Contaminants: Prepared by the Forty-Ninth Meeting of the JECFA, Series 40. World Health Organization. 1998. Available online: http: //www.inchem.org/documents/jecfa/jecmono/v040je16.htm (accessed on 15 February 2024).
- 390. Andrade, P.D.; Caldas, E.D. Aflatoxins in cereals: Worldwide occurrence and dietary risk assessment. *World Mycotoxin J.* 2015, *8*, 415–431. [CrossRef]
- FDA U.S. Food & Drugs Administration. Guidance for Industry: Estimating Dietary Intake of Substances in Food. 2006. Available online: https://www.fda.gov/regulatory-information/search-fda-guidance-documents/guidance-industry-estimating-dietaryintake-substances-food (accessed on 15 February 2024).
- 392. JECFA. Joint FAO/WHO Expert Committee on Food Additives. 66th Meeting (Residues of Veterinary Drugs), Evaluation of Certain Veterinary Drug Residues in Food. WHO TRS 939. 2006. Available online: https://www.fao.org/3/at875e/at875e.pdf (accessed on 15 February 2024).
- 393. European Food Safety Authority (EFSA). Guidance Document on Scientific Criteria for Grouping Chemicals into Assessment Groups for Human Risk Assessment of Combined Exposure to Multiple Chemicals. 2021. Available online: https://www.efsa.europa.eu/sites/default/files/2021-12/7033.pdf (accessed on 15 February 2024).
- 394. Codex Alimentarius Commission (CAC). Maximum Residue Limits (MRLs) and Risk Management Recommendations (RMRs) for Residues of Veterinary Drugs in Foods. 2021. Available online: https://www.fao.org/fao-who-codexalimentarius/sh-proxy/en/ ?lnk=1&url=https%253A%252F%252Fworkspace.fao.org%252Fsites%252Fcodex%252FStandards%252FCXM+2%252FMRL2e. pdf (accessed on 15 February 2024).
- 395. Environmental Protection Agency (EPA). Guidelines for Carcinogen Risk Assessment EPA/630/P-03/001F. 2005. Available online: https://www.epa.gov/sites/default/files/2013-09/documents/cancer\_guidelines\_final\_3-25-05.pdf (accessed on 15 February 2024).
- 396. ATSDR (Agency for Toxic Substances and Disease Registry). Calculating Hazard Quotients and Cancer Risk Estimates. 2022. Available online: https://www.atsdr.cdc.gov/pha-guidance/conducting\_scientific\_evaluations/epcs\_and\_exposure\_calculations/hazardquotients\_cancerrisk.html (accessed on 15 February 2024).
- 397. JECFA. Joint FAO/WHO Expert Committee on Food Additives. Summary and Conclusions of the Sixty-Forth Meeting of the Joint FAO/WHO Expert Committee on Food Additives (JECFA). 2005. pp. 7–17. Available online: https://www.fao.org/3/at877 e/at877e.pdf (accessed on 15 February 2024).

**Disclaimer/Publisher's Note:** The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.